

Board of Directors Meeting Wednesday, June 17, 2020 5:00pm https://us02web.zoom.us/j/87023071843

Or Telephone:

Dial(for higher quality, dial a number based on your current location):

US: +1 669 900 6833 or +1 346 248 7799 or +1 253 215 8782 or +1 929 205 6099 or +1 301 715 8592 or +1 312 626 6799 or 888 475 4499 (Toll Free) or 877 853 5257 (Toll Free)

Webinar ID: 870 2307 1843

Meetings are accessible to people with disabilities. Individuals who need special assistance or a disability-related modification or accommodation to participate in this meeting, or who have a disability and wish to request an alternative format for the meeting materials, should contact the Clerk of the Board at least 2 working days before the meeting at (510) 736-4981 or <u>cob@ebce.org</u>.

If you have anything that you wish to be distributed to the Committee, please email it to the clerk by 4:00 pm the day prior to the meeting.

- 1. Welcome & Roll Call
- 2. Pledge of Allegiance
- 3. Closed Session
 - A. Public Employee Performance Evaluation pursuant to Government Code § 54957. Title: Chief Executive Officer.
 - B. Conference with Legal Counsel Anticipated Litigation, pursuant to Government Code § 54956.9(b): (1 case)
- 4. General Counsel Report Out of Closed Session

5. Public Comment

This item is reserved for persons wishing to address the Board on any EBCE-related matters that are not otherwise on this meeting agenda. Public comments on matters listed on the agenda shall be heard at the time the matter is called. As with all public comment, members of the public who wish to address the Board are customarily limited to three minutes per speaker and must complete an electronic <u>speaker slip</u>. The Board Chair may increase or decrease the time allotted to each speaker.

CONSENT AGENDA

- 6. Approval of Minutes from May 20, 2020
- 7. Contracts entered into (Informational Item)
- 8. Second Amendment to Agreement with TRC Engineers Adopt a Resolution authorizing the CEO to negotiate and execute the Second Amendment to the Consulting Services Agreement with TRC Engineers, Inc. to add new scope and personnel, and allocate an additional \$20,000 to the Agreement, for a total amount not to exceed \$240,000.
- 9. First Amendment to Consulting Services Agreement with Recurve Analytics Adopt a Resolution authorizing the CEO to execute a First Amendment to the Consulting Services Agreement with Recurve Analytics ("Recurve") to extend the term, to add to the scope, and to increase the compensation by \$125,000, for a total amount not to exceed \$225,000.

REGULAR AGENDA

10. CEO REPORT

- A. Executive Committee;
- B. Finance, Administrative and Procurement Subcommittee;
- C. Marketing, Regulatory and Legislative Subcommittee; and
- D. Marketing and Account Services Update
- 11. Community Advisory Committee Report
- 12. Consideration of a merit bonus for the Chief Executive Officer, Pursuant to the Employment Agreement Between the East Bay Community Energy Authority and Nicolas Chaset, Dated October 17, 2018, as Amended Effective December 16, 2019 (Action Item)
- 13. Extension of Temporary Revision to Terms and Conditions Section "Failure to Pay" and Temporary Suspension of Delinquent Accounts and Collections Policy (Action Item) Adopt a Resolution to Extend the Temporary Suspension of Certain Customer Terms and Conditions and Administrative Policies for an Additional Period of 90 Days During the COVID-19 Pandemic, and extend the suspension of the "Failure to Pay" section of the Service Terms and Conditions of the Delinquent Accounts and Collections Policy for an additional 90 days.

- **14.** Enrollment of New Communities: Pleasanton, Newark, and Tracy (Informational Item) Receive information report from staff on proposed enrollment of customers within the new member communities of Pleasanton, Newark, and Tracy during the month of April 2021.
- 15. EBCE Resilience Program (Action Item)
 - A. Approve the selection of two vendors in response to the Distributed Resource Adequacy (RA) Request for Proposals to deliver RA in the form of Load Modification from behind-the-meter solar and storage installations delivering resilience to EBCE commercial and residential customers.
 - B. Authorize the CEO or his designee to negotiate and execute a Load Modifying Agreement with Sunrun Inc. for Load Modification up to 5.75MW/23MWh through December 31, 2032.
 - C. Authorize the CEO or his designee to negotiate and execute a Load Modifying Agreement with Enel X in substantially the same form as the Sunrun agreement for Load Modification up to 5MW/20MWh through December 31, 2032.
- 16. Update on 2020 Integrated Resource Plan (Informational Item) Receive the informational update on process and status of Integrated Resource Planning (IRP) analysis.
- 17. Fiscal Year 2020-2021 Budget (Action Item)
 - A. Approve the presented budget for fiscal year 2020-2021
 - B. Modify the Bright Choice Value Proposition to a 0.5% discount as compared to PG&E Rates, effective July 1, 2020.
 - C. Modify the Brilliant 100 Value Proposition to a 2.5% premium as compared to PG&E Rates, effective September 1, 2020.
 - D. Set a renewables procurement floor at the current year RPS, while maintaining 100% carbon-free content for Brilliant 100 for fiscal year 2020-2021
- 18. Legal Services Contracts Approval (Action Item)

Adopt a Resolution authorizing the Chief Executive Officer to negotiate and execute contracts with select legal firms in accordance with the legal budget, totaling \$970,000 for fiscal year 2020-2021.

- 19. Board Member and Staff Announcements including requests to place items on future Board Agendas
- 20. Adjournment to Date: Wednesday, July 1, 2020 Board Retreat Time: 5:00 pm



Draft Minutes Board of Directors Meeting Wednesday, May 20, 2020

6:00 pm

https://us02web.zoom.us/j/87023071843

Or Telephone:

Dial(for higher quality, dial a number based on your current location): US: +1 669 900 6833 or +1 346 248 7799 or +1 253 215 8782 or +1 929 205 6099 or +1 301 715 8592 or +1 312 626 6799 or 888 475 4499 (Toll Free) or 877 853 5257 (Toll Free) Webinar ID: 870 2307 1843

Meetings are accessible to people with disabilities. Individuals who need special assistance or a disability-related modification or accommodation to participate in this meeting, or who have a disability and wish to request an alternative format for the meeting materials, should contact the Clerk of the Board at least 2 working days before the meeting at (510) 736-4981 or cob@ebce.org.

If you have anything that you wish to be distributed to the Committee, please email it to the clerk by 5:00 pm the day prior to the meeting.

1. Welcome & Roll Call

Present: Haggerty (Alameda County), Pilch (Albany) Arreguin (Berkeley), Hernandez (Dublin), Mei (Fremont), Mendall (Hayward), Munro (Livermore), Hannon (Newark), Rood (Piedmont), Narum (Pleasanton), Hernandez (San Leandro), Arriola (Tracy), Patino (Union City), Sutter (Community Advisory), Martinez (Emeryville), and Kalb (Oakland)

2. Pledge of Allegiance

3. Public Comment

This item is reserved for persons wishing to address the Board on any EBCE-related matters that are not otherwise on this meeting agenda. Public comments on matters listed on the agenda shall be heard at the time the matter is called. As with all public comment, members of the public who wish to address the Board are customarily limited to three minutes per speaker and must complete an electronic <u>speaker slip</u>. The Board Chair may increase or decrease the time allotted to each speaker. **Blair Beekman** - Spoke regarding the energy market, PG&E and the importance of creating energy sustainability.

Aleta Dupree - Spoke in support of public participation and open discussion.

- 4. Closed Session
 - A. PUBLIC EMPLOYEE PERFORMANCE EVALUATION under California Government Code Section 54957: Performance Evaluation of Chief Executive Officer Nicolas Chaset
- 5. General Counsel Report out of Closed Session

There was no reportable action taken in Closed session.

CONSENT AGENDA

- 6. Approval of Minutes from April 22, 2020
- 7. Contracts entered into (Informational Item)
- 8. Amendment to Sponsorship Policy Adopt a Resolution authority approving the first amendment and restatement of the sponsorship policy.

R-2020-6 Policy 12.1

9. Amendment to Administrative Procurement Policy Adopt a Resolution approving a fourth amended and restated administrative procurement policy.

> R-2020-7 Policy 1.6

Director Mendall motioned to approve the Consent Calendar except for Item # 8 Amendment to Sponsorship Policy. Director Arreguin seconded the motion which carried 15/0

Director Hernandez (San Leandro) motioned to approve Item # 8 Amendment to Sponsorship Policy as amended:

A. CEO sponsorship approval level changed to \$50,000

B. <u>Report sponsorship grants to appropriate subcommittee</u>

Director Mendall seconded the motion which carried 12/3; Noes: Directors: Haggerty, Hernandez (Dublin) and Munro

Jessica Tovar - Spoke regarding concerns with sponsorship signing authority and recommended that the board require staff to present proposed sponsorship recipients to the Executive Committee.

James Hausner - Spoke in opposition of the sponsorship program and made recommendations on how it should be administered.

Blair Beekman - Requested addition information on the PG&E Carbon Allotment.

REGULAR AGENDA

10. CEO REPORT

- A. Executive Committee Meeting;
- B. Marketing and Outreach update;
- C. Local Development Business Plan Update and
- D. Update on Opt-out

Blair Beekman - Expressed interest in learning more about how EBCE works and spoke in support of the local projects.

11. Community Advisory Committee Report

The Chair will discuss the CAC meeting when each item is heard.

12. California Electronic Vehicle Infrastructure Project Funding (Action Item)

Adopt a Resolution approving participation in the California Energy Commission's 2021 round of the California Electric Vehicle Infrastructure Project (CALeVIP) in Alameda County and authorize the Chief Executive Officer to execute an agreement with the electric vehicle charging incentives as part of CALeVIP.

The Board Discussed:

- Funding
- Emissions
- Electric Buses/Fleet
- Supporting the purchase of Electric Vehicles (EV)
- Low-income and multiunit housing
- Incentives for installing EV chargers
- Seeking Air District grants

Jessica Tovar - Spoke regarding EBCPA's concerns around the funding of this program with a large portion of community not being able to afford electric vehicles and creating equity with our programs.

Ceyda Dumas - Spoke regarding concerns with the budget set aside for the CALeVIP project during the current pandemic and potential effect on the budget.

Aleta Dupree - Spoke in support of the CALeVIP program highlighting opportunity to partner with transportation agencies to reach a broader audience.

Audrey Ichinose - Spoke in support of portions of the proposal and highlighted concerns with the amount of budget allocated to project, social equity and recommended seeking outside sources to fund project.

Blair Beekman - Spoke regarding mass transit, creating public sustainability and establishing good practices.

Director Mendall motioned to adopt a Resolution Approving the participation in the California Electric Vehicle Infrastructure Project. Director Arreguin seconded the motion which carried 15/0

R-2020-8

13. 2020-2021 Draft Budget Review (Information Item)

The Board Discussed:

- Consultant budget
- Presentation and outreach
- PCIA

Blair Beekman - Spoke in support of creating a positive sustainability model.

Aleta Dupree - Spoke in support of brand awareness and investing in customer relations. Stating that public power is about developing relationships with its customers.

- 14. Board Member and Staff Announcements including requests to place items on future Board Agendas
- 15. Adjourned



Consent Item 7

TO: East Bay Community Energy Board of Directors

FROM: Nick Chaset, Chief Executive Officer

SUBJECT: Contracts Entered Into

DATE: June 17, 2020

RECOMMENDATION

Accept the CEO's report on contracts that EBCE has entered into. During the March 18th, 2020 Board Meeting, the Board of Directors requested that contracts be reported to both the Executive Committee and the Board of Directors as they had increased the CEO's signing authority limit to contracts of a value of up to \$500,000 for a term of 90 days. This authority will revert back to \$100,000 as outlined in the Administrative Procurement Policy on June 16, 2020.

Items executed after May 13 and before May 18;

C-2020-70 Abbott, Stringham and Lynch (San Jose) First Amendment to July 1, 2019 CSA to add Green-e Audit for reporting year 2019, extending the term to 12/31/2020, and adding \$2,200 to the total NTE.

C-2020-71 Eden I&R (Hayward) Grant awarded in the amount of \$10,000 to support staff salaries and overtime due to increased 211 calls, supporting staff with childcare and transitioning to remote work in response to the COVID-19 pandemic.

C-2020-72 Healthy Black Families (Berkeley) Grant awarded in the amount of \$10,000 to provide financial support to its members, including rent, groceries, utilities, health and wellness services for individuals and households in response to the COVID-19 pandemic.

C-2020-73 Tri-City Health Center (Fremont) Grant awarded in the amount of \$10,000 to support emergency rental assistance and financial support for basic needs, existing infrastructure and programs to provide payments for low-income clients and expanding programs and partnerships in response to the COVID-19 pandemic.

C-2020-74 Community Impact Lab (San Leandro) amends Community Innovation Grant Dated August 8, 2019 by extending the termination date to December 30, 2020 as the program may see delays related to COVID-19.

C-2020-75 Ecology Action (Berkeley) amends Community Innovation Grant Dated August 7, 2019 by extending the termination date to December 30, 2020 as the program may see delays related to COVID-19.

C-2020-76 People Power Solar Cooperative (Oakland) amends Community Innovation Grant Dated August 7, 2019 by extending the termination date to December 30, 2020 as the program may see delays related to COVID-19.

C-2020-77 PSMRC (Daly City) Consulting Services Agreement to provide ergonomic and workstation evaluation services to EBCE, not to exceed \$7,200 through June 30, 2020.

C-2020-78 M. Cubed (Oakland) First Amendment to July 1, 2019 CSA adds additional services related to PG&E 2019 ERRA Compliance, updates the hourly rates of compensation, adds \$10,000 to the not to exceed and extends the termination date to June 30, 2021.

C-2020-79 SMUD (Sacramento) Amendment 9 to Exhibit A, EBCE has requested a non-standard rate for one customer, SMUD is tasked to develop a solution, configuration of the new rate, end ot end testing, deployment and post deployment support, implementation by August 2020 for a one-time cost of \$22,000.

Items executed between May 18 and June 10;

C-2020-80 City of San Leandro, COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$79,000 to support small business community grants, safety net food distribution, and digital inclusion, in response to the COVID-19 pandemic.

C-2020-81 Asian Pacific Environmental Network (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support in-language information and resources, assistance with bills, food security, education materials, medical costs, and health/wellness resources in response to the COVID-19 pandemic.

C-2020-82 Centro Legal de la Raza (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support current service offerings in response to the COVID-19 pandemic.

C-2020-83 City Slicker Farms (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support production of local healthy food and educational resources on home gardening and food preparation in response to the COVID-19 pandemic.

C-2020-84 Creating New Hope (Dublin), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to provide personal protective equipment to senior care communities in response to the COVID-19 pandemic. C-2020-85 Bay Area Community Services (BACS) (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support housing and homelessness solutions and homelessness prevention programs in response to the COVID-19 pandemic.

C-2020-86 Drivers for Survivors (Fremont), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support free transportation to and from medical appointments in response to the COVID-19 pandemic.

C-2020-87 East Bay Agency for Children (EBAC) (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support their providing essential household and baby care items and supporting families with health care in response to the COVID-19 pandemic.

C-2020-88 Safe Alternatives to Violent Environments (Fremont), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support their housing first program, maintain their 24/7 crisis line, administer their safe house and prodding emergency hotel rooms, and provide counseling services in response to the COVID-19 pandemic.

C-2020-89 Mujeres Unidas y Activas (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support staff who provide services, such as mental health counseling, in response to the COVID-19 pandemic.

C-2020-90 Energy Council/StopWaste, Cooperative Agreement for Bay Area Regional Heat Pump Water Heater Contractor Incentive Program, EBCE & MCE will be working with the Energy Council to increase the number of Heat Pump Water Heaters in the Bay Area residences to reduce Greenhouse Gas emissions. Agencies will pay \$1,000 to local contractors for each HPWH installed within their service territory (NTE 250, or \$250,000), and a \$200 administrative fee (NTE \$50,000). Agreement term is April 1, 2020 to June 30, 2021.

C-2020-91 Ascend Analytics, (Boulder, CO) Consulting Services Agreement for Integrated Resource Planning Analysis and Valuation Software and Analysis not to exceed\$133,000 through October 31, 2020.

C-2020-92 World Central Kitchen (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support purchasing meals from East Bay restaurants to feed hospital staff in response to the COVID-19 pandemic.

C-2020-93 Eden Area Interfaith Council (Hayward), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support their mask project in response to the COVID-19 pandemic.

C-2020-94 Local Ecology and Agriculture Fremont (Fremont), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support scaling up existing efforts and programs to support the farm, providing gardening supplies, and supporting operational funds such as utilities and water in response to the COVID-19 pandemic. C-2020-95 City of Newark COVID-19 Community Relief Response Grant Agreement, grant awarded in the amount of \$5,000 to support a drive-through food distribution event to provide approximately 200 food boxes in response to the COVID-19 pandemic.

C-2020-96 Re-plate, Inc (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support expanding operations to bring more food deliveries to the community and to support local food banks and other food production and service institutions, in response to the COVID-19 pandemic.

C-2020-97 Choices for Freedom Inc (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support safe and affordable transportation and possible housing for formerly incarcerated people upon their release and supporting families with food and other household essentials in response to the COVID-19 pandemic.

C-2020-98 Mercy Retirement & Care Center's Mercy Brown Bag Program (Oakland) Grant awarded in the amount of \$5,000 to support existing programs for seniors and adults, purchasing food for community members, and supporting staff for programs and delivery, in response to the COVID-19 pandemic.

C-2020-99 Richards, Watson & Gershon (San Francisco), Amendment to Agreement for Legal Services adding \$70,000 to the compensation through June 30, 2020.

C-2020-100 Singer Associates (San Francisco) Consulting Services Agreement for a strategic communications services plan, total amount NTE \$30,000 through December 31, 2020.

C-2020-101 Washington Hospital Health System (Fremont), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$15,000 to support purchasing, procuring and distributing PPE to healthcare workers in Fremont and the surrounding areas in response to the COVID-19 pandemic.

C-2020-102 Prospera Community Development (Oakland), COVID-19 Community Relief Response Grant Agreement awarded in the amount of \$5,000 to support the Latina Entrepreneur Resiliency Fund in response to the COVID-19 pandemic.

C-2020-103 Myst AI (San Francisco), Subscription Agreement for Load Forecasting and Price Forecasting software interfaces for one year term through June 4, 2021, with option to renew on an annual basis. Cost for initial term is \$144,000, if extended, the cost will be \$222,000 annually.



Consent Item 8

то:	East Bay Community Energy Board of Directors
FROM:	Beckie Menten, Program Manager, Building Electrification and Energy Efficiency
SUBJECT:	Second Amendment to Agreement with TRC Engineers
DATE:	June 17, 2020

Recommendation

Adopt a Resolution authorizing the CEO to negotiate and execute the Second Amendment to the Consulting Services Agreement with TRC Engineers, Inc. to add new scope and personnel, and allocate an additional \$20,000 to the Agreement, for a total not to exceed amount of \$240,000.

Background and Discussion

EBCE's Local Programs team is seeking consulting support for a community grant program focused on kitchen electrification in EBCE's service area. The majority of commercial kitchens are currently fueled by primarily by natural gas. In order to achieve our climate goals cost effectively, all buildings must be transitioned to high efficiency electric appliances. These technologies currently exist, and in many cases out-perform their natural gas counterparts in terms of both control and speed as well as in health and safety. Despite this, persistent barriers exist to kitchen electrification, noted most starkly by the lawsuit brought against the City of Berkeley natural gas ban by the California Restaurant Association. Because of this, the need for local examples of kitchen electrification must be supported to help tackle these barriers and raise visibility regarding the benefits of kitchen electrification.

The Local Programs staff is developing a community grant program targeting the full or partial electrification of community and commissary kitchens in the EBCE service area. This grant program will provide financial incentives and technical support to grantees interested in upgrading their kitchens with high efficiency electric alternatives. To support implementation of this grant program, and in accordance with EBCE's procurement policies, EBCE staff sought three informal bids from consulting firms with expertise in this area. After receiving three bids, EBCE staff determined that the bid provided by TRC Engineers best met the needs of staff as well as the budget.

The Resolution before you amend the existing agreement with TRC to increase the budget by \$20,000, for a total amount not to exceed \$240,000, and adds additional scope specific to supporting the kitchen electrification grant program.

Fiscal Impact

This amendment allocates funding from within the FY 19/20 Local Programs budget to support these services; there is no additional impact to the FY 19/20 budget.

Attachments

- A. Resolution Authorizing the CEO to Negotiate and Execute a Second Amendment to the Consulting Services Agreement with TRC Engineers
- B. TRC Engineers Second Amendment to Consulting Services Agreement

RESOLUTION NO.

A RESOLUTION OF THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY TO AUTHORIZE THE CEO TO NEGOTIATE AND EXECUTE A SECOND AMENDMENT TO THE CONSULTING SERVICES AGREEMENT WITH TRC ENGINEERS

WHEREAS The East Bay Community Energy Authority ("EBCE") was formed as a community choice aggregation agency ("CCA") on December 1, 2016, Under the Joint Exercise of Power Act, California Government Code sections 6500 et seq., among the County of Alameda, and the Cities of Albany, Berkeley, Dublin, Emeryville, Fremont, Hayward, Livermore, Piedmont, Oakland, San Leandro, and Union City to study, promote, develop, conduct, operate, and manage energy-related climate change programs in all of the member jurisdictions. The cities of Newark and Pleasanton, located in Alameda County, along with the City of Tracy, located in San Joaquin County, were added as members of EBCE and parties to the JPA in March of 2020.

WHEREAS in February 2019 the Board of Directors authorized the CEO to execute an Agreement with TRC Engineers, Inc. ("TRC") to support reach code development in the EBCE service area. TRC has supported eight EBCE member agencies (Fremont, Hayward, Emeryville, Piedmont, Albany, Oakland, Dublin, Berkeley) in the pursuit of all electric reach codes, providing a high level of support and helping EBCE achieve greenhouse gas reduction goals in the EBCE service area.

WHEREAS in February of 2020 the Board amended the Consulting Services Agreement with TRC to allocate additional funds to support Reach Code Services to cities.

WHEREAS the Local Programs team conducted an informal solicitation to recruit additional consulting services to support local efforts to expand kitchen electrification activities and received three informal bids from consulting firms, of which the TRC bid best met the needs of EBCE.

NOW, THEREFORE, THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY DOES HEREBY RESOLVE AS FOLLOWS:

<u>Section 1.</u> The Board of Directors hereby authorizes the CEO to negotiate and execute an Amendment to the Consulting Services Agreement with TRC Engineers, Inc. to increase the total compensation by \$20,000, for a total amount not to exceed \$240,000, and to add new scope and personnel, in substantially the form attached hereto, with clarifying or clerical changes proposed by the CEO and approved by the General Counsel.

ADOPTED AND APPROVED this 17th day of June, 2020.

Dan Kalb, Chair

ATTEST:

Stephanie Cabrera, Clerk of the Board

Second Amendment to Consulting Services Agreement by and Between East Bay Community Energy Authority and TRC Engineers

This Second Amendment to the Agreement with TRC Engineers for Consulting Services ("Second Amendment") is made this 17 of June, 2020, by and between the East Bay Community Energy Authority, a Joint Powers Agency formed under the laws of the State of California ("EBCE") and TRC Engineers ("TRC"), for the purposes of adding additional compensation, adding additional services to the scope, extending the termination date, and updating the hourly rates of compensation.

Recitals

- A. EBCE and TRC entered into that certain Consulting Services Agreement dated March 21, 2019 ("Agreement"), wherein TRC agreed to provide consulting services to EBCE, with compensation not to exceed \$200,000.
- B. EBCE and TRC entered into that certain First Amendment to the Consulting Services Agreement on 22 February 2020 to add additional compensation, increasing the not-to-exceed amount by \$20,000 for a total amount not to exceed \$220,000.
- C. EBCE and TRC now desire to amend the Agreement to add additional compensation, increasing the not-to-exceed amount by \$20,000 to cover consulting services to for a total amount not to exceed \$240,000, to add additional services to the scope of services including technical assistance to support the implementation of a kitchen electrification grant program, and to add additional staff.

Now therefore, for good and valuable consideration, the amount and sufficiency of which is hereby acknowledged, the Parties agree as follows:

- 1. Section 6.1 of the Agreement ("Compensation and Payment") is amended in part to increase the total compensation of the Agreement by \$20,000 for a total amount not to exceed two hundred and forty thousand dollars (\$240,000).
- 2. Exhibit A ("Scope of Work") of the Agreement is replaced in its entirety by Exhibit A, attached hereto.
- 3. Exhibit C ("Consultant Staffing") of the Agreement is replaced in its entirety by Exhibit C, attached hereto.
- 4. Exhibit E ("Compensation/Budget") of the Agreement is replaced in its entirety by Exhibit E, attached hereto.
- 5. All other terms and conditions in the Agreement not otherwise modified by this Second Amendment will remain in full force and effect.

In witness whereof, the Parties have entered this Amendment on the date written above.

East Bay Community Energy Authority,	TRC Engineers,
A Joint Powers Authority	A California corporation

Nick Chaset Chief Executive Officer	Name: Title:	
Date:	Date:	_
Approved as to form:	Name: Title:	
EBCE General Counsel	Date:	_

Exhibit A

Scope of Work

Program Overview: Reach Code Work

East Bay Community Energy's (EBCE) mission includes reducing greenhouse gas (GHG) emissions in EBCE member municipalities within Alameda County. Emissions from transportation and building natural gas use comprise the largest sources of GHGs within the territories. This program aims to develop and advance Building Electrification and Electric Vehicle (EV) Infrastructure "reach" codes with member municipalities to update local building codes for new residential and commercial construction and major remodels. These codes are to be developed in alignment with and build on the 2019 California statewide building code updates. Codes for new buildings and major modifications are the most cost-effective points at which to install Electric Vehicle charging infrastructure and electric appliances (such as heat pump water heaters) compared to retrofitting existing buildings. As used in this Exhibit A, "municipalities" refers to EBCE's 12-member jurisdictions, including the 11 cities and unincorporated Alameda County, except where context clearly indicates otherwise.

In consideration of the payments set forth in <u>Exhibit E</u>, Consultant shall provide the following services:

1 Administrative Tasks

1.1 Kickoff Meeting

Participate in a kickoff meeting with EBCE to review objectives, budget, timeline, administrative processes and contract at a mutually determined time.

1.2 Monthly Progress Report and Call

Provide a monthly 1-2 page progress report and associated call by the 5th day of the following month to the designated EBCE contract administrator outlining the project's progress, challenges encountered, and objectives for the following month. Deliverables developed in the course of work must be submitted with the progress report such as research summaries, charette notes and attendees, and other deliverables.

1.3 Expense Report

Provide a monthly expense report, submitted with the corresponding monthly progress report, documenting expenses including: labor (hours, rate, total), subconsultant expenses (with invoices), and expenses (with receipts). The expense report must include the total expenditures and remaining funds.

1.4 Final Report

As part of the final invoice at the conclusion of the Term of this Agreement, Consultant shall submit a final report that includes a concise description, summary, or outline of the following:

- 1. Summary of the process undertaken, and stakeholders engaged.
- 2. Outline of the model codes developed including thresholds and performance requirements.
- 3. Brief description of the outcomes, including number of municipalities successfully adopting reach codes, number of municipalities who participated substantively (but may not have adopted new codes).
- 4. Itemization of the tools and resources developed.
- 5. Discussion of challenges and recommendations (including potential program and tariff strategies to advance electrification).

2 Program Tasks

2.1 Research Existing State and Local Codes

Consultant will research relevant state and local building codes to be developed related to building electrification (track 1), electric vehicle (EV) infrastructure (track 2) and alterations (track 3). These codes may serve as models or foundations for the local model reach codes. This research must examine both mandate- and incentive-based codes. The key features will be summarized in format easy for stakeholders to review.

Deliverable: Existing State and Local Reach Code Summary and Links.

2.2 Lead Development of Model Reach Codes

Consultant will work with EBCE to define the process and schedule, provide key education to, and take input from, municipalities and other identified stakeholders to draft model reach codes to achieve program objectives. Specific activities and deliverables are as follows:

2.2.1 Process Summary

The following table summarizes a potential process of model code development, review and revision.

	Building Electrification (BE) Track	Electric Vehicle (EV) Track	Alterations (BA) Track	
Step 1	Consultant disseminates:	Consultant disseminates:	Consultant disseminates:	
	 BE model code variant 1, version 1 BE model code variant 2, version 1 	EV model code	 BE model code variant 1, version 1 BE model code variant 2, version 1 	
Step 2				
	individual stakeholder meetings, and other input channels			
Step 3	Consultant revises all model codes based on stakeholder feedback			
Step 4	Consultant disseminates:	Consultant disseminates:	Consultant disseminates:	
	 BE model code variant 1, version 2 BE model code variant 2, version 2 	EV model code	 BE model code variant 1, version 1 BE model code variant 2, version 1 	
Step 5				
Step 6		revisions to all model codes		

The process, schedule, and stakeholder engagement will be refined in consultation with EBCE.

Deliverable: Final process definition and schedule.

2.2.2 Regional Coordination

Consultant will coordinate with regional and statewide entities that are carrying out building electrification reach code activities (the Building Decarb Coalition, BayREN, NRDC, Sierra Club, etc.), which may be leveraged by or could benefit from the EBCE effort.

Deliverable: Provide regular updates to project team on key resources and activities by regional and state entities

2.2.3 Stakeholder Engagement

EBCE and Consultant will work together to determine which stakeholders to include in the reach code development process, but the stakeholders are anticipated to include elected officials, municipal staff (including relevant planning, building, sustainability staff), architects, developers, affordable housing organizations, electric vehicle experts, building electrification experts, Pacific

Gas & Electric, building trades, environmental groups, building electrification and EV advocates, the groups identified under Task 2.2.2, and other members of the community. The facilitated processes for the three tracks may be handled separately as deemed appropriate and will also include detailing the code development and adoption sequencing plus the program's time-sensitive nature.

The stakeholder engagement process will include a minimum of four charettes. The charettes may be divided by service territory, topic area (Buildings and EVs), or stakeholder class (ex: developers and others) as deemed most effective. Consultant shall conduct individual and stakeholder group (planning and building staff, developers, etc.) outreach with at least ten in person and/or telephone calls per track. To support this process, EBCE will provide facilitation support for their respective territories including charette and meeting locations, invitations, and facilitation support.

Deliverable: Meeting notes and attendee lists for at least 2 charettes and 13 meetings or calls

2.2.4 Model Code Development

The model reach code options must be compliant with California's and extend 2019 Building Code and in conjunction with task 2.3.1 are expected to leverage the cost-effectiveness analysis results from the 2019 Statewide Codes & Standards Program for the building electrification track.

Likewise, for the EV infrastructure track, the proposed codes should integrate Task 2.3.2.

It is expected that two to four variants of each model reach code, with varying levels of stringency, will be developed per track. A single variant of a model reach code for building electrification is expected to address multiple building uses/prototypes (e.g. single family, low-rise multifamily, medium retail, medium office, etc.), and may include prescriptive or performance-based options. The model code development process will include a minimum of one revision of each model code variant.

Task output will consist of a proposed model reach code package, consisting of a summary noting the purpose and adoption timing, a matrix highlighting the differences between the model reach codes, and two to four versions of each model reach code with varying levels of stringency. The Consultant will disseminate the model reach code package to the stakeholders with easily accessible online distribution and will review the model code package in a meeting or webinar. One meeting or webinar will be conducted for each service territory and for each reach code track.

Deliverables:

- 1. Online document distribution method.
- 2. Version 1 of model code variants.
- 3. Final version of model reach codes with supporting documentation.

2.3 Package Building Electrification Cost-Effectiveness Study and Develop EV Cost-Benefit Analysis

2.3.1 Building Electrification Cost-Benefit Analysis

The Consultant will utilize the Statewide Codes & Standards program costeffectiveness analysis for the Building Electrification model codes. The Consultant will package the results of the analysis as needed as a component of the proposed model reach code package in Task 2.2. The Consultant may be requested to provide additional enhanced analysis to supplement the Statewide cost-benefit analysis. If so, this will be treated as technical assistance under Task 2.5.

Deliverable: Building Electrification Cost-Effectiveness Study Results Summary

2.3.2 Develop EV Cost-Effectiveness Analysis

For the EV model codes, Consultant will also conduct an analysis of all costs and benefits – monetary and non-monetary (e.g. emissions impacts) – of the measures included. The EV cost-effectiveness analysis will be developed in a manner consistent with State and industry standards. In addition, the analysis will determine the:

- 2019 CALGreen Standards incremental compliance construction costs;
- Incremental construction costs resulting from the reach code ("first costs");
- Retrofitting costs to install the same charging density; and
- Unit construction cost (per residential unit, ft2 commercial development or other appropriate metric plus any associated parking capital costs).

From the above data, the Consultant will calculate the percentage increase in construction costs resulting from one of the three EV charging infrastructure deployment scenarios plus associated carbon and energy reductions and other non-monetary impacts. The analysis will be documented in a format easy for stakeholders to review and for use by non-EBCE jurisdictions as a self-standing document.

Finally, the Consultant will integrate the cost-benefit analysis as a component of the proposed model reach code package in Task 2.2 and for dissemination as part of Task 2.4.

Deliverable: EV Cost-Effectiveness Analysis Document (may be delivered in a consolidated document with 2.3.1)

2.4 Identify Tools and Processes for Adoption and Implementation

This task includes identifying tools and process improvements to consider for Tasks 2.5 and 2.6. This may include but is not limited to:

2.4.1 Adoption tools

Consultant will identify tools and process improvements including template staff reports, findings for local governments to use when submitting Building codes to the California Building Standards Commission for acceptance, a proposed reach code template presentation for use by city staff in presenting to their City Councils, and/or other tools needed to move forward with the code adoption process at the local government and State level.

Deliverable: List of Adoption Tools to develop with descriptions.

2.4.2 Implementation tools

Consultant will identify tools and process improvements including building department training, permit and inspection streamlining (including documenting all process efficiencies and benefits enabled by a suite of proposed streamlined permitting processes), builder community education, and/or other tools needed to move forward with the code implementation process.

Recommendations and deliverables will be developed with and vetted by the interested municipalities, development community, and other key stakeholders.

Deliverable: List of Implementation Tools to develop with descriptions.

2.5 Provide Technical Assistance to Municipalities for Adoption

For each jurisdiction within EBCE's service territory, the Consultant will provide technical assistance to interested municipalities for the adoption process, including developing the tools mutually agreed to in Task 2.4. In addition, the technical assistance may include presenting to or answering technical questions at Commission and City Council meetings and preparing and submitting Building Electrification codes to the California Energy Commission for approval. EBCE has a total of 12 jurisdictions within its service territory.

In advance of Task 2.5, EBCE will identify jurisdictions planning to participate and estimate the level of support required to implement Tasks 2.5 and 2.6.

Deliverables:

- 1. Produced Adoption Tools (template documents, etc.).
- 2. Presentation slides delivered to municipalities.
- 3. California Energy Commission submission packets.
- 4. List of cities and assistance provided.

2.6 Provide Technical Assistance to Municipalities for Code Implementation

For each jurisdiction within EBCE's service territory, the Consultant will provide technical assistance to municipalities for the implementation process, including developing the tools mutually agreed to in Task 1.4, and which may include staff training sessions as requested.

Deliverables:

- 1. Produced Implementation Tools (template documents, etc.).
- 2. Training slides delivered to municipalities.
- 3. List of cities and assistance provided.

3 <u>Schedule</u>

All work must be completed with sufficient time for municipalities to adopt ordinances in 2019 that will go into effect on January 1, 2020. Code implementation and process streamlining under Task 2.6 may carry over to December 31, 2020 with EBCE prior written approval.

Deliverable: Consultant to provide a detailed schedule within 2 weeks of execution of this Agreement.

Task 4: Kitchen Electrification Grant Program Support

TRC will provide support to EBCE on implementation of a commercial kitchen electrification grant program, including supporting 1-5 commissary kitchens in the EBCE service area with partial or full electrification. Compensation for this task is not to exceed \$20,000, unless fund shifting from other categories is approved in writing by EBCE staff.

Task 4a: Assist EBCE in design of grant applications.

TRC will review the package of grant documents prepared by EBCE to help improve the quality of responses EBCE will receive. This input includes review of application requirements, eligibility criteria, evaluation criteria, and suggestion of any additional application requirements to aid and assist in EBCE's review and selection of strong applicants. This task is to be billed at fixed fee as outlined in Exhibit E.

Task 4b: Assist in review of applications.

TRC will assist EBCE in review and selection of applicants. This review will include an assessment of project feasibility, additional potential opportunities, adequacy of proposed project budget, and potential of the project for advancing the awareness of and receptivity to induction cooking. This task is to be billed as fixed fee as outlined in Exhibit E.

Task 4c: Serve as technical consultant to EBCE and grantees to help achieve success with selected projects.

TRC will be available on a time and materials basis as a resource to both EBCE and grantees to help achieve success with selected projects. This support could include identifying possible technology solutions for transitioning existing equipment to all-electric alternatives, reviewing feasibility of integration electrification-related improvements into existing building infrastructure, bill analysis to understand bill implications of kitchen electrification, and other technical matters related to the electrification component of the projects.

Task 4d: Assist in development training materials for chefs for technology installed at grantee kitchens.

TRC will support development of training materials to be left behind at the selected project sites to facilitate new chef orientation to installed equipment. Materials left with grantees should include spec sheets and warranty information for any selected and installed technology and a fact sheet providing specific tips for success when working with new electric alternatives to previous equipment.

Additional Services:

Consultant will not provide additional services outside of the services identified in this <u>Exhibit A</u>, unless it obtains advance written authorization from the EBCE contract administrator or lead EBCE representative prior to commencement of any additional services.

Exhibit C

Consultant Staffing

TRC Engineers, Inc. anticipates the following staff to provide services on this contract:

Staff Title	Staff Name
Vice President	Catherine Chappell, Abhijeet Pande
Associate Vice	Carmen Henrikson, Katie Wilson
President	
Director	Gwelen Paliaga, Colman Snaith
Associate Director	Michael Mutmansky, Marian Goebes
Senior Project	Nicholas Dunfee
Manager	
Project Manager	Farhad Farahmand, Rupam Singla, Dove Feng, David
	Douglass-Jaimes
Associate Project	Avani Goyal, Yamini Arab, Parul Gulati, Robert Mason
Manager	
Associate	Neil Perry, Nolan Stephens, Mia Nakajima, Richard Williams,
	Josiah Norton, Kristen Bellows, Amanda Pinion
Assistant	Yolanda Beesemyer

<u>Exhibit E</u>

Compensation/Budget

In consideration of the services provided by Consultant described in <u>Exhibit A</u> and subject to the terms of the Agreement, EBCE shall pay Consultant based on the following fee schedule and terms:

- 1. Time and materials up to \$70,000 cost for administration, model code development and cost-effectiveness analysis (Task 1, Tasks 2.1 through 2.4, tasks 4c-4d).
- 2. Time and materials up to \$145,050 for adoption and implementation support (Tasks 2.5 and 2.6). These costs are not to exceed \$10,000 per municipality unless approved in writing by EBCE.
- Tasks 4a and 4b are to be billed at a fixed cost. 4a to be billed at \$1,200 / review and task 4b to be billed at \$750 / application reviewed. Fixed costs are not to exceed \$4,950, unless otherwise approved in writing by EBCE.
- 4. Billings will utilize the rate schedules below as fully loaded rates. Nonlabor expenses are not to exceed \$3,500.
- 5. Invoices may be submitted no more than monthly, with hours identified by staff, rate and task.

Staff Title	Staff Name	Rate (\$/Hour)
Vice President	Catherine Chappell, Abhijeet Pande	\$275
Associate Vice	Carmen Henrikson, Katie Wilson	\$230
President		
Director	Gwelen Paliaga, Colman Snaith	\$215
Associate	Michael Mutmansky, Marian Goebes	\$190
Director		
Senior Project	Nicholas Dunfee	\$175
Manager		
Project	Farhad Farahmand, Rupam Singla, Dove Feng,	\$160
Manager	David Douglass-Jaimes	
Associate	Avani Goyal, Yamini Arab, Parul Gulati, Robert	\$115
Project	Mason	
Manager		
Associate	Neil Perry, Nolan Stephens, Mia Nakajima, Richard Williams, Josiah Norton, Kristen Bellows, Amanda	\$105
	Pinion	
Assistant	Yolanda Beesemyer	\$90

TRC Rate Schedule



Consent Item 9

TO:East Bay Community Energy Board of DirectorsFROM:Beckie Menten, Program Manager, Building Electrification and Energy
EfficiencySUBJECT:First Amendment to Consulting Services Agreement with Recurve
AnalyticsDATE:June 17, 2020

Recommendation

Adopt a Resolution authorizing the CEO to execute a First Amendment to the Consulting Services Agreement with Recurve Analytics ("Recurve") to extend the term, to add to the scope, and to increase the compensation by \$125,000, for a total amount not to exceed \$225,000.

Background and Discussion

Background

In October of 2019, EBCE contracted with Recurve Analytics to develop a baseline across EBCE service area, against which to evaluate the potential for time based pay for performance demand side programs. EBCE staff hopes to discover through pay for performance pilots if the procurement of flexible demand side resources can offer a strategy for achieving lower cost carbon free resources during evening peak periods. The initial work product with Recurve focused around developing a baseline against which to calculate program impacts; now EBCE staff seeks to extend the term of the Agreement with Recurve to support implementation of three pay for performance pilots. These pilots will help EBCE understand the feasibility of achieving evening peak reductions from energy efficiency strategies and would help inform and support a potential funding application to the California Public Utilities Commission.

EBCE is working towards achieving a 100% free electricity portfolio; to achieve this strategy cost-effectively, EBCE must find solutions for low-cost, carbon-free resources available during evening peak hours, i.e. 3-9 pm. Currently, there is a mismatch between supply of renewable power and demand for power during the evening peak period. Renewable power is most abundant in the middle of the day, whereas energy consumption peaks in the evening hours - this results in renewable power facilities being curtailed and a dearth of carbon free resources in the evening hours.

EBCE has an interest in exploring the ability to procure demand side resources, specifically flexible load resources, as a strategy for reducing carbon emissions and achieving rate stability. Flexible load demand side resources refer to behind the meter strategies to not only reduce overall energy use, i.e. energy efficiency improvements, but also targets load shifting strategies deployed at the same time. This could include, for example, an HVAC system retrofit that incorporates a programmable thermostat, enabling a homeowner to save money on bills through overall efficiency, but also maximize savings to time of use billing by "pre-cooling" a home during midday hours when renewable energy is cheap and abundant, and avoiding cooling during evening peak hours, when energy is expensive and more carbon intensive. Incorporating both overall baseload reduction as well as the ability to shift when energy demand occurs can help to minimize the overall need for energy procurement during expensive and carbon intensive evening hours (figure 1).



Figure 1. Image source: Harald Schütt for pv magazine. Data from CASIO.

There are many potential benefits associated with procuring these resources from behind the meter strategies, versus front of the meter storage. Energy efficiency offers many benefits to participants, including increased comfort and safety, reduced bills, and higher quality of life. Energy efficiency programs also stimulate the local economy, both through saving customers money on bills -which can be spent elsewhere - and through local workforce development. The Local Development Business Plan cites energy efficiency as one of the best local job creators, estimating to generate around 10.5 jobs per million dollars invested¹. This pilot has the potential to generate over 7 new, local jobs; at scale, it could generate as many as 50.

EBCE Pilot Procurements

Starting in Q4 2019, Recurve imported EBCE customer data into their data analytics platform. Based on protocols defined in the CalTRACK² methodology, the Recurve dashboard then allows EBCE to select certain subsegments of the population to determine the potential benefits associated with demand side programs. Based on this, EBCE staff have designed three potential pilots targeting evening peak reductions. Programs are designed to test: 1) market readiness for P4P programs, 2) the Recurve platform for customer targeting and program operation and, 3) the ability of P4P programs to encourage "flexible load," targeting savings during the evening ramp and evening peak hours (4-9 pm). The three pilots target single-family residential, commercial, and low-income residential customers. The proposed pilots are intended to run for one year, allowing EBCE to test both the logic of the program as well as the performance of contractors. If the pilot successfully delivers cost effective savings and demonstrates market readiness for programs that pay incentives on meter based savings, the program can be to be scaled. A scaled program would focus on much larger customer participation targets and would pay for incentives over a longer time frame, such as 3-5 years; this would increase interest and participation among contractors and improve the reliability and persistence of the energy savings.

The Recurve platform enables these pilots by offering a measurement and verification function. By developing a baseline from EBCE customer data, the Recurve platform calculates the expected energy consumption in the absence of any interventions, i.e. an energy efficiency retrofit. Once a project has been performed on a property, the actual meter based consumption post-project is compared against the expected baseline in a "no-intervention" scenario, and the difference is attributed to the project and becomes the basis of incentive payments. In other words, pay for performance programs pay an incentive based on the program impacts as measured at the meter.

¹ Local Development Business Plan at p. 72. (<u>https://ebce.org/wp-content/uploads/Local-Development-Business-Plan_FINAL_7-12-18_hi-res.pdf</u>)

² CalTRACK refers to a methodology for establishing a baseline and measuring impacts of a project at the meter. This methodology was developed as an open source framework and has been adopted by the California Public Utilities Commission for use in ratepayer funded pay-for-performance pilots. http://www.caltrack.org/#:~:text=What%20is%20CalTRACK%3F,intervention%20had%20not%20taken%2 Oplace.

An additional benefit of this platform is that is allows EBCE to do targeted outreach for these pilots, ensuring that the participants targeted for the program are those who may benefit the most from the program. Figure 2 shows the load shape of customers selected for the program (in this case, single family residential), as compared to the total population of customers and the "non-selected" customers. This figure demonstrates that the targeted customers use significantly more energy in the evening peak time frame; thus, these are customers who would benefit the most from a program targeting evening peak.



Figure 2. Load Shapes of Targeted vs. Non-Targeted Customers. Targeted customers are represented in blue, nontargeted customers are in gray, and the total population is in dashed red.

Recommendation

EBCE is pleased with the outcome of Recurve's work product from Phase I of the effort. Recurve has been very involved in development of the CalTRACK methodology, an open source methodology used for determining savings from meter based programs in California Public Utilities Commission programs. This situates EBCE well to draw from outcomes of this pilot to inform a potential CPUC funding request. If the pilot succeeds at delivering savings cost effectively and demonstrates that contractors are comfortable with pay for performance programs that measure savings at the meter, then it would be recommended to scale the program. The ultimate goal is to pursue administration of ratepayer funded energy efficiency programs at the California Public Utilities Commission. EBCE staff estimates a potential annual budget of \$5 million / year could be available to support EBCE administered energy efficiency programs. These funds could be used to establish a program framework and provide incentive payments for the energy efficiency funding is July 1, 2021 - coinciding with the 21-22 fiscal year.

The pilot will also provide an important foundation upon which to build into an eventual ratepayer funded EE program. This pilot will help EBCE prepare for a filing for, and eventual implementation of, an EE program in three ways: first, it will reduce time necessary to draft cost effectiveness filings by providing empirical data to populate these filings (this includes values such as measure costs, participation

levels, appropriate incentive levels, and savings forecasts.) Secondly, it will provide a data-based foundation upon which these filings are based, which is likely to reduce stakeholder protests and supplemental filing requests, overall shortening the regulatory cycle from filing to draft Resolution. Finally, it will ensure the program is "launch ready" from day one. Frequently when first time administrators propose EE filings, there is significant guesswork involved around assumptions such as participation rates, what incentive is necessary to move the market, and the need to build awareness of programs in the contractor community. Having some certainty and program experience before going into the filing will help to ensure that forecasted results are much more closely aligned to actual results, helping EBCE to deliver on goals and allowing for any "wiggle room" provided to first time implementors to be applied towards innovation and improvement, rather than trueing up to initial estimates. This will help to solidify EBCE's status as a leader in innovation and implementation right out of the gate.

EBCE staff therefore seek to amend the existing Agreement with Recurve to move into Phase II of this effort, namely the implementation of pay for performance pilots. In developing this recommendation, staff has reviewed the EBCE Administrative Procurement Policy; as Recurve is a preferred vendor resulting from a Southern California Public Power Authority Request for Qualifications for a substantially similar scope of work, the policy allows for EBCE to engage Recurve for these services under Section VII. Cooperative Purchasing (2.).

The proposed Amendment proposes to increase compensation by \$125,000, with an overall not to exceed amount of \$225,000, amends the contract to introduce additional scope, and extends the term to December 31, 2021.Additions to the Scope include refreshing the data in the Recurve dashboard to the most recently available, introduction of new EBCE member agencies into the platform, and measurement and verification services to support the implementation of pay for performance pilots. The amendment also adds a contingency budget of \$25,000 (included in the total above) for additional staff support and platform processing hours.

Fiscal Impact

The proposed amendment will be funded from an existing allocation in the Local Development Budget.

Attachments

- A. Resolution Authorizing the CEO to Execute a First Amendment to the Agreement with Recurve Analytics
- B. First Amendment to Consulting Services Agreement with Recurve Analytics

RESOLUTION NO.

A RESOLUTION OF THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY AUTHORIZING THE CEO TO EXECUTE A FIRST AMENDMENT TO THE AGREEMENT WITH RECURVE ANALYTICS TO PROVIDE TECHNICAL SERVICES FOR ENERGY EFFICIENCY PROGRAMS

WHEREAS The East Bay Community Energy Authority ("EBCE") was formed as a community choice aggregation agency ("CCA") on December 1, 2016, Under the Joint Exercise of Power Act, California Government Code sections 6500 *et seq.*, among the County of Alameda, and the Cities of Albany, Berkeley, Dublin, Emeryville, Fremont, Hayward, Livermore, Piedmont, Oakland, San Leandro, and Union City to study, promote, develop, conduct, operate, and manage energy-related climate change programs in all of the member jurisdictions. The cities of Newark and Pleasanton, located in Alameda County, along with the City of Tracy, located in San Joaquin County, were added as members of EBCE and parties to the JPA in March of 2020.

WHEREAS in October 2019 EBCE entered into a Consulting Services Agreement with Recurve Analytics ("Recurve") to develop a baseline for time-based demand side resource procurement programs and to provide technical energy efficiency consulting services.

WHEREAS there is a desire to pilot procuring cost-effective demand-side resources through pay for performance programs in the EBCE service area.

WHEREAS Recurve possess the skill, experience, ability, background, and knowledge to provide measurement and valuation services to support implementation of time based pay for performance pilots.

WHEREAS EBCE seeks to amend the existing Agreement with Recurve to increase the compensation by \$125,000 for a total not-to-exceed of \$225,000, add additional scope to support the implementation of pay for performance pilots, and extend the Agreement term to December 31, 2021.

NOW, THEREFORE, THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY DOES HEREBY RESOLVE AS FOLLOWS:

<u>Section 1.</u> The Board of Directors hereby authorizes the CEO to negotiate and execute an Amendment to the Consulting Services Agreement with Recurve Analytics to increase the total compensation by \$125,000, increase the total not-to exceed to \$225,000, to add new scope, and extend the term of the Agreement, in substantially the form attached hereto, with clarifying or clerical changes proposed by the CEO and approved by the General Counsel.

ADOPTED AND APPROVED this 17th day of June 2020.

Dan Kalb, Chair

ATTEST:

Stephanie Cabrera, Clerk of the Board

First Amendment to Consulting Services Agreement by and Between East Bay Community Energy Authority and Recurve Analytics

This First Amendment to the Agreement with Recurve Analytics for Consulting Services ("First Amendment") is made this 17 of June, 2020, by and between the East Bay Community Energy Authority, a Joint Powers Agency formed under the laws of the State of California ("EBCE") and Recurve Analytics ("Recurve"), a Delaware corporation, for the purposes of adding additional compensation, adding additional services to the scope, and extending the termination date.

Recitals

- A. EBCE and Recurve entered into that certain Consulting Services Agreement dated 15th October 2019 ("Agreement"), wherein Recurve agreed to provide data analytics and energy efficiency ("EE") baseline assessment services to EBCE, with compensation not to exceed \$100,000.
- B. EBCE and Recurve now desire to amend the Agreement to add additional compensation, increasing the not-to-exceed amount by \$125,000 to cover platform and program management support for pay for performance pilot programs for a total amount not to exceed \$225,000, to add additional services to the scope of services including refreshing the data in the Recurve platform and providing platform support and measurement and valuation services for pay for performance pilots, and to extend the term of the Agreement.

Now therefore, for good and valuable consideration, the amount and sufficiency of which is hereby acknowledged, the Parties agree as follows:

- 1. Section 2 of the Agreement ("Term") is amended to extend the term of the Agreement through December 31, 2021.
- 2. Section 6.1 of the Agreement ("Compensation and Payment") is amended in part to increase the total compensation of the Agreement by \$125,000 for a total amount not to exceed two hundred thousand five dollars (\$225,000).
- 3. Exhibit A ("Scope of Work") of the Agreement is replaced in its entirety by Exhibit A, attached hereto.
- 4. Exhibit B ("Schedule") of the Agreement is replaced in its entirety by Exhibit B, attached hereto.
- 5. Exhibit E ("Compensation/Budget") of the Agreement is replaced in its entirety by Exhibit E, attached hereto.
- 6. Section 7.2 of the Agreement ("Records Retention and Ownership of Work Product") is amended as per below.

- a. The last sentence of 7.2 reads, "EBCE agrees that the Work Product will not be disclosed, sold, assigned, leased or otherwise disposed of or made available to third parties by EBCE, commercially exploited by or on behalf of EBCE, nor provided or made available to any other party without written authorization."
- b. That sentence is deleted in its entirety, and the following language is appended to 7.2:

"7.2(a) For any purposes other than pursuing and exploiting the energy efficiency and procurement opportunities within the EBCE Service territory and otherwise enjoying the intended benefits of the Work Product, EBCE agrees that the Work Product will not be disclosed, sold, assigned, leased or otherwise disposed of or made available to third parties by EBCE, commercially exploited by or on behalf of EBCE, nor provided or made available to any other party without written authorization.

7.2(b) For the purposes of pursuing and exploiting the energy efficiency and procurement opportunities within the EBCE Service territory and otherwise enjoying the intended benefits of the Work Product, EBCE agrees that the Work Product will not be sold, assigned, or leased by EBCE, or commercially exploited by or on behalf of EBCE. For these purposes, EBCE can make Work Products available to third parties without written authorization but must provide notification to Recurve within thirty (30) days. EBCE cannot make Recurve software available to third parties for any purpose."

7.2(c) Notwithstanding the above, it is understood that EBCE is subject to the California Public Records Act (Gov. Code § 6250 et seq.). If a request under the California Public Records Act is made to view the Work Product, EBCE shall notify Recurve of the request and the date that such records will be released to the requester unless Recurve obtains a court order enjoining that disclosure. If Recurve fails to obtain a court order enjoining that disclosure, EBCE will release the requested information on the date specified and EBCE shall have no liability regarding such release. Section 12.6 of the Agreement is added to the agreement. Section 12.6 reads:

> "EBCE agrees, at its sole cost and expense, to indemnify, defend, with counsel reasonably approved by Consultant, protect and hold harmless Consultant, its officers, directors, employees, agents, attorneys, designated volunteers, successors and assigns, and those Consultant agents serving as independent contractors in the role of Consultant staff (collectively "Consultant Indemnitees") from and against any and all damages, costs, expenses, liabilities, claims, demands, causes of action, proceedings, penalties, judgements, liens and losses of whatever nature that arise, directly or indirectly, in whole or in part, out of or are in any way related to EBCE's unauthorized release of any personally identifiable information
contained in Consultant Work Product as described in Sections 7.2(a) and 7.2(b)."

7. All other terms and conditions in the Agreement not otherwise modified by this First Amendment will remain in full force and effect.

In witness whereof, the Parties have entered this Amendment on the date written above.

East Bay Community Energy Authority, A Joint Powers Authority	Recurve, A Recurve Analytics
Nick Chaset Chief Executive Officer	Name: Title:
Date:	Date:
Approved as to form:	Name:
	Title: Date:
EBCE General Counsel	

<u>Exhibit A</u>

Scope of Work

I.Scope of Work

Recurve will make its platform available until October 31, 2021. Platform will access customer consumption records and other data agreed upon by both parties to support development and deployment of pilot programs as described below. Data outputs will be stored in Recurve's Data Warehouse and will be made available to EBCE through the Recurve dashboard(s), the Recurve API, and by database transfer (copy of the Recurve dataset back to the EBCE Google Cloud Platform BigQuery project, or other similar mechanism).

Task 1. Data Refresh: Recurve will refresh its existing territory-wide analysis of EBCE's building stock, using all available 2019-2020 data, and update its data warehouse tables as well as its "Portfolio Summary and Analytics" and "Targeting Analytics" dashboards.

Recurve utilizes Google Cloud Storage and related tools for raw data storage and cloud-based data science. Recurve will upload Item 17 (or equivalent) consumption data provided by EBCE corresponding to the electric and natural gas meters in the EBCE service territory. EBCE will provide the data in the same structure and method as it has employed in the initial Scope of Work.

Consumption data is geocoded and mapped to building sites, which are in turn mapped to known DERs. This mapping allows Recurve to create longitudinal profiles of the installed DER base in the service territory. The relational database structure provides the foundation for subsequent analytics.

Recurve fits building energy models to all sites within the service territory following CalTRACK protocols (docs.caltrack.org). First, the relationship between changes in outdoor temperature and changes in energy consumption (heating and cooling) are determined for both daily and hourly increments in all buildings. Along with these weather factors, this method produces useful model error metrics that provide visibility into buildings with erratic energy consumption. Additional metrics like peak to shoulder ratio, summer and winter consumption, gross consumption, and load shape profiles provide the basis for customer targeting for future DER procurements.

Deliverable: Recurve dashboard will be updated with 2019 data and inclusive of new EBCE communities (cities of Pleasanton, Newark, Tracy).

Task 2: Resource Planning for P4P Pilots

The same data that is used to create the baseline calculations, as well as the targeting profiles, will be used to operationalize procurements for load resources. Data that was used for planning will be repurposed to support pay for performance pilot implementation, including through establishing baselines, quantifying program impacts, and targeting customers for participation in the pilots. Recurve, in conjunction with EBCE staff, will develop a measurement and verification (M&V) strategy and related guidance for determining net impact to load shape and associated incentive payment obligations to contractors.

Deliverables:

• M&V plan for three P4P pilots

Task 3: Fleet Management: Recurve will provide program tracking for three pilot programs. The savings results for these programs will be updated no less frequently than monthly and a final savings value will be calculated for each participant based on the program design either at the end of the program reporting period or at the point of termination of this Scope of Work. Recurve will track customer enrollment to determine if contractors are to receive one-time "kicker" incentives agreed upon by the parties (e.g. enrollment bonuses.)

Deliverables:

- Monthly status reports (including enrolled projects, contractors, and eligible enrollment bonuses)
- Final report on program impacts, including at a minimum program energy savings with savings during peak periods (3-9 pm) being separately reported, total incentive payment, total participating contractors, total projects
- Transfer of enrollment data back to EBCE on a monthly basis via the EBCE Google Cloud Platform, or other similar mechanism.

Task 4. Customer Success Package: Recurve will provide up to 100 hours of Consultant Staffing time to support EBCE, covering both Resource Planning and Fleet Management.

III. Consultant Staffing

Matt Golden McGee Young Adam Scheer Natalie Kozlowski Mariano Teehan

<u>Exhibit B</u>



<u>Exhibit E</u>

Compensation/Budget

IV. <u>Compensation/Budget</u>

- 1. Recurve Platform
 - a. \$5,000/month, billable on a quarterly basis, prepayment starting the month following the end of the current agreement term
- 2. Resource Planning
 - a. Refresh of Customer Baseline/Targeting analysis = \$16,000 billable on a deliverable basis.
- 3. Customer Success
 - a. 100 hours = \$24,000 billable on an annual prepayment
- 4. Contingency Budget: \$25,000
 - a. Support hours above 100 available at \$275/hour as accrued.
 - b. vCPU hours above 5,000 per month available at \$1.18/vCPU hour as accrued



Staff Report Item 10

TO: East Bay Community Energy Board of Directors

FROM: Nick Chaset, Chief Executive Officer

SUBJECT: CEO Report (Informational Item)

DATE: June 17,2020

Recommendation

Accept Chief Executive Officer (CEO) report on update items below.

Executive Committee Meeting

The Executive Committee met on May 22, 2020 and CEO Chaset presented the FY 2020-2021 Draft Budget for the Committee to review. Marie Fontenot, the Senior Director of Power Resources, gave an update on the Integrated Resource Plan. The next meeting will be held on Friday June 26, 2020 at 12 pm.

Finance, Administration and Procurement Subcommittee Meeting

The Finance, Administration and Procurement Subcommittee met on May 15, 2020 and reviewed the Draft Budget for FY 2020-2021. The next meeting will be held on Friday June 12, 2020 at 3 pm.

Marketing Regulatory and Legislative Subcommittee Meeting

The Marketing, Regulatory and Legislative Subcommittee met on May 21, 2020 and received updates from both the Marketing and Account Services Team and the Policy Team. There will be no meeting held in June or July. The next one will be held on Thursday August 6, 2020 at 11 am.

Marketing and Account Services Update

May Rate Updates

The billing systems were updated since the last Board meeting to reflect changes to the PCIA and PG&E generation dates. Updated rate sheets are available at <u>https://ebce.org/rates/</u>. PG&E is currently working on the Joint Rate Comparison with these updated rates. Once received, that comparison will also be posted to the Rates webpage. **Promoting Customer Financial Assistance through CARE**

In early May, EBCE staff launched several resources and pushed communications related to the California Alternate Rates for Energy Program, also known as CARE. The state-funded program provides a discount of 35% on the electricity bill for low-income and qualified households. The resources included a landing page at https://ebce.org/care/ with a brief tutorial video, Frequently Asked Questions, and an easily accessible link to apply for the program. Staff facilitated public webinars about CARE on June 9, 2020 (English) and June 11, 2020 (Spanish).

As of June 8, the dedicated page has had 2,830 pageviews with over a 7% click through rate to the enrollment application. EBCE has also conducted outbound calls to an initial pilot of 100 customers to encourage participation in the CARE program, with a 50% connection rate and reported enrollment rate of 5% of the total calls, 10% of customers successfully reached. A follow-up pilot of 200 customers is currently underway.

Below is are charts showing the total enrollment each week for the last six months and the percentage change week over week for enrollment in CARE and FERA programs. Total enrollment in these programs has increased by 10% from the beginning of Shelter-in-Place through the month of May.





We will continue to push this message out to local residents through additional digital advertising, social media, emails, and direct mail. Staff will report on on-going work and results at future meetings.



Staff Report Item 13

- TO: East Bay Community Energy Board of Directors
- FROM: Annie Henderson, VP of Marketing and Account Services
- SUBJECT: Extension of Temporary Revision to Terms and Conditions Section "Failure to Pay" and Temporary Suspension of Delinquent Accounts and Collections Policy (Action Item)

DATE: June 17, 2020

Recommendation

Adopt a Resolution to Extend the Temporary Suspension of Certain Customer Terms and Conditions and Administrative Policies for an Additional Period of 90 Days During the COVID-19 Pandemic, and extend the suspension of the "Failure to Pay" section of the Service Terms and Conditions of the Delinquent Accounts and Collections Policy for an additional 90 days.

Background and Discussion

On March 18, 2020, the Board approved a resolution temporarily increasing the signing authority of the chief executive officer for a period of 90 days to maintain operations during the COVID-19 pandemic. The resolution also included the following:

<u>Section 2.</u> For a period of 90 days, the Board hereby authorizes the CEO to suspend the following customer terms and conditions and administrative policies:

- (a) Customer terms and conditions related to 'Failure to Pay' and
- (b) Delinquent Accounts and Collections Policy.

That period of 90 days will end on June 16, 2020.

EBCE would like to extend the suspension of the "Failure to Pay Section" of the Service Terms and Conditions of the Delinquent Accounts and Collections Policy for an additional 90 days, through September 14, 2020.

Staff will bring this item back to the Board at its September meeting to determine if another extension is necessary and appropriate.

Financial Impacts

The financial impact is minimal given the limited pool of customers who would be eligible for return to PG&E and for collections.

Attachments

- A. Resolution to Extend the Temporary Suspension of Certain Customer Terms and Conditions and Administrative Policies for an Additional Period of 90 days during the Covid-19 Pandemic
- B. P-2018-4 Delinquent Accounts and Collections Policy

RESOLUTION NO. R-2020 -

A RESOLUTION OF THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY TO EXTEND THE TEMPORARY SUSPENSION OF CERTAIN CUSTOMER TERMS AND CONDITIONS AND ADMINISTRATIVE POLICIES FOR AN ADDITIONAL PERIOD OF 90 DAYS DURING THE COVID-19 PANDEMIC

WHEREAS The East Bay Community Energy Authority ("EBCE") was formed as a community choice aggregation agency ("CCA") on December 1, 2016, Under the Joint Exercise of Power Act, California Government Code sections 6500 *et seq.*, among the County of Alameda, and the Cities of Albany, Berkeley, Dublin, Emeryville, Fremont, Hayward, Livermore, Piedmont, Oakland, San Leandro, and Union City to study, promote, develop, conduct, operate, and manage energy-related climate change programs in all of the member jurisdictions. The cities of Newark and Pleasanton, located in Alameda County, along with the City of Tracy, located in San Joaquin County, were added as members of EBCE and parties to the JPA in March of 2020.

WHEREAS, in December 2019, an outbreak of respiratory illness due to a novel coronavirus (a disease now known as COVID-19), was first identified in Wuhan City, Hubei Province, China, and has spread outside of China, impacting more than 75 countries, including the United States; and WHEREAS, since the federal Centers for Disease Control and Prevention ("CDC") confirmed the first possible case of community spread of COVID-19 in the United States on February 26, 2020, there has been a significant escalation of United States domestic cases and deaths from COVID-19; and

WHEREAS, on March 4, 2020, Gavin Newsom, Governor of the State of California, proclaimed a state of emergency to exist in California due to the spread of COVID-19; and

WHEREAS, on March 13, 2020, the President of the United States declared a national state of emergency due to COVID-19; and

WHEREAS, as of June 1, 2020, the California Department of Public Health reported a total of 113,006 cases and 4,251 fatalities in the state; and,

WHEREAS, as of June 2, 2020, the State of California remains under Executive Order to Shelter in Place, with some limited re-opening of businesses; and

WHEREAS, many EBCE customers continue to face ongoing economic hardships as fall out the COVID-19 global health pandemic; and

WHEREAS, on March 18, 2020, the Board of Directors resolved to temporarily suspend, for a period of 90 days, certain customer terms and conditions and administrative policies associated with collections and customer payments; and

WHEREAS, EBCE desires to extend the temporary suspension of certain customer payment obligations.

NOW, THEREFORE, THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY DOES HEREBY RESOLVE AS FOLLOWS:

<u>Section 1.</u> For a period of an additional 90 days, the Board hereby authorizes the CEO to suspend the following customer terms and conditions and administrative policies:

(a) Customer terms and conditions related to 'Failure to Pay' and

(b) Delinquent Accounts and Collections Policy.

<u>Section 2.</u> Unless extended by the Board, this Resolution shall expire and be of no further force or effect 90 days after the date of its adoption (September 14, 2020), at which point the approve terms and policy shall resume to be effective.

ADOPTED AND APPROVED this 17th day of June 2020.

Dan Kalb, Chair

ATTEST:

Stephanie Cabrera, Clerk of Board



POLICY NO. 2018 - 4 Delinquent Accounts and Collections

Agenda: February 7, 2018 Item No.: 7 Resolution: No Resolution associated with policy

Subject: Delinquent Accounts and Collections Policy

Policies:

Delinquent Accounts:

Pursuant to Electric Rules 8 and 11, Pacific Gas & Electric (PG&E) uses the following process to determine past due accounts and the necessary action:

Residential Accounts		Non-Residential Accounts	
Day 1 - Issuance of Bill	Customer Receives Bill	Day 1 - Issuance of Bill	Customer Receives Bill
Day 22	Past Due	Day 18	Past Due
Day 27 - 33	15 Day Notice on Next Bill	Day 21	7 Day Notice Delivered
Day 41-47	48 Hour Notice via Mail	Day 29	24 Hour Phone Call or In Person Notice
Day 45 - 51	24 Hour Phone Call or In Person Notice	Day 32	Account is Eligible for Disconnection
Day 50-56	Account is Eligible for Disconnection		

Aging Accounts:

All EBCE accounts, whether Residential or Non-Residential, identified in the month aging accounts receivable report, as provided by PG&E, with outstanding balances over 90 days or more are eligible to be returned to PG&E.

EBCE Residential customer accounts exceeding \$250 in charges overdue for more than 90 days will be sent a late payment notification by EBCE. The customer will be provided 60 days to pay or make payment arrangements. If payment in full is not received within 60 days from the date of notification, or the terms of an activated payment arrangement are not fulfilled, the EBCE customer account may be closed and returned to PG&E bundled generation service on the next account meter read date. Residential customers returned to PG&E will be charged the applicable EBCE opt-out fee.

EBCE Non-Residential customer accounts exceeding \$500 in aggregate in unpaid charges for 60 days or more will be sent a late payment notification by EBCE. The customer will be provided 30 days to pay or make payment arrangements. If payment in full is not received within 30 days of the date of notification, or the terms of an activated payment arrangement are not fulfilled, the EBCE customer account may be closed and returned to PG&E bundled generation service on the next account meter read date. Non-residential customers returned to PG&E will be charged the applicable EBCE opt-out fee.

Collections:

Closed East Bay Community Energy accounts with overdue amounts greater than \$100 may be referred to a collection agency. Amounts \$100 or less may be written off. PG&E may close customer accounts before payment delinquencies bring them to the attention of EBCE operations. When PG&E closes customer accounts, these accounts are also closed in the EBCE program. In these cases, the thresholds outlined in the preceding paragraphs apply in either referring closed accounts to collections or writing off balances.



Staff Report Item 14

TO: East Bay Community Energy Board of Directors
 FROM: Annie Henderson, VP Marketing and Account Services
 SUBJECT: Enrollment of New Communities: Pleasanton, Newark, and Tracy (Informational Item)
 DATE: June 17, 2020

Recommendation

Receive information report from staff on proposed enrollment of customers within the new member communities of Pleasanton, Newark, and Tracy during the month of April 2021.

Background and Discussion

The cities of Pleasanton, Newark, and Tracy passed resolutions to join the East Bay Community Energy Joint Powers Authority during the Fall of 2019. The Board approved these jurisdictions for membership in the JPA and expansion of our service territory at the November 20, 2019 regular meeting. Staff prepared and submitted an addendum to the Implementation Plan to include new communities to the California Public Utilities Commission on December 20, 2019 and received approval through a certification letter from the CPUC on March 9, 2020. New board members were sworn in during the regular meeting on April 22, 2020.

The certification letter from the CPUC allows enrollment of customers within the new member jurisdictions beginning in 2021. A number of items were taken into account to develop a proposed timeline for enrollment during April 2021. These considerations included:

- Notifications
 - Timing of enrollment notifications
 - Other customer notifications (Joint Rate Mailer, Power Content Label, Time-of-Use transition)
- Time-of-Use rate default transition within Alameda County
- Commercial: March 2021, Residential: May 2021
- Procurement Costs
 - Seasonal cost of power
 - Cost of resource adequacy
- Customer Experience and Rates
 - Cost to customers/ Seasonal rates

- Timing of first bill
- Assumptions on timing of PG&E rate changes (typically in January)
- EBCE Annual Rate Setting (typically in June)
- Agency Finances
 - Impacts to revenue and expenses
 - Impacts to margin

Based on analysis of these factors, staff recommends enrolling the new communities in April 2021. Customers are typically enrolled according to their individual meter-read dates and PG&E billing cycles. This means electric accounts within the new cities will transfer to EBCE service (unless they opt out) over the course of the month (as opposed to all at once). New customers will then receive their first bill with EBCE charges in May-June. Staff will present a marketing and outreach plan at a future Board meeting.

Fiscal Impact

The costs and projected revenue associated with the new communities was included in the budget report to the Board on May 24, 2020, as shown in Attachment B.

Attachments

- A. Timeline for Enrollment
- B. New Community Budget and Fiscal Impact

Attachment A



4. New Communities: Newark, Pleasanton, and Tracy

- Inclusion of new communities is expected to have a positive lift on margins based on an April enrollment timeframe estimated at \$3.5MM in net margin based on current market conditions.
 - April-June timeframe represents higher margin months. Full calendar year net margin is lower.
- Positive margin is driven by a modest increase in operating overhead related to serving these customers and softer energy prices.
- Additional analysis is underway to determine the optimal timing based on different customer rates and procurement risks/opportunities



	F1 2020-21	Tiew	riscal Tear
	BUDGET	Communities	Adjusted
REVENUE AND OTHER SOURCES			
Electricity Sales	391,499,000	13,409,000	404,908,000
Uncollectables	(9,787,000)	(201,000)	(9,988,000)
Total Revenue and Other Sources	382,943,000	13,208,000	396,151,000
EXPENSES AND OTHER USES			
ENERGY OPERATIONS			
Cost of Energy	349,917,000	9,065,000	358,982,000
Data Management/Customer Service	7,704,000	299,000	8,003,000
PG&E Service Fees (Billing/Metering)	2,396,000	87,000	2,483,000
Total Energy Operating Costs	360,677,000	9,451,000	370,128,000
OVERHEAD OPERATIONS			
Marketing, Outreach, Communications	1,499,000	274,000	1,773,000
Total Overhead Operating Costs	20,346,000	274,000	20,620,000
NET INCREASE (DECREASE) IN POSITION	1,116,000	3,483,000	4,599,000

FY 2020-21

New Fiscal Year



Staff Memo Action Item 15

TO: East Bay Community Energy Board of Directors
FROM: JP Ross, Sr. Director Local Development, Electrification, and Innovation
SUBJECT: EBCE Resilience Program (Action Item)
DATE: June 17, 2020

Recommendation

- A. Approve the selection of two vendors in response to the Distributed Resource Adequacy (RA) Request for Proposals to deliver RA in the form of Load Modification from behind-the-meter solar and storage installations delivering resilience to EBCE commercial and residential customers.
- B. Authorize the CEO or his designee to negotiate and execute a Load Modifying Agreement with Sunrun Inc. for Load Modification up to 5.75MW/23MWh through December 31, 2032.
- C. Authorize the CEO or his designee to negotiate and execute a Load Modifying Agreement with Enel X in substantially the same form as the Sunrun agreement for Load Modification up to 5MW/20MWh through December 31, 2032.

Background

On November 5, 2019, EBCE, in coordination with Silicon Valley Clean Energy, Peninsula Clean Energy and Silicon Valley Power, issued a competitive solicitation for Distributed Resource Adequacy (RA) Capacity. The solicitation was for a minimum of 10MW of Resource Adequacy from solar and storage systems for EBCE to deliver resilience to single and multi-family residential and commercial customers.

Through the Request for Proposals (RFP), EBCE desired to enter into Resource Adequacy ("RA") contracts with one or more vendors to deliver new behind-the-meter solar and storage systems, or retrofit storage on existing solar systems, on customer properties. However, during the intervening months EBCE has worked with the California Energy Commission to develop an innovative alternative approach to meeting our RA obligation through reducing EBCE's peak demand through the targeted dispatch of behind the meter (BTM) storage systems. This "Load Modifying Resource" will be dispatched on a daily basis during EBCEs highest peak hours, which will decrease EBCE's RA obligation and reduce EBCE's wholesale energy procurement volumes.

The aggregation of distributed behind-the-meter batteries to deliver both Load Modification and resilience is a first of its kind initiative. This program will deliver multiple benefits to our community by adding resilience while also creating jobs and delivering Load Modification to EBCE. Load Modification payments to vendors will be shared with customers to reduce the cost of batteries and encourage customer participation. EBCE will procure Load Modification from these resources for up to ten (10) years. EBCE is working with vendors and the California Energy Commission to make sure that the full value of Load Modification provided is counted.

EBCE will work with selected vendors on acquiring customers for the program, with a goal of increasing resilience during Public Safety Power Shutoffs (PSPS). Systems are required to island from the grid so that homes and businesses will have power during PSPS events. A minimum of 20% of systems will be installed in Disadvantaged Communities (DACs), Low Income Communities and CARE/FERA or Medical Baseline customers properties.

Staff expects the program to deliver resilience to over 1000 residential customers. Commercial customer participation rates are harder to estimate prior to program launch as system sizes are extremely variable based on the nature of the load and site constraints. Staff will report participation rates to the Board during program operation.

Both Sunrun and Enel X have made commitments in line with the Oakland Clean Energy Initiative for local workforce development and hiring practices. This will ensure that jobs are located in Alameda County EBCE will maximize the benefit to local communities by requiring Workforce Development commitments from vendors.

Discussion

On November 5, 2019, EBCE issued an RFP for Distributed RA Capacity. Responses to the RFP were due December 23, 2019. EBCE received bids from 19 vendors, which proposed a range of RA pricing across customer categories. EBCE staff from Local Development, Marketing, Procurement and Operations reviewed each of the proposals. Each bid was evaluated on the basis of competency to perform scope of work, best fit, and price. EBCE then interviewed eight vendors based on the technical evaluation criteria in the RFP shown in Table 1 below, and as described in further detail in Appendix 1.

Category Weighting	Proposal Section(s)
A. Completeness of Response to RFP	Pass/Fail
B. Capacity Requirements & Deployment Timeline	Pass/Fail
C. Technology Safety Certifications & Standards	Pass/Fail
D. Islanding Capability	Pass/Fail
E. Pricing	40
F. Experience Developing Similar Projects & Providing RA Capacity	15
G. Experience & Qualifications of Personnel	10
H. Quality & Detail of Go-to-Market & Customer Engagement Plan	15

J. Local Hiring/Workforce Development Plan 10	s 10
	ppment Plan 10
TOTAL SCORE 100	100

Table 1: Scoring Criteria

Staff interviewed the eight vendors with the highest scoring credible proposals that ensured a cross section of customer categories including; single family and multi-family residential, commercial and health care facilities, would be covered. The two finalists were selected for the price and capacity proposed, their approach to customer engagement, and staff's confidence in their ability to successfully execute the program. The selected portfolio will meet the 20% threshold of priority customers. The vendors are shown in Figure 1 below in according to their final stage in the submission process.



Figure 1: Vendor Selection Funnel

Sunrun and Enel X were selected based on their ability to deliver Load Modification at competitive pricing while delivering customers a compelling value proposition. EBCE is committed to partnering with vendors to deliver resilience across a range of residential and commercial customer segments, while also ensuring 20% of capacity is deployed in disadvantaged and low-income communities, CARE/FERA and Medical Baseline customers premises. Vendors willingness and ability to work with EBCE to develop marketing campaigns

that build awareness of EBCE was a critical important factor as this will be EBCE's first large outbound marketing campaign.

In conjunction with the Load Modification Agreement, EBCE will sign Customer Data Sharing Non-Disclosure Agreements and Co-Marketing Agreements. These agreements will ensure customer data security while defining the rules of engagement with respect to marketing campaigns. EBCE will lead on marketing efforts to our customers, using multiple data sources to identify customers who can benefit from this program. EBCE marketing materials will drive customers to an EBCE hosted website where customers can provide their contact information to express interest in the program. Customers will be validated by SMUD and then passed to EBCE. EBCE will then add additional information prior to passing customer information to the vendors including electricity usage, EBCE rate and the solar potential of the building as determined by Google's SunRoof solar estimation tool. Customer data security will be carefully managed through this process.

Time is of the essence. EBCE is striving to launch the program as soon before the 2020 fire season. Acquiring customers and getting new solar and storage systems installed can take up to 6 months, customer enrollment and system installation will continue through 2022.

Sunrun and EBCE had successfully agreed to contract terms in order to request Board approval. However, the June 11 decision by the CPUC granting PG&E the sole right to procure Local Resource Adequacy reduces the value of this procurement and puts this program in jeopardy. Although the CPUC publicly states support for resilience the central RA buyer decision reduces the incentive for LSEs like EBCE to support behind the meter solar and storage programs that can improve community resilience. EBCE is still evaluating ways that this program can still be successful, and therefore requests the Board authorize the CEO to negotiate and execute these contracts.

Attachments

- A. Resolution to Authorize the CEO to Negotiate and Execute Agreements for Up to 10.75 Megawatts of Load Modification
- B. Distributed Resource Adequacy Capacity RFP
- C. Sunrun contract

Appendix 1: Detailed Evaluation Criteria for the Distributed RA Capacity RFP

A. Completeness of Response to RFP		Weighting
	Responses to this RFP must be complete. ALL required Proposal sections (Cover Letter, Developer Qualifications, Capacity & Program Approach and Pricing Form) are included. Additionally, within the Proposed Capacity & Program Approach section, all Mandatory Proposal Parameters must be addressed. Responses that do not include the proposal content requirements identified within this RFP and subsequent Addenda will be considered incomplete, and be rated a Fail in the Evaluation Criteria.	Pass/Fail
B. Capacity Requirements & Deployment Timeline		
	All proposed RA capacity is sited on eligible customer sites, uses eligible technologies and meets the 20% minimum requirements for DAC, low-income, MUD, CARE/FERA and medical baseline projects in the residential and commercial customer categories. Additionally, Proposer indicates ability to meet desired deployment timeline of initial proposed capacity deployment by September 30th, 2020 and the remaining proposed capacity by September 30th, 2021.	Pass/Fail
C. Technology Safety Certifications & Standards		
	Proposer indicates that proposed PV and battery types, or any other proposed technologies, meet all relevant safety certification and standards including, but not limited to, the standards described in the RFP (UL 1642, UL 1741, UL 1973, UL 9540, IEC 62109-1, IEC 62619, AC 156)	Pass/Fail
D. Islanding Capability		
	All storage systems installed by the Proposer must have the ability to island from the electrical grid and provide backup power for the customer site.	Pass/Fail
E. Pricing		
	Scoring of proposed price based on bins shown in the scoring matrix tab	40
F. Experience Developing Similar Projects & Providing RA Capacity		
	The Proposer has significant experience developing and constructing solar + storage systems in California, particularly the Bay Area. The Proposer demonstrates expertise in developing dispatchable solar + storage systems and provides solutions capable of providing the dual value streams of customer savings and market participation.	5

	Proposer has experience successfully participating in PDR (or other alternative proposed mechanism as specified in a given Proposal) and demonstrates a deep understanding of the PDR program and the requirements for CAISO participation.	7
	The Proposer has demonstrated success developing DER projects in disadvantaged and low-income communities in California and/or has experience utilizing public - private partnerships to market DER technologies.	3
G. Experience & Qualifications of Personnel		
	The listed personnel has experience designing, developing and managing projects/programs that include CAISO market participation, particularly through PDR, and/or DER development at a regional scale. Added value may be assigned to personnel involved in creation and design of relevant CAISO market products.	10
H. Quality & Detail of Go-to-Market & Customer Engagement Plan		
	The proposed go-to-market strategy approach is innovative, comprehensive and collaborative. It indicates both a willingness and a clear approach to creating a tailored program for development of DER for RA capacity beyond current or past marketing efforts of the Proposer.	5
	The Customer Engagement Strategy includes clear descriptions of how the Proposer will effectively use EBCE marketing assets to reach priority communities.	5
	Proposal includes a discussion of how the installed systems' operations will be optimized between serving customer needs (resilience and bill savings) and providing the promised RA capacity. This discussion should describe the Proposers battery management software and sizing approach	3
	Proposal includes a line-item description of how the relevant financing mechanisms, incentives and RA payment stack up to provide a unique value proposition to EBCE's customers. This description should also include an indication of how, and how much of, EBCE's RA payment will be passed on to customers.	2
I. Proposed Fire Safety Measures		
	Proposer indicates a familiarity with and experience meeting the new NFPA 855 standards in past projects and a clear strategy for coordinating with AHJs for permitting PV + BESS projects.	5
	All impacts of meeting stated safety criteria, including additional time, expense, or limits on feasible project sites, are reasonable and made clear by Proposer.	5
J. Local Hiring/Workforce Development Plan		

Proposer provides indication that they are willing and able to meet workforce and hiring requirements and describe experience hiring Alameda County residents in past projects. Proposer provides concrete examples and descriptions of a strategy to further enhance local hiring beyond the "at least 50%" goal stated in the RFP. To achieve the highest score, Proposers should indicate a willingness to, experience with, and strategy for working with state- certified apprenticeship training programs.	10
TOTAL SCORE	100

RESOLUTION NO. R-2020-__

A RESOLUTION OF THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY APPROVING LOAD MODIFYING AGREEMENTS WITH ENEL X AND SUNRUN INC. FOR UP TO 10MW/40MWH LOAD REDUCTION TO DELIVER RESILIENCE TO EBCE CUSTOMERS

WHEREAS, The East Bay Community Energy Authority ("EBCE") was formed as a community choice aggregation agency ("CCA") on December 1, 2016, Under the Joint Exercise of Power Act, California Government Code sections 6500 *et seq.*, among the County of Alameda, and the Cities of Albany, Berkeley, Dublin, Emeryville, Fremont, Hayward, Livermore, Piedmont, Oakland, San Leandro, and Union City to study, promote, develop, conduct, operate, and manage energy-related climate change programs in all of the member jurisdictions. The cities of Newark and Pleasanton, located in Alameda County, along with the City of Tracy, located in San Joaquin County, were added as members of EBCE and parties to the JPA in March of 2020.

WHEREAS, On July 18, 2018, the Board approved the Local Development Business Plan ("LDBP") and budget. The LDBP identifies a series of local development early actions, including collaborative procurement, and outlines and defines a series of policy principles and metrics to measure the impact on Alameda County.

WHEREAS, Public Safety Power Shutoffs ("PSPS") have already impacted over 50,000 EBCE customers and are expected to continue for the next 10 years; and

WHEREAS, PSPS events are particularly difficult for medical baseline, CARE and frontline communities; and

WHEREAS, Solar and battery storage systems can provide resilience to sustain residential customers in both single and multi-family dwellings as well as commercial customers through multi-day PSPS events; and

WHEREAS, EBCE jointly issued a public solicitation with Peninsula Clean Energy, Silicon Valley Clean Energy, and Silicon Valley Power on December 23 to request proposals for a combined total of 32.7MW of Resource Adequacy from vendors; and

WHEREAS, EBCE has completed a review of proposals and selected two vendors, Sunrun Inc. and Enel X, to partner with to deliver solar and storage solutions to EBCE customers for the purpose of increasing resilience in our community and delivery peak load reduction to EBCE; and

WHEREAS, EBCE has worked with the California Energy Commission to develop an alternative approach to delivering Resource Adequacy (RA) to LSEs by aggregating and dispatching a portfolio of behind-the-meter (BTM) solar and storage assets to reduce EBCE's peak demand and RA obligation; and

WHEREAS, the Board desires to engage with Sunrun Inc. and Enel X to deliver up to 10.75MW/43MWh of solar + storage on the homes and businesses of EBCE customers.

NOW, THEREFORE, THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY DOES HEREBY FIND, RESOLVE AND ORDER AS FOLLOWS:

<u>Section 1.</u> The CEO or his designee is hereby designated to negotiate and execute a Load Modifying Agreement with Sunrun Inc. attached in Exhibit A, for Load Modification up to 5.75MW/23MWh through December 31, 2032.

Section 2. The CEO or his designee is hereby designated to negotiate and execute a Load Modifying Agreement with Enel X in substantially the same form as the Sunrun agreement attached in Exhibit A, for Load Modification up to 5MW/20MWh through December 31, 2032.

Section 3. The CEO is further authorized to execute any clarifying or clerical changes to the Load Modifying Agreements with Sunrun Inc. and Enel X executed in accordance with this resolution, provided that such changes are reviewed and approved by General Counsel.

ADOPTED AND APPROVED this 17th day of June, 2020.

Dan Kalb, Chair

ATTEST:

Stephanie Cabrera, Clerk of the Board

CONFIDENTIAL

EBCE BEHIND THE METER LOAD MODIFYING AGREEMENT JUNE 2020

BEHIND THE METER

ENERGY STORAGE AGREEMENT FOR RESILIENCE

COVER SHEET

Seller: Sunrun Inc., a Delaware corporation

Buyer: East Bay Community Energy Authority, a California joint powers authority

Execution Date:

Description of Project:

Contract Amount (Load Modification):

Minimum of _1,350__kW ("Minimum Commitment") for _4_hours per day

Not to Exceed _5,750_kW ("Maximum Commitment") for _4_hours per day

Contract Price (Load Modification):

Contract Amounts (Capacity Attributes):1

<u>RA Attributes</u>: _Minimum of 400 kW ("Minimum Commitment"); Not to Exceed 1,650 kW ("Maximum Commitment") NQC

Local RA Attributes: Minimum of 400 kW ("Minimum Commitment"); Not to Exceed 1,650 kW ("Maximum Commitment") NQC

Contract Price (Capacity Attributes):

Milestones:²

Milestone	Date
Customer Agreements Signed	10% of Contract Amount by: January 15, 2021 25% of Contract Amount by: September 15, 2021 50% of Contract Amount by: July 15, 2022 Remaining Contract Amount by: October 15, 2022

¹ NTD: Capacity Attributes will apply in event of Product Transition from LM to RA pursuant to Section 5.2(b).

 $^{^{2}}$ NTD: Host customer-related matters to be addressed in Co-Marketing Agreement; milestones in this agreement focus on other key project development tasks.

Load Modifying Supply Plan Submitted	March 1, 2021
Construction Start Deadlines	10% of Contract Amount by March 31, 2021 25% of Contract Amount by: November 15, 2021 50% of Contract Amount by: August 15, 2022 Remaining Contract Amount by: November 15, 2022
Expected Initial Delivery Date	December 31, 2022

Delivery Term: _10___ Contract Years

TABLE OF CONTENTS

ARTICLE 1:	DEFINITIONS	1
1.1	Contract Definitions	1
1.2	Rules of Interpretation	20
ARTICLE 2:	TERM	21
2.1	Effectiveness	21
2.2	Term	22
ARTICLE 3:	PROJECT DEVELOPMENT	22
3.1	Project Construction	22
3.2	Interconnection	23
3.3	Metering	23
3.4	Progress Reports	23
3.5	Remedial Action Plans	23
ARTICLE 4:	INITIAL DELIVERY DATES	24
4.1	Timing of the Initial Delivery Dates	24
4.2	Conditions Precedent to the Initial Delivery Dates	24
4.3	Cooperation in Connection with Initial Delivery Dates	27
4.4	Confirmation of Initial Delivery Dates	27
ARTICLE 5:	TRANSACTION	27
5.1	Product	27
5.2	Purchase and Sale Obligation	28
5.3	Allocation of CAISO Payments and Costs	29
5.4	Customers	29
ARTICLE 6:	OPERATIONS	30
6.1	Operations	30
6.2	Charging Energy	31
6.3	Standard of Care	31
6.4	Buyer's Compliance Obligations; Certification of Product	31
6.5	Scheduling	31
6.6	Information Sharing and Shared Learning	32
6.7	Changes in Law	32
ARTICLE 7:	EVENTS OF DEFAULT; REMEDIES	34
7.1	Events of Default	34
7.2	Remedies; Declaration of Early Termination Date	35
7.3	Termination Payment	36
7.4	Notice of Payment of Termination Payment	36
7.5	Disputes with Respect to Termination Payment	37
7.6	Rights and Remedies Are Cumulative	37
ARTICLE 8:	FORCE MAJEURE	37
8.1	Force Majeure	37

1

	9: INVOICING AND PAYMENT	38
9.1	Invoicing	38
9.2	Payment	38
9.3 9.4	Payment Adjustments; Billing Errors	38 39
9.4 9.5	Billing Disputes	39 39
9.5	Netting of Payments	39
	10: PERFORMANCE ASSURANCE	39
10.1	1 2	39
10.2	5	40
10.3		40
10.4	Seller Financial Information	41
ARTICLE	11: SAFETY	41
11.1	5	41
11.2	1 0	42
11.3	Remediation	42
ARTICLE	12: TAXES	42
12.1	Taxes	42
12.2	2 Cooperation	43
ARTICLE	13: LIMITATIONS	43
13.1	No Consequential Damages	43
13.2	2 Waiver and Exclusion of Other Damages	43
ARTICLE	14: REPRESENTATIONS; WARRANTIES; COVENANTS	44
14.1	Seller's Representations and Warranties	44
14.2	Buyer's Representations and Warranties	45
14.3	General Covenants	46
14.4	Covenants of Seller	46
ARTICLE	15: INDEMNITIES	47
15.1	Indemnity by Seller	47
15.2		47
15.3	Defense of Claims	48
15.4	Subrogation of Rights	48
15.5	Rights and Remedies are Cumulative	49
ARTICLE	16: INSURANCE	49
16.1	Insurance	49
16.2	Evidence of Insurance	50
16.3	Failure to Comply	50
ARTICLE	17: RECORDS AND AUDIT RIGHTS	50
17.1	Operations Logs	50
17.2	Records and Audit	50
17.3	General Audit Right	51

2

17.4	State Auditor	51
17.5	Data Request Cooperation	51
17.6	Access Rights	51
ARTICLE 18: ASSIGNMENT		51
18.1	General Prohibition on Assignment	51
18.2	Collateral Assignment	51
18.3	Permitted Assignment by Seller	52
18.4	Unauthorized Assignment; Costs	52
ARTICLE 19: DISPUTE RESOLUTION		52
19.1	Governing Law	52
19.2	Dispute Resolution	53
ARTICLE 20: CONFIDENTIAL INFORMATION		53
20.1	Confidential Information	53
20.2	Permitted Disclosures	53
20.3	Remedies	54
20.4	Customer Information	54
ARTICLE 21	: GENERAL PROVISIONS	54
21.1	Entire Agreement; Integration; Exhibits	54
21.2	Amendments	54
21.3	Survival	54
21.4	Waivers	55
21.5	No Agency, Partnership, Joint Venture or Lease	55
21.6	Severability	55
21.7	Mobile Sierra	55
21.8	1	55
21.9	Electronic Delivery	55
	No Recourse to Members of Buyer	56
	Forward Contract	56
21.12	Further Assurances	56
ARTICLE 22: NOTICES		56
22.1	Addresses for the Delivery of Notices	56
22.2	Time of Delivery	56

APPENDICES

Appendix I – Description of Portfolio	I-1
Appendix II – Operational Characteristics	II-1
Appendix III – Progress Reporting Form	III-1
Appendix IV – Certification for Commercial Operation	IV-1
Appendix V – Form of Letter of Credit	V-1
Appendix VI – Portfolio Safety Plan and Documentation	VI-1
Appendix VII – MUA Decision Attestation	VII-1
Appendix VIII – Notices	VIII-1
Appendix IX – Portfolio Modification Certification	IX-1
Appendix X – Form of Quarterly Report	X-1
Appendix XI – [RESERVED]	XI-1
Appendix XII – Terms and Conditions for Product Transition	XII-1
Appendix XIII – Form of Annual Load Modifying Resource Dispatch Plan	XIII-1

ENERGY STORAGE AGREEMENT FOR RESILIENCE

This Energy Storage Agreement for Resilience ("<u>Agreement</u>") is made by and between the buyer ("<u>Buyer</u>") and the seller ("<u>Seller</u>") as of the execution date ("<u>Execution Date</u>"), in each case as set forth on the cover sheet ("<u>Cover Sheet</u>") to this Agreement. Seller and Buyer are referred to each individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>."

RECITALS

WHEREAS, Seller intends to develop, design, permit, construct, own, and operate the Projects in the Portfolio; and

WHEREAS, Seller desires to sell, and Buyer desires to purchase, on the terms and conditions set forth in this Agreement, the Product;

NOW THEREFORE, in consideration of the mutual covenants and agreements herein contained, and for other good and valuable consideration, the sufficiency and adequacy of which are hereby acknowledged, the Parties agree to the following:

ARTICLE 1: DEFINITIONS

1.1 <u>Contract Definitions</u>. Capitalized terms used in this Agreement have the following meanings, unless otherwise specified:

"<u>Affiliate</u>" of a Person means any other Person that (a) directly or indirectly controls the specified Person; (b) is controlled by or is under direct or indirect common control with the specified Person; or (c) is an officer, director, employee, representative or agent of the Person. For the purposes of this definition, "control", when used with respect to any specified Person, means the power to direct the management or policies of the specified Person, directly or indirectly, through one or more intermediaries.

"<u>Agreement</u>" has the meaning set forth in the preamble.

"Ancillary Services" has the meaning set forth in the CAISO Tariff.

"Availability Incentive Payments" has the meaning set forth in the CAISO Tariff.

"Availability Standards" has the meaning set forth in the CAISO Tariff.

"Balancing Authority" has the meaning set forth in the CAISO Tariff.

"<u>Bankrupt</u>" means with respect to any entity, such entity that (a) files a petition or otherwise commences, authorizes or acquiesces in the commencement of a proceeding or cause of action under any bankruptcy, insolvency, reorganization or similar Law, (b) has any such petition filed or commenced against it which remains unstayed or undismissed for a period of ninety (90) days, (c) makes an assignment or any general arrangement for the benefit of creditors, (d) otherwise becomes bankrupt or insolvent (however evidenced), (e) has a liquidator, administrator,

receiver, trustee, conservator or similar official appointed with respect to it or any substantial portion of its property or assets, or (f) is generally unable to pay its debts as they fall due.

"Bid" shall have the meaning in the CAISO Tariff.

"<u>Business Day</u>" means any day except a Saturday, Sunday, or a Federal Reserve Bank holiday in California.

"Buyer" has the meaning set forth in the preamble to this Agreement.

"<u>Buyer Group</u>" has the meaning set forth in <u>Section 15.1</u>.

"<u>CAISO</u>" means the California Independent System Operator Corporation or any successor entity performing similar functions.

"<u>CAISO Grid</u>" means the system of transmission lines and associated facilities of the Participating Transmission Owners that have been placed under the CAISO's operational control.

"CAISO Markets" has the meaning set forth in the CAISO Tariff.

"<u>CAISO Tariff</u>" means the CAISO Fifth Replacement FERC Electric Tariff and protocol provisions, including any CAISO-published procedures or business practice manuals, as they may be amended, supplemented or replaced (in whole or in part) from time to time.

"Capacity Attributes" means, any and all of the following attributes:

- (a) RA Attributes,
- (b) Local RA Attributes,
- (c) Flexible RA Attributes, and
- (d) Other Capacity Attributes.

"<u>CARB</u>" means the California Air Resources Board or any successor entity performing similar functions.

"<u>CEC</u>" means the California Energy Commission or any successor entity performing similar functions.

"<u>Change of Control</u>" means, except in connection with public market transactions of equity interests of Seller's Ultimate Parent, any circumstance in which Seller's Ultimate Parent as of the Execution Date ceases to be the Ultimate Parent or ceases to own, directly or indirectly through one or more intermediate entities, more than fifty percent (50%) of the outstanding equity interests in Seller; provided that in calculating ownership percentages for all purposes of the foregoing:

(a) any ownership interest in Seller held by Ultimate Parent indirectly through one or more intermediate entities shall not be counted towards Ultimate Parent's ownership interest in Seller unless Ultimate Parent directly or indirectly owns more than fifty percent (50%) of the outstanding equity interests in each such intermediate entity; and

(b) ownership interests in Seller owned directly or indirectly by any Lender (including any tax equity provider) shall be excluded from the total outstanding equity interests in Seller.

"<u>Charging Energy</u>" means the amount of Energy withdrawn from the Utility Distribution Company's electrical system, Participating Transmission Owner's electrical system, the CAISO Grid, or otherwise, to be stored by the Project.

"<u>Claim</u>" has the meaning set forth in <u>Section 15.1(a)</u>.

"<u>Compliance Obligations</u>" means the RAR, Local RAR, Flexible RAR, and any other resource adequacy or capacity procurement requirements, or associated reliability requirements, imposed on Load Serving Entities (as defined in the CAISO Tariff) by the CPUC pursuant to the CPUC Decisions, by the CAISO, by the WECC, by NERC, or by any other Governmental Authority having jurisdiction.

"<u>Compliance Showings</u>" means the total combination of (a) through (d) below that a Load Serving Entity (as defined in the CAISO Tariff) is required to make to the CPUC pursuant to the CPUC Decisions, or to any Governmental Authority having jurisdiction: (a) Local RAR compliance or advisory showings (or similar or successor showings), (b) RAR compliance or advisory showings (or similar or successor showings), (c) Flexible RAR compliance or advisory showings (or similar or successor showings), and (d) other Capacity Attributes compliance or advisory showings (or similar or successor showings).

"<u>Conditions Precedent</u>" has the meaning set forth in <u>Section 4.2</u>.

"Confidential Information" has the meaning set forth in Section 20.1.

"<u>Confirmation Notice</u>" means a Downward Change Confirmation Notice or an Upward Change Confirmation Notice, as applicable.

"<u>Construction Delay Cure Period</u>" has the meaning set forth in <u>Section 3.1(d)</u>.

"<u>Construction Delay Damages</u>" means liquidated damages in an amount equal to (a) the total Development Security amount required hereunder, divided by (b) one hundred twenty (120).

"<u>Construction Start</u>" means (i) acquisition of all Governmental Approvals necessary for the construction of the applicable Project(s), (ii) engagement of each Contractor, (iii) execution of each required Customer Agreement for such Project(s), and (iv) ordering of all essential equipment and supplies, in each case (i)-(iv), as reasonably necessary so that physical construction of the applicable Projects of the Portfolio may begin and proceed to completion without foreseeable interruption of material duration.

"Construction Start Deadline" has the meaning set forth in the Cover Sheet.
"<u>Contract Price</u>" means the amount specified in <u>Section 5.2(d)</u>.

"<u>Contract Year</u>" means a period of twelve (12) consecutive months; the first Contract Year shall commence on the Initial Delivery Date and each subsequent Contract Year shall commence on the anniversary of the Initial Delivery Date. The final Contract Year may be a period of less than twelve (12) consecutive months.

"<u>Contractor</u>" means the EPC Contractor and its subcontractors, as well as Seller or Seller's Affiliates if any such entities are developing, constructing, operating or maintaining the Project during the Term, and any entity or person under contract with Seller or Seller's Affiliates for the purpose of developing, constructing, operating or maintaining the Project during the Term.

"<u>Costs</u>" means, with respect to the Non-Defaulting Party, brokerage fees, commissions and other similar third party transaction costs and expenses reasonably incurred by such Party either in terminating any arrangement pursuant to which it has hedged its obligations or entering into new arrangements that replace this Agreement; and all reasonable attorneys' fees and expenses incurred by the Non-Defaulting Party in connection with the termination of this Agreement.

"<u>Cover Sheet</u>" has the meaning set forth in the preamble to this Agreement.

"<u>CPM</u>" means "Capacity Procurement Mechanism" and has the meaning set forth in the CAISO Tariff and is inclusive of any successor mechanisms authorized by CAISO.

"CPM Capacity" has the meaning set forth in the CAISO Tariff.

"<u>CPUC</u>" or "<u>Commission</u>" means the California Public Utilities Commission or any successor entity performing similar functions.

"<u>CPUC Decisions</u>" means CPUC Decisions 04-01-050, 04-10-035, 05-10-042, 06-04-040, 06-06-064, 06-07-031, 07-06-029, 08-06-031, 09-06-028, 10-06-036, 11-06-022, 12-06-025, 13-06-024, 14-06-050, and any other existing or subsequent decisions, resolutions or rulings related to resource adequacy, as may be amended from time to time by the CPUC.

"<u>Credit Rating</u>" means, with respect to any entity, the rating then assigned to such entity's unsecured, senior long-term debt obligations (not supported by third party credit enhancements) or if such entity does not have a rating for its senior unsecured long-term debt, then the rating then assigned to such entity as an issuer rating by S&P or Moody's. If ratings by S&P and Moody's are not equivalent, the lower rating shall apply.

"<u>Critical Milestone</u>" has the meaning set forth in <u>Section 3.5</u>.

"<u>Customer</u>" means a Person that is a retail electric customer of Buyer.

"<u>Customer Agreement</u>" means Seller's agreement with a Customer for the installation and operation of a Project as part of the Project.

4

"<u>Customer Information</u>" means Customer-related information that is subject to CPUC rules, regulations or orders or other applicable Laws regarding customer privacy, including California Public Utilities Code Section 8380 et seq.

"<u>DAC</u>" means disadvantaged community as defined by CPUC regulations and published guidance.

"<u>Damage Payment</u>" means the dollar amount equal to the total amount required to be posted by Seller as Development Security pursuant to <u>Section 10.1</u>.

"<u>Defaulting Party</u>" means the Party that is subject to an Event of Default.

"Delay Damages" means any Construction Delay Damages or IDD Delay Damages.

"Delay Notice" has the meaning set forth in Section 4.1(b)(i).

"Delivered Quantities" has the meaning set forth in Section 5.2(c).

"<u>Delivery Term</u>" has the meaning set forth in <u>Section 2.2(b)</u>.

"<u>Delivery Term Security</u>" means (i) cash or (ii) a Letter of Credit in an amount equal to forty dollars **been** per kilowatt of the Contract Amount that has reached the Initial Delivery Date at the time of calculation.

"<u>Development Security</u>" means (i) cash or (ii) a Letter of Credit in an amount equal to forty dollars per kilowatt of the Minimum Commitment Contract Amount.

"Disclosing Party" has the meaning set forth in Section 20.2.

"Disclosure Order" has the meaning set forth in Section 20.2(a).

"Disclosure Request" has the meaning set forth in Section 20.2(b).

"<u>Dispatch Plan Deadline</u>" means, with respect to each year of the Delivery Term, March [20], or such other date which is [30] days prior to the date on which Buyer is required to submit its annual aggregate peak demand forecast pursuant to rules and published guidance of the CEC.

"Downward Change Confirmation Notice" has the meaning set forth in Section 6.7(a)(i).

"Downward Change Notice" has the meaning set forth in Section 6.7(a)(i).

"Early Termination Date" has the meaning set forth in Section 7.2(a).

"<u>EFC</u>" or "<u>Effective Flexible Capacity</u>" has the meaning given to "Effective Flexible Capacity" in the CAISO Tariff.

"<u>Electric System Upgrades</u>" means any upgrades, including, Network Upgrades, Distribution Upgrades, or Interconnection Facilities (as these terms are defined in the CAISO Tariff), that are determined to be necessary by the CAISO, Participating TO, or Utility Distribution Company as applicable, to physically and electrically interconnect the Project to the Utility Distribution Company's/Participating TO's electric system for delivery of Energy from the Project such that the Project can provide Product at all times during the Delivery Term.

"<u>Energy</u>" means single- or three-phase, 60-cycle alternating current electric energy, measured in MWhs.

"Environmental Costs" means costs incurred in connection with acquiring and maintaining all environmental permits and licenses for the Product, and the Product's and Project's compliance with all applicable environmental Laws, rules and regulations, including capital costs for pollution mitigation or installation of emissions control equipment required to permit or license the Product or Project, all operating and maintenance costs for operation of pollution mitigation or control equipment, costs of permit maintenance fees and emission fees as applicable, and the costs of all emission reduction credits, marketable emission trading credits, and any costs related to greenhouse gas emissions, required by any applicable environmental Laws, rules, regulations, and permits to operate, and costs associated with the disposal and clean-up of hazardous substances introduced to a Site or the Project.

"<u>EPC Contract</u>" means the Seller's engineering, procurement and construction contract with the EPC Contractor.

"<u>EPC Contractor</u>" means Seller's engineering, procurement and construction contractor or such Person performing those functions.

"Event of Default" means a Seller's Event of Default or a Party's Event of Default.

"Execution Date" has the meaning set forth in the preamble.

"<u>Exigent Circumstance</u>" means actual or imminent harm to life or safety, public health, third-party owned property, including a Site, or the environment due to or arising from the Project or portion thereof.

"Expected Initial Delivery Date" has the meaning set forth on the Cover Sheet.

"<u>FERC</u>" means the Federal Energy Regulatory Commission or any successor entity performing similar functions.

"<u>Flexible RA Attributes</u>" means any and all flexible resource adequacy attributes, as may be identified at any time during the Delivery Term by the CPUC, CAISO or other Governmental Authority having jurisdiction that can be counted toward Flexible RAR, exclusive of any RA Attributes, Local RA Attributes, or Other Capacity Attributes.

"<u>Flexible RAR</u>" means the flexible resource adequacy requirements established for Load Serving Entities (as defined in the CAISO Tariff) by the CPUC pursuant to the CPUC Decisions, or by any other Governmental Authority having jurisdiction.

"<u>Force Majeure</u>" means any event or circumstance which wholly or partly prevents or delays the performance of any material obligation arising under the Agreement, but only if and to

the extent (x) such event is not within the reasonable control, directly or indirectly, of the Party seeking to have its performance obligation excused thereby, (y) the Party seeking to have its performance obligation excused thereby has taken all reasonable precautions and measures in order to prevent or avoid such event or mitigate the effect of such event on such Party's ability to perform its obligations under the Agreement and which by the exercise of due diligence such Party could not reasonably have been expected to avoid and which by the exercise of due diligence it has been unable to overcome, and (z) such event is not the direct or indirect result of the negligence or the failure of, or caused by, the Party seeking to have its performance obligations excused thereby. Additionally:

> Force Majeure may include: (a)

acts of God, including landslide, lightning, earthquake, storm, (i) hurricane, flood, drought, tornado, or other natural disasters and weather-related events affecting an entire region which caused failure of performance;

transportation accidents affecting delivery of equipment only if such (ii) accident occurs prior to the applicable Commercial Operation Date;

- (iii) sabotage, riot, acts of terrorism, war and acts of public enemy; or
- restraint by court order or other Governmental Authority. (iv)
- (b) Force Majeure does not include:

Covid-19 related restrictions imposed by Governmental Authorities (i) as currently in place as of the Effective Date;

a failure of performance of any third party, including Participating (ii) TO, Utility Distribution Company, or any other party providing electric interconnection, distribution or transmission service;

(iii) breakage or malfunction of equipment (except to the extent that such failure was caused by an event that would otherwise satisfy the definition of a Force Majeure event as defined above);

(iv) a strike, work stoppage or labor dispute limited only to any one or more of Seller, Seller's Affiliates, the EPC Contractor or subcontractors thereof or any other third party employed by Seller to work on the Project;

Seller's ability to sell the Product at a price greater than the price set (v) forth in this Agreement;

Seller's inability to obtain permits or approvals of any type for the (vi) construction, operation or maintenance of the Project, unless caused solely by an event of Force Majeure of the specific type described in any of subsections (a)(i) through (a)(iv) above;

(vii) Seller's inability to complete interconnection by the applicable Expected Initial Delivery Date, unless such delay is caused solely by an event of Force Majeure of the specific type described in any of subsections (a)(i) through (a)(iv) above;

(viii) Seller's inability to obtain sufficient Charging Energy, fuel, power or materials to operate the Portfolio Project (including as a result of any Public Safety Power Shutoff), except if Seller's inability to obtain sufficient Charging Energy, fuel, power or materials is caused solely by an event of Force Majeure of the specific type described in any of subsections (a)(i) through (a)(iv) above;

(ix) Seller's failure to obtain additional funds, including funds authorized by a state or the federal government or agencies thereof, to supplement the payments made by Buyer pursuant to this Agreement;

(x) Seller's failure to obtain or retain Customers; or

(xi) any failure of a Customer to perform (whether or not due to Force Majeure affecting a Customer).

"Force Majeure Failure" has the meaning set forth in Section 8.1(d).

"<u>Gains</u>" means, with respect to any Party, an amount equal to the present value of the economic benefit to it, if any (exclusive of Costs), resulting from the termination of this Agreement. Factors used in determining the economic benefit to a Party may include, without limitation, reference to information supplied by one or more third parties, which shall exclude Affiliates of the Non-Defaulting Party, including without limitation, quotations (either firm or indicative) of relevant rates, prices, yields, yield curves, volatilities, spreads or other relevant market data in the relevant markets, comparable transactions, forward price curves based on economic analysis of the relevant markets, settlement prices for comparable transactions at liquid trading hubs, all of which should be calculated for the remaining Delivery Term to determine the value of the Product.

"<u>Generally Accepted Accounting Principles</u>" means the standards for accounting and preparation of financial statements established by the Federal Accounting Standards Advisory Board (or its successor agency) or any successor standards adopted pursuant to relevant SEC rule.

"<u>Governmental Approval</u>" means all authorizations, consents, approvals, waivers, exceptions, variances, filings, permits, orders, licenses, exemptions, notices to and declarations of or with any Governmental Authority and shall include those siting and operating permits and licenses, and any of the foregoing under any applicable environmental Law, that are required for the development, use and operation of the Project, including any approvals required under the California Environmental Quality Act.

"<u>Governmental Authority</u>" means any federal, state, provincial, local or municipal government, any political subdivision thereof or any other governmental, congressional or parliamentary, regulatory, or judicial instrumentality, authority, body, agency, department, bureau, or entity with authority to bind a Party at law, including CAISO; *provided, however*, that "Governmental Authority" shall not in any event include any Party.

"<u>Hazardous Substance</u>" means, collectively, (a) any chemical, material or substance that is listed or regulated under applicable Laws as a "hazardous" or "toxic" substance or waste, or as a "contaminant" or "pollutant" or words of similar import, (b) any petroleum or petroleum products, flammable materials, explosives, radioactive materials, asbestos, urea formaldehyde foam insulation, and transformers or other equipment that contain polychlorinated biphenyls, and (c) any other chemical or other material or substance, exposure to which is prohibited, limited or regulated by any Laws.

"IDD Cure Period" has the meaning set forth in Section 4.1(b)(i).

"<u>IDD Delay Damages</u>" means liquidated damages in an amount equal to (a) the total Development Security amount required hereunder, divided by (b) sixty (60).

"Indemnifiable Loss(es)" has the meaning set forth in <u>Section 15.1(a)</u>.

"Initial Delivery Date" has the meaning set forth in Section 2.2(b).

"Interconnection Agreement" means the agreement(s) and associated documents (or any successor agreement and associated documentation approved by FERC or the CPUC) by and among Seller or a Customer and, as applicable, the Utility Distribution Company, the Participating Transmission Owner, and the CAISO, governing the terms and conditions of the interconnection of the Projects with the Utility Distribution Company's or CAISO's grid, including any description of the plan for interconnecting the Projects to the applicable grid.

"Interest Rate" has the meaning set forth in Section 9.2.

"Joint Powers Agreement" means that certain Joint Powers Agreement of Buyer, dated as of December 1, 2016, by and among the County of Alameda, the City of Albany, the City of Berkeley, the City of Dublin, the City of Emeryville, the City of Fremont, the City of Hayward, the City of Livermore, the City of Oakland, the City of Piedmont, the City of San Leandro, and the City of Union City.

"<u>Law</u>" means any statute, law, treaty, rule, regulation, ordinance, code, Permit, enactment, injunction, order, writ, decision, authorization, judgment, decree or other legal or regulatory determination or restriction by a court or Governmental Authority of competent jurisdiction, including any of the foregoing that are enacted, amended, or issued after the Execution Date, and which become effective during the Term; or any binding interpretation of the foregoing.

"Lender" means, collectively, any Person (i) providing senior or subordinated construction, interim, back leverage or long-term debt, equity or tax equity financing or refinancing for or in connection with the development, construction, purchase, installation or operation of the Project, whether that financing or refinancing takes the form of private debt (including back-leverage debt), equity (including tax equity), public debt or any other form (including financing or refinancing provided to a member or other direct or indirect owner of Seller), including any equity or tax equity investor directly or indirectly providing financing or refinancing for the Project or purchasing equity ownership interests of Seller or its Affiliates for purposes of providing financing or refinancing for the Project, and any trustee or agent or similar representative acting on their behalf, (ii) providing interest rate or commodity protection under an

agreement hedging or otherwise mitigating the cost of any of the foregoing obligations or (iii) participating in a lease financing (including a sale leaseback or leveraged leasing structure) with respect to the Project.

"Letter of Credit" means one or more irrevocable, standby letters of credit issued by a U.S. commercial bank or a foreign bank with a U.S. branch with such bank having a Credit Rating of at least A- with an outlook designation of "stable" from S&P or A3 with an outlook designation of "stable" from Moody's, in a form substantially similar to the letter of credit set forth in <u>Appendix V</u>.

"<u>LIBOR</u>" means the London Interbank Offered Rate for the corresponding deposits of U.S. dollars.

"<u>Load Modification</u>" means the reduction of Buyer's aggregate peak demand forecast pursuant to applicable CEC regulations and published guidance.

"Load Modifying Resource" means a behind the meter generation or storage system that is capable of being dispatched to provide Load Modification.

"Load Modifying Resource Dispatch Plan" means a one-year plan, submitted for each calendar year of the Delivery Term, specifying on a month by month basis the hours of day during which the Contract Amounts will be delivered from the Project and substantially in the form of Appendix XIII.

"Local Capacity Area" has the meaning set forth in the CAISO Tariff.

"<u>Local RA Attributes</u>" means any and all resource adequacy attributes or other locational attributes related to a Local Capacity Area, as may be identified at any time during the Delivery Term by the CPUC, CAISO or other Governmental Authority having jurisdiction, associated with a physical location or point of electrical interconnection within the CAISO's Balancing Authority, that can be counted toward a Local RAR.

"Local RAR" means the local resource adequacy requirements established for Load Serving Entities (as defined in the CAISO Tariff) by the CPUC pursuant to CPUC Decisions, or by any other Governmental Authority having jurisdiction. Local RAR may also be known as local area reliability, local resource adequacy, local resource adequacy procurement requirements, or local capacity requirement in other regulatory proceedings or legislative actions.

"Losses" means, with respect to any Party, an amount equal to the present value of the economic loss to it, if any (exclusive of Costs), resulting from termination of this Agreement. Factors used in determining economic loss to a Party may include, without limitation, reference to information supplied by one or more third parties, which shall exclude Affiliates of the Non-Defaulting Party, including without limitation, quotations (either firm or indicative) of relevant rates, prices, yields, yield curves, volatilities, spreads or other relevant market data in the relevant markets, comparable transactions, forward price curves based on economic analysis of the relevant markets, settlement prices for comparable transactions at liquid trading hubs, all of which should be calculated for the remaining Delivery Term to determine the value of the Product. If the Non-Defaulting Party is the Seller, then "Losses" shall exclude any loss of federal or state tax credits,

grants, or benefits related to the Project or generation therefrom or any costs or fees related to the Site or Project.

"Maximum Commitment Contract Amount" has the meaning set forth in the Cover Sheet.

"Minimum Commitment Contract Amount" has the meaning set forth in the Cover Sheet.

"Monthly Payment" has the meaning set forth in Section 5.2(d).

"Moody's" means Moody's Investors Service, Inc., or its successor.

"MUA Decision" has the meaning set forth in Section 4.2(p).

"Must Offer Obligations" means Seller's obligation to Bid or cause Seller's SC to Bid the Project into the CAISO Markets due to delivery of the Product to Buyer and in accordance with the CAISO Tariff.

"<u>MW</u>" means megawatts in alternating current, unless expressly stated in terms of direct current.

"NERC" means the North American Electric Reliability Corporation or any successor entity performing similar functions.

"Non-Availability Charges" has the meaning set forth in the CAISO Tariff.

"Non-Defaulting Party" has the meaning set forth in Section 7.2(a).

"Non-Spinning Reserve" has the meaning set forth in the CAISO Tariff.

"Notice" unless otherwise specified in this Agreement, means written communications by a Party to be delivered by hand delivery, United States mail, overnight courier service, or electronic messaging (e-mail).

"Notice of Claim" has the meaning set forth in Section 15.2(a).

"Notify" means to provide a Notice.

"NQC" or "Net Qualifying Capacity" has the meaning given to "Net Qualifying Capacity" in the CAISO Tariff as may be updated from time to time.

"Operational Characteristics" means the operational characteristics set forth in Appendix II.

"Other Capacity Attributes" means, exclusive of RA Attributes, Local RA Attributes, and Flexible RA Attributes, any (a) current or future capacity characteristics or attributes, including the ability to generate or charge at given capacity levels, the ability to ramp up or down at a given rate, flexibility or dispatchability attributes, and locational attributes, as may be identified at any time during the Delivery Term by any applicable Law, or voluntary or mandatory program of any Governmental Authority or other Person, (b) certificate, tag, or credit, intended to commoditize or otherwise attribute value resulting from or associated with the characteristics set forth in subsection (a) of this definition, and (c) any accounting construct so that the characteristics or values set forth in subsections (a) or (b) hereof may be counted toward any Compliance Obligations.

"<u>Other Programs</u>" has the meaning set forth in <u>Section 5.4(f)</u>.

"<u>Participating Transmission Owner</u>" or "<u>Participating TO</u>" means an entity that (a) owns, operates and maintains transmission lines and associated facilities or has entitlements to use certain transmission lines and associated facilities and (b) has transferred to the CAISO operational control of such facilities or entitlements to be made part of the CAISO Grid.

"<u>Party</u>" or "<u>Parties</u>" has the meaning set forth in the preamble of this Agreement.

"Party's Event of Default" has the meaning set forth in Section 7.1(b).

"<u>Performance Assurance</u>" means collateral provided by Seller to Buyer to secure Seller's obligations under this Agreement and includes Development Security and Delivery Term Security.

"<u>Permit</u>" means any waiver, exemption, variance, franchise, permit, authorization, consent, ruling, certification, license or similar order of or from, or filing or registration with, or notice to, any Governmental Authority that authorizes, approves, limits or imposes conditions upon a specified activity.

"<u>Person</u>" means any individual, sole proprietorship, corporation, limited liability company, limited or general partnership, joint venture, association, joint-stock company, trust, incorporated organization, institution, public benefit corporation, unincorporated organization, government entity or other entity.

"Portfolio Modification" has the meaning set forth in Section 5.4(c).

"Pricing Node" or "PNode" has the meaning set forth in the CAISO Tariff.

"<u>Product</u>" has the meaning set forth in <u>Section 5.1(a)</u>.

"<u>Product Transition Trigger</u>" shall mean either of the following (1) the CEC no longer accepts Buyer's submission of Load Modifying Resources as a means of reducing Buyer's annual peak demand forecast, or (2) the proportion between the aggregate amount of Load Modifying Resources submitted by Buyer, to the aggregate CEC reduction in Buyer's annual peak demand forecast, falls below eighty percent (80%).

"<u>Progress Report</u>" means a reasonably detailed progress report including the items set forth in <u>Appendix III</u>.

"<u>Project</u>" or "<u>Projects</u>" means the energy storage facilities installed at the Sites as more particularly described in <u>Appendices I</u>, <u>II</u>, and <u>IX</u>, including all appurtenant facilities,

communications and control systems, and equipment, from which Seller has agreed to provide the Product to Buyer pursuant to this Agreement.

"<u>Portfolio Safety Plan</u>" means Seller's written plan that includes the Safeguards and plans to comply with the Safety Requirements, as such Safeguards and Safety Requirements are generally outlined in <u>Appendix VI</u>.

"Proxy Demand Resource" or "PDR" has the meaning set forth in the CAISO Tariff.

"<u>Prudent Operating Practice</u>" means those practices, methods, codes and acts engaged in or approved by a significant portion of the electric power industry and applicable to energy storage facilities during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time a decision is made, that could have been expected to accomplish a desired result at reasonable cost consistent with good business practices, reliability, safety and expedition. Prudent Operating Practices are not intended to be limited to the optimum practices, methods, or acts to the exclusion of others, but rather to those practices, methods and acts generally accepted or approved by a significant portion of the electric power industry in the relevant region, during the relevant time period, as described in the immediately preceding sentence. Prudent Operating Practices also includes taking reasonable steps to ensure that:

(a) Safeguards are implemented and maintained for the Portfolio and at each Site and are sufficient to address reasonably foreseeable incidents;

(b) equipment, material, and supplies are sufficient and accessible to operate the Portfolio safety and reliably;

(c) operating personnel are trained, equipped, and capable of responsible operation and maintenance of the Portfolio and at each Site, including identifying and responding to System Emergencies, emergencies, or Exigent Circumstances originating from or impacting the Portfolio or Site;

(d) the Portfolio's material components and control systems are designed, manufactured, and configured to meet the standard of durability and safety generally used for electric power or energy storage facilities operating in the relevant region; and

(e) the Portfolio is appropriately designed, operated, maintained, monitored, and tested to ensure it continues to function safely, reliably, and consistent with the intended design specifications, applicable Laws, and Permits, and over the complete range of environmental conditions reasonably expected to occur at each Site.

"<u>**Ouarterly Report**</u>" means the report provided by Seller to Buyer within 30 days after each March 31, June 30, September 30, and December 31 following the Initial Delivery Date in the form attached as <u>Appendix X</u>.

"<u>RA Attributes</u>" means, any and all resource adequacy attributes, exclusive of any Local RA Attributes, Flexible RA Attributes and Other Capacity Attributes, as may be identified at any

time during the Delivery Term by the CPUC, CAISO or other Governmental Authority having jurisdiction, that can be counted toward RAR and Local RAR.

"<u>Receiving Party</u>" has the meaning set forth in <u>Section 20.2(b)</u>.

"<u>Regulation Down</u>" has the meaning set forth in the CAISO Tariff.

"<u>Regulation Up</u>" has the meaning set forth in the CAISO Tariff.

"<u>Regulatory Disclosure</u>" has the meaning set forth in <u>Section 20.2(a)</u>.

"<u>Reliability Demand Response Resource</u>" or "<u>RDRR</u>" has the meaning set forth in the CAISO Tariff.

"<u>Reliability Organization</u>" means an "Electric Reliability Organization" as defined in Section 215(a)(2) of the Federal Power Act or a "regional entity" as defined in Section 215(a)(7) of the Federal Power Act.

"Remedial Action Plan" has the meaning set forth in Section 3.5.

"<u>Remediation Event</u>" means the occurrence of any of the following with respect to the Portfolio or a Site: (a) an Exigent Circumstance (b) a Serious Incident; (c) a change in the nature, scope, or requirements of applicable Laws, Permits, codes, standards, or regulations issued by Governmental Authorities which requires modifications to the Safeguards; (d) a material change to the manufacturer's guidelines that requires modification to equipment or the Portfolio's operating procedures; (e) a failure or compromise of an existing Safeguard; (f) Notice by Buyer pursuant to <u>Section 11.2</u>, in its reasonable discretion, that the Portfolio Safety Plan is not consistent with the Safety Requirements; or (g) any actual condition related to the Portfolio or a Site with the potential to adversely impact the safe construction, operation, or maintenance of the Portfolio or a Site.

"<u>Remediation Period</u>" means the time period between the first occurrence of the Remediation Event and the resolution of such Remediation Event which period may not exceed a total of ninety (90) days; provided, however, that Seller may request to extend the Remediation Period by up to ninety (90) days if Seller is unable to resolve the Remediation Event within the initial ninety (90) day period despite exercising diligent efforts (and Buyer shall not unreasonably withhold approval of such extension).

"<u>Requirements</u>" means Prudent Operating Practices and all applicable requirements of Law, the Utility Distribution Company, the Transmission Provider, Governmental Approvals, the CAISO, CEC, CPUC, CARB, FERC, NERC and WECC.

"<u>Resold Product</u>" has the meaning set forth in <u>Section 5.1(b)</u>.

"<u>Resource Adequacy</u>" means the procurement obligation of load serving entities, including Buyer, as such obligations are described in CPUC Decisions D.04-10-035 and D.05-10-042 and subsequent CPUC decisions addressing Resource Adequacy issues, as those obligations

may be altered from time to time, and all other capacity procurement obligations established by any other entity, including the CAISO.

"Resource Adequacy Plan" has the meaning set forth in the CAISO Tariff.

"<u>Resource Adequacy Requirement</u>" or "<u>RAR</u>" means the Resource Adequacy or successor program requirements established by the CPUC, CAISO or any other regional entity, including submission of a Supply Plan or Resource Adequacy Plan.

"<u>RMR</u>" means "<u>Reliability Must-Run</u>" and has the meaning set forth in, and as used in, the CAISO Tariff.

"<u>RMR Contract</u>" has the meaning set forth in the CAISO Tariff.

"<u>RMR Generation</u>" has the meaning set forth in the CAISO Tariff.

"<u>S&P</u>" means Standard and Poor's Financial Services, LLC (a subsidiary of The McGraw-Hill Companies, Inc.) or its successor.

"<u>Safeguard</u>" means any procedures, practices, or actions with respect to the Portfolio, a Site or Work for the purpose of preventing, mitigating, or containing foreseeable accidents, injuries, damage, release of hazardous material or environmental harm.

"<u>Safety Remediation Plan</u>" means a written Notice from Seller to Buyer containing information about a Remediation Event, including (a) the date, time and location of first occurrence, (b) the circumstances surrounding cause, (c) impacts, and (d) detailed information about Seller's plans to resolve the Remediation Event.

"<u>Safety Requirements</u>" means Prudent Operating Practices and all applicable safetyrelated (construed broadly) requirements of Law, the Utility Distribution Company, the Transmission Provider, Governmental Approvals, the CAISO, CARB, FERC, NERC and WECC.

"<u>Scheduling Coordinator</u>" or "<u>SC</u>" has the meaning set forth the CAISO Tariff. Under the terms of this Agreement, the SC may be Seller or Seller's designated agent (i.e., a third-party).

"<u>SEC</u>" means the U.S. Securities and Exchange Commission, or any successor entity performing similar functions.

"<u>Security Interest</u>" has the meaning set forth in <u>Section 10.3(a)</u>.

"Seller" has the meaning set forth in the preamble to this Agreement.

"Seller's Event of Default" has the meaning set forth in Section 7.1(a).

"Seller's Initial Portfolio List" has the meaning set forth in Section 4.2.

"<u>Seller's Portfolio</u>" or "<u>Portfolio</u>" means the Projects, Customers and corresponding Sites assembled by Seller for purposes of delivering the Product to Buyer under this Agreement, as such Seller's Portfolio may be amended in accordance with this Agreement from time to time.

"<u>Seller's Portfolio List</u>" has the meaning set forth in <u>Section 5.4(d)</u>.

"Serious Incident" means a harmful event that occurs on a Site during the Term arising out of, related to, or connected with the Project or the Site that results in any of the following outcomes: (a) any injury to or death of a member of the general public; (b) the death or permanent, disabling injury to operating personnel, Seller's Contractors or subcontractors, Seller's employees, agents, or consultants, or authorized visitors to the Site; (c) any property damage greater than one hundred thousand dollars (\$100,000.00); (d) release of hazardous material above the limits, or violating the requirements, established by Permits, codes, standards, regulations, Laws or Governmental Authorities; (e) environmental impacts exceeding those authorized by Permits or applicable Law.

"<u>Settlement Amount</u>" means an amount equal to the greater of (a) the amount of Delivery Term Security required under this Agreement and (b) the Non-Defaulting Party's Costs and Losses, on the one hand, netted against its Gains, on the other. The Settlement Amount does not include consequential, incidental, punitive, exemplary or indirect or business interruption damages.

"Shortfall" means any difference between a Delivered Quantity and a Contract Amount.

"<u>Shortfall LDs</u>" means a dollar amount equal to twelve (12) times the product of the Contract Amount and the Contract Price, paid by Seller to Buyer within ten (10) Business Days of a failure by Seller to deliver Delivered Quantities equal to the Contract Amounts in any month during the Delivery Term, or thereafter drawn by Buyer from the Delivery Term Security.

"<u>Showing Month</u>" incorporates each day of each calendar month of the Delivery Term that is the subject of the Compliance Showings, as set forth in the CPUC Decisions. For illustrative purposes only, pursuant to the CPUC Decisions in effect as of the Execution Date, the monthly Compliance Showings made in June are for the Showing Month of August and the annual Compliance Showing made in October is for the twelve (12) Showing Months of the following year.

"Site(s)" means the real property on which the Project is located, as identified in <u>Appendix</u> I and <u>Appendix VIII</u>, as updated by Seller pursuant to <u>Section 5.4(d)</u>.

"Spinning Reserve" has the meaning set forth in the CAISO Tariff.

"Supply Plan" has the meaning set forth in the CAISO Tariff.

"<u>Supplying Party</u>" has the meaning set forth in <u>Section 20.2(b)</u>.

"System Emergency" has the meaning set forth in the CAISO Tariff.

"<u>Tax</u>" or "<u>Taxes</u>" means all U.S. federal, state and local and any foreign taxes, levies, assessments, surcharges, duties and other fees and charges of any nature imposed by a Governmental Authority, whether currently in effect or adopted during the Term, including ad valorem, excise, franchise, gross receipts, import/export, license, property, sales and use, stamp,

transfer, payroll, unemployment, income, and any and all items of withholding, deficiency, penalty, additions, interest or assessment related thereto.

"<u>Term</u>" has the meaning set forth in <u>Section 2.2(a)</u>.

"<u>Terminated Transaction</u>" has the meaning set forth in <u>Section 7.2(a)</u>.

"Termination Payment" has the meaning set forth in Section 7.3.

"Transmission Provider" means the CAISO.

"<u>Ultimate Parent</u>" means the Person that owns, directly or indirectly through one or more intermediate entities, more than fifty percent (50%) of the outstanding equity interests in Seller and is not controlled (as defined within the definition of "Affiliate") by any other Person; provided that in calculating ownership percentages or determining "control" for all purposes of the foregoing:

(a) any ownership interest in Seller held by a Person indirectly through one or more intermediate entities shall not be counted towards such Person's ownership interest in Seller unless such Person directly or indirectly owns more than fifty percent (50%) of the outstanding equity interests in each such intermediate entity; and

(b) except (i) for purposes of determinations in accordance with paragraph (a) above and (ii) for purposes of determining ownership of the direct equity interests in Seller, any ownership interest held or control exercised by a natural person shall not be taken into account.

"Upward Change Confirmation Notice" has the meaning set forth in Section 6.7(a)(ii).

"<u>Upward Change Notice</u>" has the meaning set forth in <u>Section 6.7(a)(ii)</u>.

"Utility Distribution Company" has the meaning set forth in the CAISO Tariff.

"<u>WECC</u>" means the Western Electricity Coordinating Council or its successor entity with similar functions.

"<u>Work</u>" means (a) work or operations performed by a Party or on a Party's behalf; and (b) materials, parts or equipment furnished in connection with such work or operations; including (i) warranties or representations made at any time with respect to the fitness, quality, durability, performance or use of "a Party's work"; and (ii) the providing of or failure to provide warnings or instructions.

1.2 **<u>Rules of Interpretation</u>**. In this Agreement, except as expressly stated otherwise or unless the context otherwise requires:

(a) headings and the rendering of text in bold and italics are for convenience and reference purposes only and do not affect the meaning or interpretation of this Agreement;

(b) words importing the singular include the plural and vice versa and the masculine, feminine and neuter genders include all genders;

(c) the words "hereof", "herein", and "hereunder" and words of similar import shall refer to this Agreement as a whole and not to any particular provision of this Agreement;

(d) a reference to an article, section, paragraph, clause, Party, appendix, or exhibit is a reference to that section, paragraph, clause of, or that Party, appendix, or exhibit to, this Agreement unless otherwise specified;

(e) a reference to a document or agreement, including this Agreement means such document, agreement or this Agreement including any amendment or supplement to, or replacement, novation or modification of this Agreement, but disregarding any amendment, supplement, replacement, novation or modification made in breach of such document, agreement or this Agreement;

assigns;

(f) a reference to a Person includes that Person's successors and permitted

(g) the term "including" means "including without limitation" and any list of examples following such term shall in no way restrict or limit the generality of the work or provision in respect of which such examples are provided;

(h) references to any statute, code or statutory provision are to be construed as a reference to the same as it may have been, or may from time to time be, amended, modified or reenacted, and include references to all bylaws, instruments, orders and regulations for the time being made thereunder or deriving validity therefrom unless the context otherwise requires;

(i) in the event of a conflict, a mathematical formula or other precise description of a concept or a term shall prevail over words providing a more general description of a concept or a term;

(j) references to any amount of money shall mean a reference to the amount in United States Dollars;

(k) words, phrases or expressions not otherwise defined herein that (i) have a generally accepted meaning in Prudent Operating Practice shall have such meaning in this Agreement or (ii) do not have well known and generally accepted meaning in Prudent Operating Practice but that have well known and generally accepted technical or trade meanings, shall have such recognized meanings;

(1) the terms "year" and "calendar year" mean the period of months from January 1 through and including December 31; the term "month" means a calendar month unless otherwise indicated, and a "day" means a 24-hour period beginning at 12:00:01 a.m. and ending at 12:00:00 midnight; provided that a "day" may be 23 or 25 hours on those days on which daylight savings time begins or ends, respectively;

(m) unless otherwise specified herein, where the consent of a Party is required, such consent shall not be unreasonably withheld or unreasonably delayed;

(n) when an action is required to be completed on a Business Day, such action must be completed prior to 5:00 p.m. on such day, Pacific Standard time, and actions occurring after 5:00 p.m. (such as the delivery of a Notice) will be deemed to have occurred on the following Business Day;

(o) all references to Product mean each and all components of the Product unless the context infers a particular component of Product; and

(p) each Party acknowledges that it was represented by counsel in connection with this Agreement and that it or its counsel reviewed this Agreement and that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement.

Article 2: TERM

2.1 <u>Effectiveness</u>. Upon execution by both Parties, this Agreement shall be effective and binding as of the Execution Date.³

2.2 <u>Term</u>.

(a) The "<u>Term</u>" of this Agreement shall commence upon the Execution Date and shall continue until the expiration of the Delivery Term (unless terminated earlier in accordance with the terms of this Agreement).

(b) The "<u>Delivery Term</u>" is the period commencing on the Initial Delivery Date and continuing for the number of Contract Years specified on the Cover Sheet. The "<u>Initial</u> <u>Delivery Date</u>" is the first day of the first calendar month for which Product in the Contract Amounts is delivered; provided that Initial Delivery Date may not occur until satisfaction of the Conditions Precedent set forth in <u>Article 4</u>

Article 3: PROJECT DEVELOPMENT

3.1 **Project Construction**.

(a) Seller shall develop, design and construct the Project in timely fashion in order to perform Seller's obligations under this Agreement.

(b) Seller shall cause Construction Start to occur for the applicable Projects no later than the applicable Construction Start Deadline. Seller shall provide Notice to Buyer certifying the satisfaction of this Section 3.1(c) on or before the applicable Construction Start Deadline.

³ NTD: Prior to Execution Date EBCE will require (A) Seller's most recent annual report, audited consolidated financial statements; and (B) Seller's organizational documents and any amendments thereto.

If Construction Start is not achieved on or before the applicable (c)Construction Start Deadline, then for each day beginning with the day after the applicable Construction Start Deadline through and including the date on which Construction Start occurs for the required Projects, for a period beyond the applicable Construction Start Deadline lasting no more than ninety (90) days ("Construction Delay Cure Period"), Seller shall pay Construction Delay Damages to Buyer. Buyer shall invoice Seller for the amount of any Construction Delay Damages. Seller agrees that Buyer may draw any Construction Delay Damages that are due to Buyer from the Development Security after providing an invoice for the amounts due. Prior to the expiration of the Construction Delay Cure Period, so long as Seller has paid Construction Delay Damages to Buyer in accordance with this Section 3.1(d), Seller's failure to achieve Construction Start on or before the applicable Construction Start Deadline shall not be deemed a Seller's Event of Default. If Seller achieves the Initial Delivery Date on the applicable Expected Initial Delivery Date, any Construction Delay Damages paid by Seller for the applicable Construction Start Deadline shall be refunded to Seller. Seller shall include the request for refund of the Construction Delay Damages with the first invoice to Buyer after the applicable Initial Delivery Date.

3.2 <u>Interconnection</u>. Seller shall (a) execute all necessary Interconnection Agreements for the Projects in the Project, (b) comply with all terms and conditions contained therein as necessary for the safe and reliable delivery of the Product, and (c) arrange, schedule and be responsible for any and all electric distribution and transmission service including any Governmental Approvals required for the foregoing. Seller shall fulfill all contractual, metering and applicable interconnection requirements, including Electric System Upgrades and those requirements set forth in the Utility Distribution Company's applicable tariffs, the Participating Transmission Owner's applicable tariffs, the CAISO Tariff and implementing CAISO standards and requirements, so as to be able to deliver the Product to Buyer.

3.3 <u>Metering</u>. At Seller's expense, Seller shall obtain and maintain one or more CAISO resource IDs dedicated exclusively to the Project and shall install, or cause to be installed, all necessary metering and telemetry required by the CAISO to deliver the Product. CAISO-required meters shall be owned, maintained and operated by Seller at Seller's sole cost and expense. In addition, Seller shall install, and shall maintain throughout the Delivery Term unless Buyer exercises its Product Transition rights under <u>Section 5.2</u>, a revenue-grade meter capable of measuring and demonstrating on a monthly basis the sufficiency of the Load Modification Delivered Quantities.

3.4 **<u>Progress Reports</u>**. Within fifteen (15) days after the close of every fiscal quarter, starting with the first full fiscal quarter following the Execution Date and until the Initial Delivery Date, Seller shall provide to Buyer a Progress Report in a Notice and agree to regularly scheduled meetings between representatives of Buyer and Seller to review such monthly reports and discuss Seller's construction progress. Seller shall also provide Buyer with any reasonable requested documentation, subject to the confidentiality restrictions set forth in this Agreement, directly related to the achievement of Critical Milestones within ten (10) Business Days of receipt of such request by Seller.

3.5 March 2021 and March 2022 Elections.

(a) No later than March 1, 2021,

(i) Seller shall specify by written Notice to Buyer whether it elects to increase the Minimum Commitment Contract Amount to be delivered hereunder to an amount that is no greater than the Maximum Commitment Contract Amount. Seller shall include in such written Notice a new first page to the Cover Sheet specifying the new Minimum Commitment Contract Amount and shall be responsible for ensuring that the Development Security is reflective of the new Minimum Commitment Contract Amount within ten (10) Business Days of such Notice.

(ii) Buyer shall revise Appendix XIII hereto to reflect the revised Load Modifying Resource Dispatch Plan, taking into consideration the new Minimum Commitment Contract Amount to be delivered hereunder (as may be revised pursuant to this Section 3.5).

(iii) Buyer at its option may elect to change the accounting for the Load Modification Contract Amount from a kW-Month capacity construct to a kWh energy construct so long as no changes to the Seller's compensation and revenue hereunder results from such change.

(b) No later than March 1, 2022,

(i) Seller shall specify by written Notice to Buyer whether it elects to increase the Minimum Commitment Contract Amount to be delivered hereunder to an amount that is no greater than the Maximum Commitment Contract Amount. Seller shall include in such written Notice a new first page to the Cover Sheet specifying the new Minimum Commitment Contract Amount and shall be responsible for ensuring that the Development Security is reflective of the new Minimum Commitment Contract Amount within ten (10) Business Days of such Notice.

(ii) Buyer shall revise Appendix XIII hereto to reflect the revised Load Modifying Resource Dispatch Plan, taking into consideration the new Minimum Commitment Contract Amount to be delivered hereunder (as may be revised pursuant to this Section 3.5).

3.6 <u>**Remedial Action Plans.</u>** If Seller anticipates that it will not be able to timely satisfy any Milestone set forth on the Cover Sheet, except as the result of Force Majeure or a Buyer Event of Default, Seller shall submit to Buyer no later than thirty (30) days prior to the relevant deadline a remedial action plan ("<u>**Remedial Action Plan**</u>"), which will describe in detail the actual delay, any anticipated delay beyond the scheduled deadline, the cause of the delay, and Seller's proposed course of action to achieve the missed deadline, any subsequent Milestones, and the Initial Delivery Date by the Expected Initial Delivery Date. Delivery of a Remedial Action Plan shall not relieve Seller of any obligation under this Agreement.</u>

Article 4: INITIAL DELIVERY DATES

4.1 <u>Timing of the Initial Delivery Dates</u>.

(a) <u>Initial Delivery Dates</u>. Seller shall cause the Initial Delivery Date to occur on, and not prior to, the applicable Expected Initial Delivery Date.

(b) Failure to Meet Expected Initial Delivery Date.

(i) Seller shall provide Buyer with advance Notice of any delay in achieving the Initial Delivery Date by the Expected Initial Delivery Date, including a true and reasonably detailed explanation of the cause of such delay ("<u>Delay Notice</u>"), at least ninety (90) days in advance of the Expected Initial Delivery Date (or, if Seller's anticipation of such delay does not arise until after such advance window, then as soon as reasonably possible following such anticipation arising). For each day beginning with the day after the Expected Initial Delivery Date through and including the date on which the Initial Delivery Date occurs, for a period beyond the Expected Initial Delivery Date lasting no more than one hundred eighty (180) days ("<u>IDD Cure</u> <u>Period</u>"), Seller shall pay IDD Delay Damages to Buyer. Buyer shall invoice Seller for the amount of any IDD Delay Damages. Buyer may draw any IDD Delay Damages that are due to Buyer from the Development Security after providing an invoice for the amounts due.

(ii) Prior to the expiration of the IDD Cure Period, so long as Seller has provided the Delay Notice to Buyer and paid IDD Delay Damages to Buyer in accordance with this Section 4.1(b), Seller's failure to achieve the Initial Delivery Date on the Expected Initial Delivery Date shall not be deemed a Seller's Event of Default. Upon (A) Seller's failure to provide a Delay Notice to Buyer in accordance with this Section 4.1(b), (B) Seller's failure to pay IDD Delay Damages in accordance with this Section 4.1(b), or (C) Seller's failure to achieve the Initial Delivery Date prior to the expiration of the IDD Cure Period, in each case for any reason other than a Force Majeure extension or a Buyer Event of Default, Seller will be deemed a Defaulting Party pursuant to Section 7.1(a)(v).

(iii) Upon the expiration of the IDD Cure Period, if Seller has failed to achieve the Initial Delivery Date with respect to the entire Project but is capable of achieving the Initial Delivery Date with respect to Projects of the Project which could provide no less than eighty percent (80%) of the Contract Amount, then Buyer shall have the right, in lieu of terminating this Agreement based on such event of default, to reduce the Contract Amount and Contract Amount accordingly and allow Seller to declare the Initial Delivery Date based on such reductions, in exchange for Seller's one-time payment to Buyer of per kw of such reductions.

4.2 <u>Conditions Precedent to the Initial Delivery Date</u>. The following obligations of Seller are conditions precedent to the Initial Delivery Date (collectively the "<u>Conditions</u> <u>Precedent</u>") and must be satisfied by Seller, to Buyer's reasonable satisfaction, at least seventy-five (75) days before the Initial Delivery Date, unless a different deadline is expressly set forth below, in which case such other deadline shall govern:

(a) Seller shall have secured all CAISO and Governmental Approvals as are necessary for the safe and lawful operation and maintenance of the Project and to enable Seller to deliver the Product to Buyer at the Contract Amounts.

(b) Seller shall have executed a Customer Agreement with each Customer whose Project is to be a part of the Project as necessary for the safe and lawful operation of the Project and to enable Seller to deliver the Product to Buyer in the Contract Amounts, and such Customer Agreements shall remain valid and in full force and effect.

(c) Seller shall have constructed or caused to be constructed the Projects that are to be part of the Project as of the Initial Delivery Date in accordance with this Agreement to

enable Seller to satisfy the obligations of the Seller herein, including the provision of the Product in the Contract Amounts from the Project.

(d)Seller shall have executed and complied with any necessary Interconnection Agreement(s) and installed any necessary metering to deliver Product to Buyer in the Contract Amounts, in each case in accordance with the CAISO Tariff and any applicable tariffs of the Utility Distribution Company and the Participating Transmission Owner.

> (e) Seller shall have submitted to Buyer a Portfolio Safety Plan.

Seller shall have provided Buyer with (A) Seller's Portfolio List in (f)accordance with Section 5.4(d) that demonstrates Projects and Customers under executed Customer Agreements with Seller sufficient for Seller to deliver, in aggregate, Product in the Contract Amounts ("Seller's Initial Portfolio List"), and (B) a description of the Project and Projects set forth in Appendix I. If Seller provides to Buyer Seller's Initial Portfolio List, but prior to the occurrence of the Initial Delivery Date Seller changes Seller's Initial Portfolio List, then this Condition Precedent shall not be satisfied and the date on which Seller provides to Buyer a changed Seller's Portfolio List in accordance with this Agreement shall constitute provision of Seller's Initial Portfolio List for purposes of this Condition Precedent.

(g) Seller shall have provided to Buyer an attestation, in the form attached hereto as Appendix VII, that Seller is following all of the rules set forth in CPUC Decision 18-01-003 on Multiple-Use Application Issues, issued January 17, 2018, regarding the multiple-use applications of energy storage facilities (the "MUA Decision").

Seller shall have provided to Buyer documentation of Seller's compliance (h) with the dual participation requirements of Section 5.4(f).

(i) Seller shall have delivered to Buyer all insurance documents required under Article 16.

Seller shall have provided Delivery Term Security to Buyer as required by (j) Section 10.2.

Seller shall have paid Buyer for all amounts owing under this Agreement, (k) if any, including Delay Damages

(1)No Event of Default on the part of Seller shall have occurred and be continuing and no Remediation Event shall have occurred and remain unresolved.

Seller shall have satisfied the following conditions specific to Load (m)Modifying Resources:

(i) No later than the Dispatch Plan Deadline, Seller shall have delivered to Buyer a Load Modifying Resource Dispatch Plan for the first year of the Delivery Term.

Cooperation in Connection with Initial Delivery Date. The Parties agree that, 4.3 in order for Seller to achieve the Initial Delivery Date, the Parties may have to perform certain of their Delivery Term obligations in advance of the Initial Delivery Date. The Parties shall cooperate with each other in order for Buyer to be able to utilize the Contract Amounts of Product beginning on the applicable Initial Delivery Date.

4.4 <u>Confirmation of Initial Delivery Dates</u>. Once each of the Conditions Precedent has been satisfied by Seller (other than any Condition Precedent that may only be satisfied as of the Initial Delivery Date), Seller shall certify such satisfaction to Buyer in a Notice confirming the anticipated occurrence of the applicable Initial Delivery Date. Buyer shall Notify Seller of any disagreement that Seller has satisfied such Conditions Precedent (with reasonable detail in regard to each Condition Precedent) within fifteen (15) Business Days of Seller's Notice. No later than seven (7) Business Days following the applicable Initial Delivery Date, Buyer shall provide a Notice to Seller confirming the occurrence of the applicable Initial Delivery Date.

Article 5: TRANSACTION

5.1 **<u>Product</u>**.

(a) During the Term, Seller grants, pledges, assigns and otherwise commits and shall deliver exclusively to Buyer, all Load Modification or, upon a Product Transition pursuant to <u>Section 5.2(b)</u>, all Capacity Attributes, in each case that may be calculated or derived from the Operational Characteristics, which must be exclusively from the Projects in the Portfolio (collectively, the "<u>Product</u>"), pursuant to the terms and conditions contained herein (including Portfolio Modification provisions in Section 5.4(e)). Operational Characteristics shall not be modified during the Term.

(b) Buyer shall have the right to re-sell all or a portion of the Product purchased under this Agreement ("**Resold Product**").

5.2 **Purchase and Sale Obligation**.

(a) <u>Initial Product.</u> During the Delivery Term, unless and until a Product Transition occurs under <u>Section 5.2(b)</u>, Seller shall deliver to Buyer all Load Modification, including in the Contract Amounts, pursuant to the protocols and in consideration of the payments set forth in <u>Section 5.2(c)</u>.

(b) <u>Product Transition</u>. At any time during the Delivery Term, but only upon the occurrence of a Product Transition Trigger, Buyer may, by no less than 75 days written notice to Seller, elect to receive Capacity Attributes rather than Load Modification from the Project (the "<u>Product Transition</u>"). Upon receipt of a Product Transition notice from Buyer, and as a condition to Buyer's obligation to take and pay for Capacity Attributes for the remainder of the Delivery Term, Seller shall within ninety (90) days satisfy the Capacity Attribute Conditions set forth in Part One of Appendix XII. Commencing on the first day of the First Showing Month after Seller's satisfaction of the Capacity Attribute Conditions, and for each day of each Showing Month during the remainder of the Delivery Term, Seller shall deliver to Buyer and Buyer shall pay for all Capacity Attributes of Product, including in the Contract Amounts. pursuant to the protocols and in consideration of the payments set forth in Part Two of Appendix XII. Upon a Product Transition, and for remainder of the Delivery Term the Product shall be Capacity Attributes and not Load Modification and Seller shall have no further obligation to provide, or to comply with any provision of this Agreement that is limited to, Load Modification. For clarity, the Product Transition may occur a maximum of one (1) time during the Delivery Term.

(c) <u>Payment</u>. The provisions of this <u>Section 5.2(c)</u> shall apply unless a Product Transition Occurs, in which case the provisions of Part Two of Appendix XII shall become effective and replace this <u>Section 5.2(c)</u>.

(i) Each monthly invoice prepared by Seller pursuant to Article 9 shall include meter data substantiating the quantity of Load Modification, in kWh, delivered to Buyer from the Project consistent with the Load Modifying Resource Dispatch Plan for the applicable month (the "<u>Delivered Quantities</u>").

(ii) For all Load Modification that Seller delivers during the Delivery Term in accordance with this Agreement, Buyer shall, in accordance with Article 9, pay Seller a monthly payment ("Monthly Payment" or "MP") as follows:

 $MP = DQ \times CP$

where,

DQ = The Delivered Quantities;

and

CP = The contract price set forth on the Cover Sheet ("Contract Price").

(iii) For clarity, Buyer has no obligation to pay Seller for Load Modification that is delivered outside the parameters of the applicable Load Modifying Resource Dispatch Plan.

5.3 Allocation of CAISO Payments and Costs.

(a) Except as may otherwise be provided in this Agreement, Seller shall retain any revenues it may receive from and pay all costs charged by, the CAISO or any other third party with respect to the Product or Project.

(b) To the extent that the Project is subject to the terms of the Availability Standards, Non-Availability Charges, and Availability Incentive Payments of the CAISO Tariff, the Parties agree that any Availability Incentive Payments are for the benefit of Seller and for Seller's account and that any Non-Availability Charges are the responsibility of Seller and for Seller's account.

5.4 <u>Customers</u>.

(a) <u>Seller Obligation to Obtain Customers</u>. Seller shall obtain or cause to be obtained the Customers necessary to enable the safe and reliable delivery of Product in the Contract Amounts to Buyer during the Delivery Term. Seller shall enter into a Customer Agreement with each such Customer, complete all necessary registration forms regarding such Customers, submit any necessary documentation regarding such Customers and comply with all other requirements of the Utility Distribution Company, CAISO, CPUC or any other applicable Governmental Authority. In connection with but without limiting the foregoing, Seller shall at all times comply with the Co-Marketing Agreement.

(b) <u>Customers Eligible for Inclusion in Project</u>. Product provided to Buyer must come exclusively from Projects installed at Sites listed in Seller's Portfolio List in accordance with <u>Section 5.4(d)</u>. Unless a Product Transition has occurred, Customers must meet the definition of a "Customer" during and throughout the Delivery Term in order to be included in Seller's Portfolio List.

(c) <u>Seller's Portfolio</u>. Seller shall cause Seller's Portfolio at all times during the Delivery Term to contain at least twenty percent (20%) of Contract Capacity as delivered from assets associated with DAC, LMI, CARE/FERA or Medical Baseline Customer accounts.

(d) <u>Seller's Portfolio List</u>. As of the date first submitted in accordance with <u>Section 4.2</u> and throughout the Delivery Term, Seller shall maintain a list of Customers in Seller's Portfolio in the form and containing the information set forth in the Seller's Portfolio List in <u>Appendix III</u> ("Progress Reporting Form"). If any of the material Customer information in Seller's Portfolio List materially changes during the Delivery Term, such change shall be deemed a Portfolio Modification pursuant with <u>Section 5.4(e)</u> and upon completion of such Portfolio Modification, Seller shall submit to Buyer an updated Seller's Portfolio List reflecting all changes since the previous Seller's Portfolio List. In addition, Seller shall provide any additional Customer information reasonably requested by Buyer in connection with this Agreement.

(e) <u>Portfolio Modification</u>. The Parties agree and acknowledge that Seller may add or remove a Customer from Seller's Portfolio (a "<u>Portfolio Modification</u>") at any time during the Delivery Term, subject to the requirements of this Section 5.4 and Article 11, and any such Portfolio Modifications included in the next issued Quarterly Report. A Portfolio Modification may not alter (i) the Portfolio from the Portfolio description set forth in <u>Appendix I</u> as of the Execution Date, (ii) the Operational Characteristics, or (iii) the Portfolio's ability to deliver Product in the Contract Amounts throughout the Delivery Term. A Portfolio Modification will not alter nor relieve any of Seller's obligations under this Agreement.

(f) <u>Dual Participation</u>. Seller may include in Seller's Portfolio those Customers that are registered in programs or resources administered by Buyer, the CPUC, the CAISO, the Utility Distribution Company, or applicable Governmental Authority ("<u>Other</u> <u>Programs</u>"), provided that (i) participation of Customers in both Seller's Portfolio and Other Programs does not impair (in whole or in part) Seller's ability to perform its obligations under this Agreement (including <u>Section 14.4</u>) and (ii) Seller complies with all rules and requirements of Other Programs set forth by Buyer, the CPUC, the CAISO, the Utility Distribution Company, or applicable Governmental Authority. For the avoidance of doubt, a failure by Seller or a Customer to achieve participation in Other Programs shall not in any way limit or excuse Seller's obligations to Buyer under this Agreement.

(g) <u>Seller's Relationship with Customer</u>. The terms and conditions of the Customer Agreements governing the relationship between Seller and a Customer with respect to such Customer's participation in Seller's Portfolio are independent of Buyer, and Buyer shall have no responsibility with respect to such Customers for purposes of Seller's Portfolio. Seller shall independently resolve any disputes arising between Seller and any Customer.

Article 6: OPERATIONS

6.1 **Operations**. Seller shall at all times retain operational control of the Project and be responsible for operation and maintenance of the Project, and Buyer shall have no liability for the failure of Seller, any Customer, or any Project owner or operator to comply with any applicable Law, Requirements, or other requirement of the Transmission Provider or Utility Distribution Company, including any penalties, charges or fines imposed for such noncompliance.

(a) **Islanding**: Projects are required to be "islandable" and be able to provide back-up power to Customers in the event of a power outage.

6.2 <u>Charging Energy</u>. As between Buyer and Seller, Seller shall be responsible for procuring and delivering all of the Charging Energy to the Project and paying all of the associated costs of such Charging Energy.

6.3 <u>Standard of Care</u>. In performing all of its obligations under this Agreement, including in its scheduling, interconnection, operation and maintenance of the Project, Seller shall comply with all Requirements and Safety Requirements.

6.4 **Buyer's Use and Certification of Product**.

(a) During the Term, Seller shall take all actions, including executing all documents or instruments, complying with all applicable registration, certification and reporting requirements of all applicable Governmental Authorities and other Persons, as such requirements may be amended from time to time, that are necessary to ensure that Buyer can use Product (including, solely in the event of a Product Transition, enabling Buyer to apply Product towards Buyer's Compliance Obligations), or sell Resold Product, at all times during the Delivery Term. Promptly following Buyer's written request, Seller agrees to take all actions and execute or provide any documents, information, or instruments with respect to Product reasonably necessary to enable Buyer to comply with the requirements of any Governmental Authority.

(b) During the Delivery Term, Seller shall, at no cost to Buyer, obtain and maintain CAISO and all applicable Governmental Authority certification(s) for all elements of the Product for which certification is or may become required in order to enable Buyer to receive and use such Product (including, solely in the event of a Product Transition, use of such Product to satisfy its Compliance Obligations). If Buyer is required under applicable Law to obtain such certification, Seller shall take all actions within its control to ensure that Buyer is able to secure such certification. Seller, at no cost to Buyer, shall take all other actions during the Delivery Term, including submission of all reports and other filings with CAISO and applicable Governmental

Authorities, that are required to be taken by Seller to ensure that Buyer can receive the Product and shall take all actions within its control to assist Buyer in taking actions required to be taken by Buyer with regard to receipt of Product.

6.5 **Dispatch**. The provisions of this <u>Section 6.5</u> shall apply unless a Product Transition Occurs, in which case the provisions of Part Three of Appendix XII ("Scheduling") shall become effective and replace this <u>Section 6.5</u>.

(a) For each year of the Delivery Term, Seller shall submit to Buyer a Load Modifying Resource Dispatch Plan on or before the Dispatch Plan Deadline.

(b) No less than 15 days before the Dispatch Plan Deadline, the Parties shall meet to discuss the Load Modifying Resource Dispatch Plan for the coming year.

(c) Buyer may require seller to modify Dispatch Plan to maximize Load Modification for Buyer outside of Customers' TOU periods upon mutual agreement of an equitable adjustment to the Contract Price to reflect the demonstrable economic loss to Seller or Seller's Customers.

(d) At all times during the Delivery Term, Seller shall operate the Project to provide the Contract Amounts of Load Modification as specified in the applicable Load Modifying Resource Dispatch Plan.

6.6 Information Sharing and Shared Learning. Seller understands and acknowledges that Buyer is entering into this Agreement in part to gain operational and market information regarding the performance, efficiency, operations, maintenance, and multiple uses of energy storage and storage assets as an integral part of Buyer's portfolio of assets to meet its customers' needs as well as to gain an understanding of the impact of energy storage on load forecasting as a load serving entity. Throughout the Term, Seller agrees to share such information with Buyer, including meter data and hourly charging and discharging data at least on an aggregated basis but excluding cost or similar proprietary information, upon Buyer's reasonable request, with such information to be treated by Buyer as Confidential Information. Seller shall provide such applicable meter data to Buyer in a format and to a platform specified by Buyer that is reasonably acceptable to Seller. Except with respect to services, attributes, or products delivered to Customers pursuant to Customer Agreements or products or services related to Renewable Energy Credits (which shall not require any Notice to Buyer), for information related to Seller's multiple uses of the Project, Seller shall promptly provide Notice to Buyer any time it provides any services, attributes, or products from the Project to a third party.

6.7 Changes in Law.

(a)

(i) In the event of any change in Requirements by the CEC, CPUC, CAISO, or other Governmental Authority or Person results in a <u>decrease</u> to the amount of Product that may be calculated or derived from the Operational Characteristics (including any extension by the CEC, CPUC or CAISO of the required duration of energy storage resources for Load Modification or, after a Product Transition, Resource Adequacy), Seller shall exercise

commercially reasonable efforts to maintain the Contract Amounts. If despite such efforts Seller is unable to maintain the Contract Amounts as a result of such change in requirements, then either Party may provide Notice to the other Party, once it is reasonably evident that the Contract Amounts cannot be maintained, specifying the altered amounts of Product ("Downward Change Notice"). Following a Downward Change Notice, Buyer will confirm via Notice to Seller the amended Contract Amounts and Contract Amount based on such change and the date that Seller shall commence delivery of such amended amounts ("Downward Change Confirmation Notice").

(ii) In the event of any change in Requirements by the CEC, CPUC, CAISO, or other Governmental Authority or Person results in an increase to the amount of Product that may be calculated or derived from the Operational Characteristics (including in the case of a new category of Load Modification or after a Product Transition, Capacity Attributes), then either Party shall provide Notice to the other Party as soon as practicable following knowledge of such change specifying the altered amounts of Product ("Upward Change Notice"). Following an Upward Change Notice, Buyer shall have sole discretion over whether to accept a corresponding increase to any Contract Amount and the Contract Amount. If Buyer chooses to accept such increase, Buyer will confirm via Notice to Seller the amended Contract Amounts and Contract Amount based on such change and the date that Seller shall commence delivery of such amended amounts ("Upward Change Confirmation Notice"). If Buyer declines to accept any such increase, Seller shall have the right to sell to third parties any resulting Product that is in excess of the Contract Amounts.

(iii) The Contract Amounts and Contract Amount shall automatically adjust upon the date set forth in the Confirmation Notice without further need for the Parties to amend this Agreement. Until such date, Seller shall continue to deliver the Contract Amounts of Product as stated prior to the Confirmation Notice, unless otherwise required by Law or other Requirements.

(b) In the event a centralized capacity market develops within the WECC region, Buyer will have exclusive right to offer, bid, or otherwise submit the Product for re-sale in such markets, or to cause Seller or Seller's SC to do so, and Buyer shall retain and receive any and all related revenues.

If a change in CAISO, CEC or CPUC Requirements renders this Agreement (c)or any provisions hereof incapable of being performed or administered, then either Party may request that Buyer and Seller enter into negotiations to make the minimum changes to this Agreement necessary to make this Agreement capable of being performed and administered, while attempting to preserve to the maximum extent possible the benefits, burdens, and obligations set forth in this Agreement as of the Execution Date. Upon delivery of such a request, Buyer and Seller shall engage in such negotiations in good faith. If Buyer and Seller are unable, within sixty (60) days after delivery of such request, to agree upon changes to this Agreement or to resolve issues relating to changes to this Agreement, then either Party may submit issues pertaining to changes to this Agreement to the dispute resolution process set forth in Section 19.2. Notwithstanding the foregoing, (i) a change in cost shall not in and of itself be deemed to render this Agreement or any of the provisions hereof incapable of being performed or administered, or constitute, or form the basis of, a Force Majeure, and (ii) all of unaffected provisions of this

Agreement shall remain in full force and effect during any period of such negotiation or dispute resolution.

(d) Without limitation or modification of Section 6.7(a) or Section 6.7(c), in the event of any change in Requirements by the CPUC, CEC, CAISO, or other Governmental Authority or Person that significantly modifies the market mechanisms or regulatory construct for delivery of Load Modification or (after a Product Transition) Capacity Attributes from resources similar to and including the Project (but that does not fall under Section 6.7(c)), either Party may provide Notice to the other Party requesting that the Parties discuss in good faith changes to this Agreement that would enable each Party to preserve to the maximum extent possible the benefits, burdens, and obligations set forth in this Agreement as of the Execution Date. Upon delivery of such a request, Buyer and Seller shall engage in such negotiations in good faith for a period of at least sixty (60) days but shall have no further obligation or rights under this Section 6.7(d).

Article 7: EVENTS OF DEFAULT; REMEDIES

7.1 Events of Default.

(a) Seller will be deemed a Defaulting Party upon the occurrence of any of the following (each a "<u>Seller's Event of Default</u>"):

(i) Seller fails to satisfy a Performance Assurance requirement set forth in <u>Article 10</u> and Seller fails to provide replacement Performance Assurance within ten (10) Business Days of Buyer's written demand therefor in accordance with <u>Article 10</u>;

(ii) any material misrepresentation or omission, in any metering or submetering, Load Modifying Resource Dispatch Plan, Supply Plan, Seller's Portfolio List, report, or Notice with regard to delivery of the Product, or undue delay or withholding of such data, report or Notice, which misrepresentation or undue delay or withholding is not cured within ten (10) Business Days of Buyer's receipt thereof;

(iii) Seller intentionally or knowingly delivers, or attempts to deliver Product that is not produced by the Project, except as specified in Article 5.1;

(iv) Seller fails to achieve Construction Start by the applicable Construction Start Deadline for reasons other than Force Majeure, subject to <u>Section 3.1(d)</u>;

(v) Seller fails (A) to deliver a Delay Notice in accordance with <u>Section</u> <u>4.1(b)(i)</u> or (B) to achieve an Initial Delivery Date by the applicable Expected Initial Delivery Date for reasons other than Force Majeure, subject to <u>Section 4.1(b)(ii)</u>;

(vi) Seller fails in any month to deliver Delivered Quantities to Buyer equal to the Contract Amounts (unless Buyer has received Shortfall LDs in lieu of Delivered Quantities); provided that Shortfall LDs may not be used to avoid a Seller's Event of Default (A) during the months of August and September (unless a Product Transition has occurred) or (B) more than three (3) times in any twelve (12) month period during the Delivery Term; or (vii) In any year in the Delivery Term, aggregate Delivered Quantities are less than eighty five percent (85%) of the Contract Amounts for such period.

(a) Either Party will be deemed a Defaulting Party upon the occurrence of any of the following (each a "<u>Party's Event of Default</u>"):"

(viii) the failure by such Party to make, when due, any payment required pursuant to this Agreement and such failure is not remedied within ten (10) Business Days after Notice thereof;

(ix) any representation or warranty made by such Party under this Agreement is false or misleading in any material respect when made or when deemed made or repeated, and such default is not remedied within thirty (30) days after Notice thereof, or such longer additional period, not to exceed an additional sixty (60) days, if the Defaulting Party is unable to remedy such default within such initial thirty (30) days period despite exercising best efforts;

(x) the failure by such Party to perform any material covenant or obligation set forth in this Agreement, including any failure by Seller to comply with all Requirements and Safety Requirements in accordance with <u>Section 6.3</u>, except to the extent constituting a separate Event of Default, and such failure is not remedied within thirty (30) days after Notice thereof, or such longer additional period, not to exceed an additional ninety (90) days, if the Defaulting Party is unable to remedy such default within such initial thirty (30) days period despite exercising best efforts;

(xi) such Party becomes Bankrupt;

(xii) such Party assigns this Agreement or any of its rights hereunder other than in compliance with <u>Article 18;</u> or

(xiii) such Party consolidates or amalgamates with, or merges with or into, or transfers all or substantially all of its assets to, another entity and, at the time of such consolidation, amalgamation, merger or transfer, the resulting, surviving or transferee entity fails to assume all the obligations of such Party under this Agreement to which it or its predecessor was a party by operation of Law or pursuant to an agreement reasonably satisfactory to the other Party.

7.2 <u>Remedies; Declaration of Early Termination Date</u>. If an Event of Default with respect to a Defaulting Party has occurred, the other Party ("<u>Non-Defaulting Party</u>") shall have the following rights:

(a) to send Notice, designating a day, no earlier than the day such Notice is deemed to be received and no later than twenty (20) days after such Notice is deemed to be received, as an early termination date of this Agreement ("<u>Early Termination Date</u>") that terminates this Agreement (the "<u>Terminated Transaction</u>") and ends the Delivery Term effective as of the Early Termination Date;

(b) to accelerate all amounts owing between the Parties, and to collect as liquidated damages (i) the Damage Payment, in the case of an Event of Default by Seller occurring

before the Initial Delivery Date, including an Event of Default under Section 7.1(a)(v), or (ii) the Termination Payment calculated in accordance with Section 7.3 below, in the case of any other Event of Default by either Party;

- (c) to withhold any payments due to the Defaulting Party under this Agreement;
- (d) to suspend performance; or

(e) to exercise any other right or remedy available at law or in equity, including specific performance or injunctive relief, except where an express and exclusive remedy or measure of damages is provided under this Agreement;

(f) <u>provided</u>, that payment by the Defaulting Party of the Damage Payment or Termination Payment, as applicable, shall constitute liquidated damages and the Non-Defaulting Party's sole and exclusive remedy for any Terminated Transaction and the Event of Default related thereto.

Termination Payment. The Termination Payment ("Termination Payment") for 7.3 a Terminated Transaction shall be the aggregate of the Settlement Amount plus any or all other amounts due to or from the Non-Defaulting Party, as of the Early Termination Date, netted into a single amount. The Non-Defaulting Party shall calculate a Settlement Amount for the Terminated Transaction as of the Early Termination Date. Third parties supplying information for purposes of the calculation of Gains or Losses may include, without limitation, dealers in the relevant markets, end-users of the relevant product, information vendors and other sources of market information. The Settlement Amount shall not include consequential, incidental, punitive, exemplary, indirect or business interruption damages. Without prejudice to the Non-Defaulting Party's duty to mitigate, the Non-Defaulting Party shall not have to enter into replacement transactions to establish a Settlement Amount. Each Party agrees and acknowledges that (a) the actual damages that the Non-Defaulting Party would incur in connection with a Terminated Transaction would be difficult or impossible to predict with certainty, (b) the Damage Payment or Termination Payment described in Section 7.2 or this Section 7.3, as applicable, is a reasonable and appropriate approximation of such damages, and (c) the Damage Payment or Termination Payment described in Section 7.2 or this Section 7.3, as applicable, is the exclusive remedy of the Non-Defaulting Party in connection with a Terminated Transaction but shall not otherwise act to limit any of the Non-Defaulting Party's rights or remedies if the Non-Defaulting Party does not elect a Terminated Transaction as its remedy for an Event of Default by the Defaulting Party.

7.4 **Notice of Payment of Termination Payment**. As soon as practicable after a Terminated Transaction, Notice shall be given by the Non-Defaulting Party to the Defaulting Party of the amount of the Damage Payment or Termination Payment and whether the Termination Payment is due to or from the Non-Defaulting Party. The Notice shall include a written statement explaining in reasonable detail the calculation of such amount and the sources for such calculation. The Termination Payment shall be made to or from the Non-Defaulting Party, as applicable, within ten (10) Business Days after such Notice is effective.

7.5 **Disputes with Respect to Termination Payment**. If the Defaulting Party disputes the Non-Defaulting Party's calculation of the Termination Payment, in whole or in part, the

Defaulting Party shall, within five (5) Business Days of receipt of the Non-Defaulting Party's calculation of the Termination Payment, provide to the Non-Defaulting Party a detailed written explanation of the basis for such dispute. Disputes regarding the Termination Payment shall be determined in accordance with <u>Article 19</u>.

7.6 **<u>Rights and Remedies Are Cumulative</u>**. Except where an express and exclusive remedy or measure of damages is provided, the rights and remedies of a Party pursuant to this <u>Article 7</u> shall be cumulative and in addition to the rights of the Parties otherwise provided in this Agreement.

Article 8: FORCE MAJEURE

8.1 Force Majeure.

(a) <u>Effect of Force Majeure</u>. A Party shall not be considered to be in default in the performance of its obligations to the extent that the failure or delay of its performance is due to a Force Majeure event, and the non-affected Party shall be excused from its corresponding performance obligations for the period of the affected Party's failure or delay of performance. The burden of proof for establishing the existence and consequences of an event of Force Majeure lies with the Party initiating the claim.

(b) <u>Notice of Force Majeure</u>. The Party desiring to invoke the Force Majeure event as a cause for delay in its performance of, or failure to perform, any obligation hereunder, shall provide prompt Notice to the other Party in the form of a letter identifying the event of Force Majeure and describing in detail the particulars of the occurrence giving rise to the Force Majeure event including the expected duration, when known, and the effect of such Force Majeure event. Failure to provide timely Notice constitutes a waiver of a claim of Force Majeure. Promptly, but in any event within ten (10) days after a Notice is given pursuant to the preceding sentence, the Parties shall meet to discuss the basis and terms upon which the arrangements set out in this Agreement shall be continued taking into account the effects of such event of Force Majeure.

(c) <u>Mitigation of Force Majeure</u>. The suspension of a Party's performance under the Agreement due to a claim of Force Majeure shall be of no greater scope and of no longer duration than is required by the Force Majeure event. A Party suspending performance due to Force Majeure shall take, or cause to be taken, such action as may be necessary to void, or nullify, or otherwise to mitigate, in all material respects, the effects of such event of Force Majeure. The Parties shall take all reasonable steps to resume normal performance under this Agreement after the cessation of any Force Majeure event. If Seller cannot meet the Expected Initial Delivery Date as a result of a Force Majeure event declared by Seller in accordance with this Agreement, then Seller shall work diligently to resolve the effect of the Force Majeure and provide evidence of its efforts promptly upon Buyer's written request.

(d) <u>Force Majeure Failure</u>. Subject to <u>Section 8.1(a)</u>, Buyer shall have the right, but not the obligation, to terminate this Agreement after the occurrence of the following, each constituting a "<u>Force Majeure Failure</u>":

(i) if during the Delivery Term:

(A) due to a Force Majeure event, Seller is unable to meet its obligations under this Agreement (including any failure to deliver Delivered Quantities to Buyer equal to the Contract Amounts) for a period greater than one hundred eighty (180) days; or

of Force Majeure.

(B) the Project is destroyed or rendered inoperable by an event

(ii) if Seller is unable, due solely to a Force Majeure event, to achieve an Initial Delivery Date by one hundred eighty (180) days after the applicable Expected Initial Delivery Date.

(e) <u>Effect of Termination for Force Majeure Failure</u>. If Buyer exercises its termination right in connection with a Force Majeure Failure under <u>Section 8.1(d)</u>, then the Agreement shall terminate without further liability of either Party to the other, effective upon the date set forth in Buyer's Notice of termination, subject to each Party's satisfaction of all of the final payment and survival obligations set forth in <u>Section 21.3</u>.

Article 9: INVOICING AND PAYMENT

9.1 <u>Invoicing</u>. Seller shall make good faith efforts to deliver an invoice, in arrears, to Buyer no sooner than ten (10) Business Days after the end of each month of the Delivery Term for all amounts due from Buyer to Seller under this Agreement, including, as applicable: (a) the Monthly Payment, and (b) other compensatory adjustments required by this Agreement, including adjustments for Taxes. Each invoice shall (i) contain data sufficient to document and verify all amounts included therein, including any relevant records, invoices, or settlement data from CAISO, necessary to verify the accuracy of any amount and (ii) be in a format specified by Buyer.

9.2 **Payment**. Buyer shall make payment to Seller by wire transfer or ACH payment to the bank account provided on each monthly invoice. Buyer shall pay undisputed invoice amounts within forty-five (45) days after receipt of the invoice. If such due date falls on a weekend or legal holiday, such due date shall be the next Business Day. Payments made after the due date will be considered late and will bear interest on the unpaid balance. If the amount due is not paid on or before the due date or if any other payment that is due and owing from one Party to another is not paid on or before its applicable due date, a late payment charge shall be applied to the unpaid balance and shall be added to the next billing statement. Such late payment charge shall be calculated based on an annual interest rate equal to the three-month LIBOR, plus two percent (2%) (the "Interest Rate"). If the due date occurs on a day that is not a Business Day, the late payment charge shall begin to accrue on the next succeeding Business Day.

9.3 **Payment Adjustments: Billing Errors.** Payment adjustments shall be made if Buyer or Seller discovers there have been good faith inaccuracies in invoicing that are not otherwise disputed under <u>Section 9.4</u> or an adjustment to an amount previously invoiced or paid is required due to a correction of data by CAISO. If the required adjustment is in favor of Buyer, Buyer's next monthly payment shall be credited in an amount equal to the adjustment. If the required adjustment is in favor of Seller, Seller shall add the adjustment amount to Buyer's next monthly invoice. Adjustments in favor of either Buyer or Seller shall bear interest, until settled in full, in accordance with <u>Section 9.2</u>, accruing from the date on which the adjusted amount should have been due.

9.4 **Billing Disputes.** A Party may, in good faith, dispute the correctness of any invoice or any adjustment to an invoice rendered under this Agreement or adjust any invoice for any arithmetic or computational error within twelve (12) months of the date of the invoice, or adjustment to an invoice, was rendered. In the event an invoice or portion thereof, or any other claim or adjustment arising hereunder, is disputed, payment of the undisputed portion of the invoice shall be required to be made when due. Any invoice dispute or invoice adjustment shall be in writing and shall state the basis for the dispute or adjustment. Payment of the disputed amount shall not be required until the dispute is resolved. Upon resolution of the dispute, any required payment shall be made within five (5) Business Days of such resolution along with interest accrued at the Interest Rate from and including the original due date to but excluding the date paid. Inadvertent overpayments shall be returned via adjustments in accordance with Section 9.3. Any dispute with respect to an invoice is waived if the other Party is not notified in accordance with this Section 9.4 within twelve (12) months after the invoice is rendered or subsequently adjusted, except to the extent any misinformation was from a third party not affiliated with any Party and such third party corrects its information after the twelve (12) month period. If an invoice is not rendered within twelve (12) months after the close of the month during which performance occurred, the right to payment for such performance is waived.

9.5 <u>Netting of Payments</u>. The Parties hereby agree that they shall discharge mutual debts and payment obligations due and owing to each other on the same date through netting, in which case all amounts owed by each Party to the other Party for the purchase and sale of Product during the monthly billing period under this Agreement or otherwise arising out of this Agreement, including any related damages, interest, and payments or credits, shall be netted so that only the excess amount remaining due shall be paid by the Party who owes it.

Article 10: PERFORMANCE ASSURANCE

Seller's Development Security. To secure its obligations under this Agreement, 10.1 Seller shall deliver Development Security to Buyer within thirty (30) Business Days of the Execution Date. Seller shall maintain the Development Security in full force and effect and Seller shall within five (5) Business Days after any draw thereon replenish the Development Security in the event Buyer collects or draws down any portion of the Development Security for any reason permitted under this Agreement other than to satisfy a Termination Payment. Upon the earlier of (a) Seller's delivery of the Delivery Term Security, or (b) sixty (60) days after termination of this Agreement, Buyer shall return the Development Security to Seller, less the amounts drawn in accordance with this Agreement. If the Development Security is a Letter of Credit and the issuer of such Letter of Credit (i) fails to maintain the minimum Credit Rating specified in the definition of Letter of Credit, (ii) indicates its intent not to renew such Letter of Credit and such Letter of Credit expires prior to the Stage 1 Commercial Operation Deadline, or (iii) fails to honor Buyer's properly documented request to draw on such Letter of Credit by such issuer, Seller shall have ten (10) Business Days to either post cash or deliver a substitute Letter of Credit that meets the requirements set forth in the definition of Development Security.

Seller's Delivery Term Security. To secure its obligations under this Agreement, 10.2 Seller shall deliver Delivery Term Security to Buyer on or before the Initial Delivery Date. Seller shall maintain the Delivery Term Security in full force and effect, and shall within five (5) Business Days after any draws made by Buyer in accordance with this Agreement (other than to satisfy a Termination Payment) replenish the Delivery Term Security, until the following have occurred: (a) the Delivery Term has expired or terminated early; and (b) all payment obligations of Seller then due and payable under this Agreement, including compensation for penalties, Termination Payment, indemnification payments or other damages are paid in full (whether directly or indirectly such as through set-off or netting). Following the occurrence of both events, Buyer shall promptly return to Seller the unused portion of the Delivery Term Security. If the Delivery Term Security is a Letter of Credit and the issuer of such Letter of Credit (i) fails to maintain the minimum Credit Rating set forth in the definition of Letter of Credit, (ii) indicates its intent not to renew such Letter of Credit and such Letter of Credit expires prior to the end of the Delivery Term, or (iii) fails to honor Buyer's properly documented request to draw on such Letter of Credit by such issuer, Seller shall have ten (10) Business Days to either post cash or deliver a substitute Letter of Credit that meets the requirements set forth in the definition of Delivery Term Security.

10.3 First Priority Security Interest in Cash or Cash Equivalent Collateral.

To secure its obligations under this Agreement, and until released as (a) provided herein, Seller hereby grants to Buyer a present and continuing first-priority security interest ("Security Interest") in, and lien on (and right to net against), and assignment of the Development Security, Delivery Term Security, any other cash collateral and cash equivalent collateral posted pursuant to Sections 10.1 and 10.2 and any and all interest thereon or proceeds resulting therefrom or from the liquidation thereof, whether now or hereafter held by, on behalf of, or for the benefit of Buyer, and Seller agrees to take all action as Buyer reasonably requires in order to perfect Buyer's Security Interest in, and lien on (and right to net against), such collateral and any and all proceeds resulting therefrom or from the liquidation thereof. Upon or any time after the occurrence of a Seller's Event of Default or a Party's Event of Default on the part of Seller, an Early Termination Date resulting from a Seller's Event of Default or a Party's Event of Default on the part of Seller, or an occasion provided for in this Agreement where Buyer is authorized to retain all or a portion of the Development Security or Delivery Term Security, Buyer may do any one or more of the following (in each case subject to the final sentence of this Section 10.3):

(i) Exercise any of its rights and remedies with respect to the Development Security and Delivery Term Security, including any such rights and remedies under Law then in effect;

(ii) Draw on any outstanding Letter of Credit issued for its benefit and retain any cash held by Buyer as Development Security or Delivery Term Security; and

(iii) Liquidate all Development Security or Delivery Term Security (as applicable) then held by or for the benefit of Buyer free from any claim or right of any nature whatsoever of Seller, including any equity or right of purchase or redemption by Seller.

(b) Buyer shall apply the proceeds of the collateral realized upon the exercise of any such rights or remedies to reduce Seller's obligations under this Agreement (Seller remains liable for any amounts owing to Buyer after such application), subject to Buyer's obligation to return any surplus proceeds remaining after these obligations are satisfied in full.

10.4 <u>Seller Financial Information</u>. If requested by Buyer, Seller shall deliver to Buyer (a) within one hundred twenty (120) days following the end of each fiscal year, a copy of Seller's annual report containing unaudited consolidated financial statements for such fiscal year, if available, (or audited consolidated financial statements for such fiscal year if otherwise available) and (b) within sixty (60) days after the end of each of its first three fiscal quarters of each fiscal year, a copy of such Seller's quarterly report containing unaudited consolidated financial statements for such fiscal quarter, if available. In all cases the statements shall be for the most recent accounting period and shall be prepared in accordance with Generally Accepted Accounting Principles. Notwithstanding anything else herein to the contrary, to the extent Seller is a publicly traded corporation on a nationally recognized stock exchange, the obligation to deliver financial statements pursuant to Section 10.4 herein shall be waived.

Article 11: SAFETY

11.1 <u>Safety</u>.

(a) Seller shall, and shall cause its Affiliates and Contractors to, design, construct, operate, and maintain the Project and conduct all Work or cause all Work to be conducted in accordance with the Safety Requirements. Seller shall, and shall cause its Affiliates and Contractors to, take all actions to comply with the Safety Requirements.

(b) Prior to Seller's execution of a Contractor's contract, Seller shall demonstrate to Buyer that the Contractor has the qualifications, experience, and safety record to develop, construct, operate and maintain the Project, as applicable. During the period that a Contractor is conducting Work, Seller shall cause the Contractor to meet and comply with the Safety Requirements.

(c) Seller shall document a Portfolio Safety Plan and incorporate the Portfolio Safety Plan's features into the design, development, construction, operation, and maintenance of the Project. Seller shall submit for Buyer's review a Portfolio Safety Plan, in a format acceptable to Buyer, which must demonstrate (A) Seller's plans to comply with the Safety Requirements and (B) Seller's consideration of the Portfolio Safety Plan items in Part Two (Project Design and Description) of <u>Appendix VI</u>. Upon Notice to Buyer, Seller may deviate from any specific procedures identified in the Portfolio Safety Plan while designing, developing, constructing, operating, or maintaining the Project, if in Seller's judgment, the deviation is necessary to design, develop, construct, operate, or maintain the Project safely or in accordance with the Safety Requirements.

(d) Throughout the Delivery Term, Seller shall update the Safeguards and the Portfolio Safety Plan as required by Safety Requirements or as necessitated by a Safety Remediation Plan. Seller shall provide such updated Portfolio Safety Plan to Buyer within thirty (30) days of any such updates. Throughout the Delivery Term, Buyer shall have the right to request Seller to provide its Portfolio Safety Plan, or portions thereof, and demonstrate its compliance with the Safety Requirements within thirty (30) days of Buyer's Notice.

(e) Seller shall remove any Contractor that engages in repeated, material violations of the Portfolio Safety Plan or Safety Requirements, unless doing so would present an ongoing material adverse effect to the operation of the Project.

11.2 **<u>Reporting Serious Incidents</u>**. Seller shall provide Notice of a Serious Incident to Buyer within five (5) Business Days of occurrence. The Notice of Serious Incident must include the time, date, and location of the incident, the Contractor or Customer(s) involved in the incident (as applicable), the circumstances surrounding the incident, the immediate response and recovery actions taken, and a description of any impacts of the Serious Incident. Seller shall cooperate and provide reasonable assistance, and cause each of its Contractors and Customer(s) to cooperate and provide reasonable assistance, to Buyer with any investigations and inquiries by Governmental Authorities that arise as a result of the Serious Incident.

11.3 **<u>Remediation</u>**.

(a) Seller shall resolve any Remediation Event within the Remediation Period. Within ten (10) Business Days of the date of the first occurrence of any Remediation Event, Seller shall provide a Safety Remediation Plan to Buyer for Buyer's review.

(b) Seller shall cooperate, and cause each of its Contractors to cooperate, with Buyer in order for Seller to provide any report relating to a Remediation Event, in a form and level of detail that is acceptable to Buyer which incorporates information, analysis, investigations or documentation, as applicable or as requested by Buyer.

Article 12: TAXES

12.1 **Taxes**. Seller shall pay or cause to be paid all Taxes (a) on or with respect to the Portfolio and (b) on or with respect to the sale and making available of Product to Buyer that are imposed on Product prior to its delivery to Buyer at the time and place contemplated under this Agreement. Buyer shall pay or cause to be paid all Taxes on or with respect to the delivery to and purchase by Buyer of Product that are imposed on Product at and after its delivery to Buyer at the time and place contemplated under this Agreement (other than withholding or other Taxes imposed on Seller's income, revenue, receipts or employees), if any. If a Party is required to remit or pay Taxes that are the other Party's responsibility hereunder, such Party shall promptly pay the Taxes due and then seek and receive reimbursement from the other for such Taxes. In the event any sale of Product hereunder is exempt from or not subject to any particular Tax, Buyer shall provide Seller with all necessary documentation within thirty (30) days after the Execution Date to evidence such exemption or exclusion. If Buyer does not provide such documentation, then Buyer shall indemnify, defend, and hold Seller harmless from any liability with respect to Taxes from which Buyer claims it is exempt.

12.2 <u>Cooperation</u>. Each Party shall use reasonable efforts to implement the provisions of and administer this Agreement in accordance with the intent of the Parties to minimize all Taxes, so long as no Party is materially adversely affected by such efforts. The Parties shall cooperate to minimize Tax exposure; *provided, however*, that neither Party shall be obligated to incur any

financial or operational burden to reduce Taxes for which the other Party is responsible hereunder without receiving due compensation therefor from the other Party.

Article 13: LIMITATIONS

13.1 **No Consequential Damages**. EXCEPT TO THE EXTENT PART OF AN EXPRESS REMEDY OR MEASURE OF DAMAGES HEREIN, OR PART OF AN ARTICLE 15 INDEMNITY CLAIM, OR INCLUDED IN A LIQUIDATED DAMAGES CALCULATION, OR ARISING FROM FRAUD OR INTENTIONAL MISREPRESENTATION, NEITHER PARTY SHALL BE LIABLE TO THE OTHER OR ITS INDEMNIFIED PERSONS FOR ANY SPECIAL, PUNITIVE, EXEMPLARY, INDIRECT, OR CONSEQUENTIAL DAMAGES, OR LOSSES OR DAMAGES FOR LOST REVENUE OR LOST PROFITS, WHETHER FORESEEABLE OR NOT, ARISING OUT OF, OR IN CONNECTION WITH THIS AGREEMENT.

13.2 Waiver and Exclusion of Other Damages.

(a) EXCEPT AS EXPRESSLY SET FORTH HEREIN, THERE IS NO WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, AND ANY AND ALL IMPLIED WARRANTIES ARE DISCLAIMED. THE PARTIES CONFIRM THAT THE EXPRESS REMEDIES AND MEASURES OF DAMAGES PROVIDED IN THIS AGREEMENT SATISFY THE ESSENTIAL PURPOSES HEREOF. ALL LIMITATIONS OF LIABILITY CONTAINED IN THIS AGREEMENT, INCLUDING, WITHOUT LIMITATION, THOSE PERTAINING TO SELLER'S LIMITATION OF LIABILITY AND THE PARTIES' WAIVER OF CONSEQUENTIAL DAMAGES, SHALL APPLY EVEN IF THE REMEDIES FOR BREACH OF WARRANTY PROVIDED IN THIS AGREEMENT ARE DEEMED TO "FAIL OF THEIR ESSENTIAL PURPOSE" OR ARE OTHERWISE HELD TO BE INVALID OR UNENFORCEABLE.

(b) FOR BREACH OF ANY PROVISION FOR WHICH AN EXPRESS AND EXCLUSIVE REMEDY OR MEASURE OF DAMAGES IS PROVIDED, SUCH EXPRESS REMEDY OR MEASURE OF DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, THE OBLIGOR'S LIABILITY SHALL BE LIMITED AS SET FORTH IN SUCH PROVISION, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED.

(c) TO THE EXTENT ANY DAMAGES REQUIRED TO BE PAID HEREUNDER ARE LIQUIDATED, INCLUDING UNDER <u>SECTIONS 3.1(d)</u>, <u>4.1(c)</u>, <u>7.1(a)(vi)</u>, <u>7.2</u> AND <u>7.3</u>, THE PARTIES ACKNOWLEDGE THAT THE DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE, THAT OTHERWISE OBTAINING AN ADEQUATE REMEDY IS INCONVENIENT, AND THAT THE LIQUIDATED DAMAGES CONSTITUTE A REASONABLE APPROXIMATION OF THE ANTICIPATED HARM OR LOSS. IT IS THE INTENT OF THE PARTIES THAT THE LIMITATIONS HEREIN IMPOSED ON REMEDIES AND THE MEASURE OF DAMAGES BE WITHOUT REGARD TO THE CAUSE OR CAUSES RELATED THERETO, INCLUDING THE NEGLIGENCE OF ANY PARTY, WHETHER SUCH NEGLIGENCE BE SOLE, JOINT OR CONCURRENT, OR
ACTIVE OR PASSIVE. THE PARTIES HEREBY WAIVE ANY RIGHT TO CONTEST SUCH PAYMENTS AS AN UNREASONABLE PENALTY.

(d) THE PARTIES ACKNOWLEDGE AND AGREE THAT MONEY DAMAGES AND THE EXPRESS REMEDIES PROVIDED FOR HEREIN ARE AN ADEQUATE REMEDY FOR THE BREACH BY THE OTHER OF THE TERMS OF THIS AGREEMENT, AND EACH PARTY WAIVES ANY RIGHT IT MAY HAVE TO SPECIFIC PERFORMANCE WITH RESPECT TO ANY OBLIGATION OF THE OTHER PARTY UNDER THIS AGREEMENT.

Article 14: REPRESENTATIONS; WARRANTIES; COVENANTS

14.1 <u>Seller's Representations and Warranties</u>. As of the Execution Date, Seller represents and warrants as follows:

(a) Seller is a Corporation, duly organized, validly existing and in good standing under the laws of the State of Delaware, and is qualified to conduct business in the State of California and each jurisdiction where the failure to so qualify would have a material adverse effect on the business or financial condition of Seller.

(b) Seller has the power and authority to enter into and perform this Agreement and is not prohibited from entering into this Agreement or discharging and performing all covenants and obligations on its part to be performed under and pursuant to this Agreement. The execution, delivery and performance of this Agreement by Seller has been duly authorized by all necessary action on the part of Seller (evidence of such due authorization Seller shall provide to Buyer if requested) and does not and will not require the consent of any trustee or holder of any indebtedness or other obligation of Seller or any other party to any other agreement with Seller.

(c) The execution and delivery of this Agreement, consummation of the transactions contemplated herein, and fulfillment of and compliance by Seller with the provisions of this Agreement will not conflict with or constitute a breach of or a default under any Law presently in effect having applicability to Seller, subject to any permits that have not yet been obtained by Seller, the documents of formation of Seller or any outstanding trust indenture, deed of trust, mortgage, loan agreement or other evidence of indebtedness or any other agreement or instrument to which Seller is a party or by which any of its property is bound.

(d) This Agreement has been duly executed and delivered by Seller. This Agreement is a legal, valid and binding obligation of Seller enforceable in accordance with its terms, except as limited by laws of general applicability limiting the enforcement of creditors' rights or by the exercise of judicial discretion in accordance with general principles of equity.

(e) The Project is located in the State of California.

(f) Seller will be responsible for obtaining all permits necessary to construct and operate the Project and Seller will be the applicant on any California Environmental Quality Act documents. (g) Seller is familiar with conflict of interest Laws, including the California Political Reform Act, and Buyer's board policies governing conflicts of interest; Seller is in compliance with such Laws and board policies and does not know of any facts that would violate such Laws and board policies; Seller and its officers and agents have not, directly or indirectly, offered, paid, promised, or authorized the giving of money or anything of value to any employee, director, officer of Buyer or governmental official in Alameda County, California for the purpose of influencing any act or decision of such employee, director, officer or government official in her official capacity; no officer or agent of Seller (i) is a government official in Alameda County, California or a family member of a government official in Alameda County, California or (ii) has a personal, business, or other relationship or association with any government official in Alameda County, California or family member thereof who may have responsibility for or oversight of any activities of Buyer; Seller does not employ any government official in Alameda County, California or family member thereof.

14.2 <u>Buyer's Representations and Warranties</u>. As of the Execution Date, Buyer represents and warrants as follows:

(a) Buyer is a joint powers authority and a validly existing community choice aggregator, duly organized, validly existing and in good standing under the laws of the State of California and the rules, regulations and orders of the California Public Utilities Commission, and is qualified to conduct business in each jurisdiction of the Joint Powers Agreement members. All Persons making up the governing body of Buyer are the elected or appointed incumbents in their positions and hold their positions in good standing in accordance with the Joint Powers Agreement and other Law.

(b) Buyer has the power and authority to enter into and perform this Agreement and is not prohibited from entering into this Agreement or discharging and performing all covenants and obligations on its part to be performed under and pursuant to this Agreement, except where such failure does not have a material adverse effect on Buyer's performance under this Agreement. The execution, delivery and performance of this Agreement by Buyer has been duly authorized by all necessary action on the part of Buyer and does not and will not require the consent of any trustee or holder of any indebtedness or other obligation of Buyer or any other party to any other agreement with Buyer.

(c) The execution and delivery of this Agreement, consummation of the transactions contemplated herein, and fulfillment of and compliance by Buyer with the provisions of this Agreement will not conflict with or constitute a breach of or a default under any Law presently in effect having applicability to Buyer, the documents of formation of Buyer or any outstanding trust indenture, deed of trust, mortgage, loan agreement or other evidence of indebtedness or any other agreement or instrument to which Buyer is a party or by which any of its property is bound.

(d) This Agreement has been duly executed and delivered by Buyer. This Agreement is a legal, valid and binding obligation of Buyer enforceable in accordance with its terms, except as limited by laws of general applicability limiting the enforcement of creditors' rights or by the exercise of judicial discretion in accordance with general principles of equity.

(e) Buyer warrants and covenants that with respect to its contractual obligations under this Agreement, it will not claim immunity on the grounds of sovereignty or similar grounds with respect to itself or its revenues or assets from (i) suit, (ii) jurisdiction of court (provided that such court is located within a venue permitted in law and under the Agreement), (iii) relief by way of injunction, order for specific performance or recovery of property, (iv) attachment of assets, or (v) execution or enforcement of any judgment; provided, however that nothing in this Agreement shall waive the obligations or rights set forth in the California Tort Claims Act (Government Code Section 810 et seq.).

(f) Buyer is a "local public entity" as defined in Section 900.4 of the Government Code of the State of California.

14.3 <u>General Covenants</u>. Each Party covenants that commencing on the Execution Date and continuing throughout the Term it shall continue to be duly organized, validly existing and in good standing under the laws of the jurisdiction of its formation and to be qualified to conduct business in the State of California and each jurisdiction where the failure to so qualify would have a material adverse effect on its business or financial condition.

14.4 <u>**Covenants of Seller**</u>. Seller covenants to and for the benefit of Buyer that throughout the Delivery Term (unless another time period is specified):

(a) Seller will (i) acquire and maintain all Governmental Approvals necessary for the construction, operation, and maintenance of the Project consistent with Safety Requirements, including any approvals required under the California Environmental Quality Act, (ii) Notify Buyer of any material modifications or lapse in renewal of Governmental Approvals, and (iii) at Buyer's request, provide to Buyer digital copies of any Governmental Approvals.

(b) For Multifamily Projects, 90% of all workers employed on in support of the Projects by Seller or through Contractors shall be paid not less than the prevailing rate of wages for the appropriate craft, classification, type of worker and locality as determined by the Director of the State Department of Industrial Relations in accordance with Division 2, Part 7, Chapter 1 of the California Labor Code, or as set out in the wage determination of the U.S. Secretary of Labor, whichever is higher.

Seller shall make commercially reasonable effort to: Cause Alameda County residents to provide at least 50% of the work hours associated with the construction, operation, and maintenance of the Projects; hire graduates of state-certified apprenticeship training programs, such as Cypress Mandela and Rising Sun Energy Center, in support of the construction, operation, and maintenance of the Projects.

(c) For Single-family Projects, Seller will use reasonable efforts to ensure that employees hired by Seller, and its Contractors, that will perform construction work or provide services at the Site related to construction of the Project are paid wages at rates not less than those prevailing for workers performing similar work in the locality as provided by applicable California Law, if any. Nothing herein shall require Seller or its Contractors to comply with, or assume liability created by other inapplicable provisions of any California labor Laws. (d) Seller will deliver the Product to Buyer free and clear of all liens, security interests, claims and encumbrances or any interest therein or thereto by any Person.

(e) Seller will take no action or permit any Person (other than Buyer) to take any action that would impair in any way Buyer's ability to rely on the Project in order to satisfy its Compliance Obligations.

(f) Seller shall operate the Project during the Delivery Term in accordance with <u>Appendices I and II</u> and Safety Requirements.

(g) Seller shall comply with all Utility Distribution Company, Participating Transmission Owner, and CAISO Tariff requirements applicable to energy storage facilities.

(h) Seller shall follow all the rules set forth in Exhibit A of the MUA Decision.

Article 15: INDEMNITIES

15.1 Indemnity by Seller.

Seller shall defend, indemnify and hold harmless Buyer, its directors, (a) officers, agents, attorneys, employees, and representatives ("Buyer Group") from and against all third party or Customer claims arising out of this Agreement, demands, losses, liabilities, penalties, and expenses, including reasonable attorneys' and expert witness fees, however described (collectively, "Claims"), which arise out of or relate to or are in any way connected with (i) Seller's delivery of the Product to Buyer, (ii) Seller's or its Affiliates' ownership, development, construction, operation or maintenance of the Project, including the Project(s) and the Site(s); (iii) Seller's or its Affiliates' actions or inactions, including Seller's breach of this Agreement or other agreements related to the development, construction, ownership, operation or maintenance of the Project, the Project(s), Seller's Portfolio, Customer(s), or Site (including any Claims relating to a Shortfall); (iv) any environmental matters associated with the Project, including the disposal and transportation of Hazardous Substances by or on behalf of the Seller or at the Seller's direction or agreement; (v) any agreement between Seller or its Affiliates and a third party including any Customer Agreement; (vi) the participation of Customers in the Project (or the solicitation thereof); or (vii) Seller's or its Affiliates' violation of any applicable Law, Requirements, or other requirements of Transmission Provider, Utility Distribution Company, NERC, WECC or Reliability Organization; in each case including any loss, claim, action or suit, for or on account of injury to, bodily or otherwise, or death of, persons, or for damage to or destruction or economic loss of property belonging to Buyer, Seller, Seller's Affiliates, Customers, or others, excepting only such losses, to the extent solely caused by the willful misconduct or gross negligence of a member of the Buyer Group (collectively, "Indemnifiable Losses").

(b) Seller shall defend, indemnify and hold harmless the Buyer Group harmless from and against all Claims incurred by or brought against Buyer in connection with Environmental Costs.

15.2 Notice of Claim.

(a) <u>Notice of Claim</u>. Subject to the terms of this Agreement and upon obtaining knowledge of an Indemnifiable Loss for which it is entitled to indemnity under this <u>Article 15</u>, Buyer will promptly Notify Seller in writing of any Claim which Buyer has determined has given or could give rise to an Indemnifiable Loss under <u>Section 15.1</u>. The Notice is referred to as a "<u>Notice of Claim</u>." A Notice of Claim will specify, in reasonable detail, the facts known to Buyer regarding the Indemnifiable Loss.

(b) <u>Failure to Provide Notice</u>. A failure to give timely Notice or to include any specified information in any Notice as provided in this <u>Section 15.2</u> will not affect the rights or obligations of any Party hereunder except and only to the extent that, as a result of such failure, any Party which was entitled to receive such Notice was deprived of its right to recover any payment under its applicable insurance coverage or was otherwise materially damaged as a direct result of such failure and, provided further, Seller is not obligated to indemnify Buyer for the increased amount of any Indemnifiable Loss which would otherwise have been payable to the extent that the increase resulted from the failure to deliver timely a Notice of Claim.

15.3 Defense of Claims. If, within ten (10) days after giving a Notice of Claim regarding a Claim to Seller pursuant to Section 15.2(b), Buyer receives Notice from Seller that Seller has elected to assume the defense of such Claim as provided in the last sentence of Section 15.2(b), Seller will not be liable for any legal expenses subsequently incurred by Buyer in connection with the defense thereof; provided, however, that if Seller fails to take reasonable steps necessary to defend diligently such Claim within ten (10) days after receiving Notice from Buyer that Buyer believes Seller has failed to take such steps, or if Seller has not undertaken fully to indemnify Buyer in respect of all Indemnifiable Losses relating to the matter, Buyer may assume its own defense, and Seller will be liable for all costs or expenses, including attorneys' fees, paid or incurred in connection therewith. Without the prior written consent of Buyer, Seller will not enter into any settlement of any Claim which would lead to liability or create any financial or other obligation on the part of Buyer for which Buyer is not entitled to indemnification hereunder; provided, however, that Seller may accept any settlement without the consent of Buyer if such settlement provides a full release to Buyer and no requirement that Buyer acknowledge fault or culpability. If a firm offer is made to settle a Claim without leading to liability or the creation of a financial or other obligation on the part of Buyer for which Buyer is not entitled to indemnification hereunder and Seller desires to accept and agrees to such offer, Seller will give Notice to Buyer to that effect. If Buyer fails to consent to such firm offer within ten (10) calendar days after its receipt of such Notice, Buyer may continue to contest or defend such Claim and, in such event, the maximum liability of Seller to such Claim will be the amount of such settlement offer, plus reasonable costs and expenses paid or incurred by Buyer up to the date of such Notice.

15.4 <u>Subrogation of Rights</u>. Upon making any indemnity payment, Seller will, to the extent of such indemnity payment, be subrogated to all rights of the Buyer against any third party in respect of the Indemnifiable Loss to which the indemnity payment relates; provided that (a) Seller is in compliance with its obligations under this Agreement in respect of such Indemnifiable Loss, and (b) until Buyer recovers full payment of its Indemnifiable Loss, any and all claims of the Seller against any such third party on account of said indemnity payment are hereby made expressly subordinated and subjected in right of payment to Buyer's rights against such third party. Without limiting the generality or effect of any other provision hereof, Buyer and Seller shall

execute upon request all instruments reasonably necessary to evidence and perfect the abovedescribed subrogation and subordination rights.

15.5 **<u>Rights and Remedies are Cumulative</u>**. The rights and remedies of a Party pursuant to this <u>Article 15</u> are cumulative and in addition to the rights of the Parties otherwise provided in this Agreement, except with respect to any expressly exclusive remedies herein.

Article 16: INSURANCE

16.1 <u>Insurance</u>. Throughout the Term, Seller shall procure and maintain the following insurance coverage and require and cause its Contractors to maintain the same levels of coverage. For the avoidance of doubt, the obligations of the Seller in this <u>Section 16.1</u> constitute a material obligation of this Agreement.

(a) <u>Workers' Compensation and Employers' Liability</u>.

(i) If it has employees, workers' compensation insurance indicating compliance with any applicable labor codes, acts, Laws or statutes, California state or federal, where Seller performs Work.

(ii) Employers' liability insurance will not be less than one million dollars (\$1,000,000.00) for injury or death occurring as a result of each accident. With regard to bodily injury by disease, the one million dollar (\$1,000,000) policy limit will apply to each employee.

(b) <u>Commercial General Liability</u>.

(i) Commercial general liability insurance, including products and completed operations and personal injury insurance, in a minimum amount of two million dollars (\$2,000,000) per occurrence, and an annual aggregate of not less than five million dollars (\$5,000,000), endorsed to provide contractual liability in said amount, specifically covering Seller's obligations under this Agreement and including Buyer as an additional named insured.

(ii) An umbrella insurance policy in a minimum limit of liability of ten million dollars (\$10,000,000).

(iii) Defense costs shall be provided as an additional benefit and not included within the limits of liability. Such insurance shall contain standard cross-liability and severability of interest provisions.

(c) <u>Business Auto</u>.

(i) Business auto insurance for bodily injury and property damage with limits of one million dollars (\$1,000,000) per occurrence.

(ii) Such insurance shall cover liability arising out of Seller's use of all owned (if any), non-owned and hired vehicles, including trailers or semi-trailers in the performance of the Agreement."

(d) <u>Construction All-Risk Insurance</u>. During the construction of the Project prior to the Stage 2 Commercial Operation Date, construction all-risk form property insurance covering the Project and naming Seller (and Lender if any) as the loss payee.

(e) <u>Contractor's Pollution Liability</u>.

(i) If the scope of Work involves areas of known pollutants or contaminants, pollution liability coverage will be required to cover bodily injury, property damage, including clean-up costs and defense costs resulting from sudden, and accidental conditions, including the discharge, dispersal, release or escape of smoke, vapors, soot, fumes, acids, alkalis, toxic chemicals, hydrocarbons, liquids or gases, waste materials or other irritants, contaminants or pollutants into or upon land, the atmosphere or any water course or body of water shall be maintained.

(ii) The limit will be at least two million dollars (\$2,000,000.00) each occurrence for bodily injury and property damage.

(iii) The policy will endorse Buyer as additional insured.

16.2 <u>Evidence of Insurance</u>. Within ten (10) days after the Execution Date and upon annual renewal thereafter, Seller shall deliver to Buyer certificates of insurance evidencing the coverage required under this Agreement. These certificates shall specify that Buyer shall be given at least thirty (30) days prior Notice by Seller in the event of any material modification, reduction, cancellation or termination of coverage. Such insurance shall be primary coverage without right of contribution from any insurance of Buyer.

16.3 <u>Failure to Comply</u>. If Seller fails to comply with any of the provisions of this <u>Article 16</u>, Buyer may, but is not obligated, to purchase the insurance coverage required under <u>Section 16.1</u> and set off the cost thereof from any amounts owed by Buyer to Seller under this Agreement.

Article 17: RECORDS AND AUDIT RIGHTS

17.1 **Operations Logs**. Seller shall maintain a complete and accurate log of all material operations. Such log will include, but not be limited to, information on charging, discharging, availability, maintenance performed, outages, energy characteristics of the energy storage systems and similar information relating to the availability, testing and operation of the Project. Seller shall provide this information electronically to Buyer on a monthly basis. At the request of Buyer or any Governmental Authority having jurisdiction over any of the Requirements, Seller shall provide all records demonstrating that the Project is operated and maintained in accordance with Requirements.

17.2 **<u>Records and Audit</u>**.

(a) Seller shall provide access to such financial records and personnel required by Buyer in order to facilitate Buyer's compliance with applicable Law and Generally Accepted Accounting Principles. (b) To facilitate payment and verification, each Party shall maintain all books and records necessary for billing and payments, including copies of all invoices under this Agreement, for a period of at least five (5) years or as otherwise required by Law. Seller will make all records available to Buyer at its principal place of business during normal working hours.

17.3 <u>General Audit Right</u>. Buyer has the right during normal working hours, and after reasonable Notice, to examine Seller's records to the extent reasonably necessary to verify (a) Seller's compliance with this Agreement (including <u>Section 14.4</u>), (b) the accuracy of any statement including the Portfolio Safety Plan or other documents that supplement this Agreement, and (c) any charge, or computation made pursuant to this Agreement. If such examination reveals any material inaccuracy, necessary adjustments shall be made promptly.

17.4 <u>State Auditor</u>. In accordance with Government Code Section 8546.7, Seller may be subject to audit by the California State Auditor with regard to Seller's performance of this Agreement if the compensation under this Agreement exceeds ten thousand dollars (\$10,000.00).

17.5 **Data Request Cooperation**. Each Party shall use reasonable efforts to assist the other Party in gathering information for and preparing responses to data requests and other inquiries from Governmental Authorities or Public Records Act requests that are related to or associated with the Project, delivery of Product or this Agreement, subject to the requirements of Article 20.

Article 18: ASSIGNMENT

18.1 <u>General Prohibition on Assignments</u>. Except as provided below, neither Party may directly or indirectly assign this Agreement or its rights or obligations under this Agreement, without the prior written consent of the other Party, which consent shall not be unreasonably withheld. Any Seller Change of Control or direct or indirect change of control of Buyer (whether voluntary or by operation of law) will be deemed an assignment and will require the prior written consent of the other Party, which consent shall not be unreasonably withheld.

18.2 Collateral Assignment.

(a) Subject to the provisions of this <u>Section 18.2</u>, Seller has the right to assign this Agreement as collateral for any financing or refinancing of the Portfolio or the Projects. Seller has the right to assign the Projects and Customer Agreements in connection with any financing or refinancing of the Projects.

(b) In connection with any collateral assignment of this Agreement for purposes of the financing or refinancing of the Project by Seller, Buyer agrees to work with Seller and Lender to agree upon a consent to collateral assignment of this Agreement ("<u>Collateral</u> <u>Assignment Agreement</u>"). The Collateral Assignment Agreement must be in form and substance agreed to by Buyer, Seller and Lender.

18.3 <u>Assignment to Seller Affiliate</u>. Except as may be precluded by, or would cause Buyer to be in violation of the Political Reform Act (Cal. Gov. Code section 81000 et seq.) or the regulations thereto, Cal. Government Code section 1090, Buyer's Conflict of Interest Code/Policy or any other conflict of interest Law, Seller may, without the prior written consent of Buyer, transfer or assign this Agreement, including through a Change of Control, to an Affiliate of Seller, provided that such assignment shall not release Seller from its obligations under this Agreement unless such Affiliate has financial strength and technical capabilities equal to or greater than that of Seller.

18.4 Unauthorized Assignment; Costs.

(a)

Any assignment or purported assignment in violation of this Article 18 is

void.

(b) No assignment of this Agreement shall be effective unless such assignment is memorialized in a written agreement signed by the assignee and, except in connection with a collateral financing, in which agreement the assignee assumes all of the assignor's obligations and liabilities under this Agreement.

(c) Seller shall be responsible for Buyer's reasonable costs associated with the preparation, review, execution and delivery of documents in connection with any assignment of this Agreement, including without limitation reasonable attorneys' fees.

Article 19: DISPUTE RESOLUTION

19.1 <u>Governing Law</u>. This Agreement and the rights and duties of the Parties hereunder shall be governed by and construed, enforced and performed in accordance with the laws of the State of California, without regard to principles of conflicts of law. The Parties agree that any suit, action or other legal proceeding by or against any Party (or its Affiliates or designees) with respect to or arising out of this Agreement shall be brought in the federal courts of the United States Northern District of California, or the courts of the State of California sitting in the County of Alameda, California.

Dispute Resolution. In the event of any dispute arising under this Agreement, 19.2 within ten (10) days following the receipt of a written Notice from either Party identifying such dispute, the Parties shall meet, negotiate and attempt, in good faith, to resolve the dispute quickly, informally and inexpensively. If the Parties are unable to resolve a dispute arising hereunder within the earlier of either thirty (30) days of initiating such discussions, or within forty (40) days after Notice of the dispute, the parties shall submit the dispute to mediation prior to seeking any and all remedies available to it at Law in or equity. The Parties will cooperate in selecting a qualified neutral mediator selected from a panel of neutrals and in scheduling the time and place of the mediation as soon as reasonably possible, but in no event later than thirty (30) days after the request for mediation is made. The Parties agree to participate in the mediation in good faith and to share the costs of the mediation, including the mediator's fee, equally, but such shared costs shall not include each Party's own attorneys' fees and costs, which shall be borne solely by such Party. If the mediation is unsuccessful, then either Party may seek any and all remedies available to it at law or in equity, subject to the limitations set forth in this Agreement. To the fullest extent permitted under applicable Law, any statute of limitations applicable to a dispute that is mediated by the Parties pursuant to this Agreement shall toll during any period in which such dispute is being mediated in accordance with this Section 19.2.

Article 20: CONFIDENTIAL INFORMATION

20.1 <u>Confidential Information</u>. Throughout the Term, neither Party shall disclose (a) the non-public terms or conditions of this Agreement, (b) information disclosed to such Party by the other Party that is (i) marked or expressly identified as "confidential" and (ii) accompanied by a statutory reference to the applicable section of the Public Records Act pursuant to which such information may be kept confidential or (c) Customer Information (collectively, "<u>Confidential Information</u>") to a third party.

20.2 <u>Permitted Disclosures</u>. A Party may disclose Confidential Information: (a) to the Party's Affiliates and the Party's and its Affiliate's employees, counsel, accountants, advisors, lenders, or equity investors who have a need to know such information and have agreed to keep such terms confidential; (b) in order to comply with any applicable Law or any exchange, regulation, Balancing Authority, control area or Transmission Provider rule, or order issued by a court or entity with competent jurisdiction over the disclosing Party ("<u>Disclosing Party</u>"), other than to those entities set forth in subsection (c); or (c) in order to comply with any applicable regulation, rule, or order of the CPUC, CEC, or FERC.

(a) <u>Procedure for Permitted Disclosures</u>. In connection with requests made pursuant to <u>Section 20.2(b)</u> ("<u>Disclosure Order</u>") and disclosures pursuant to <u>Section 20.2(c)</u> ("<u>Regulatory Disclosure</u>") each Party shall, to the extent practicable, use reasonable efforts to: (A) Notify the other Party prior to disclosing the Confidential Information and (B) prevent or limit such disclosure. After using such reasonable efforts, the Disclosing Party shall not be: (I) prohibited from complying with a Disclosure Order or making the Regulatory Disclosure or (II) liable to the other Party for monetary or other damages incurred in connection with such disclosures of the Confidential Information.

Disclosure Requests. If a Party ("Receiving Party") receives a request (b) from a third party for access to, or inspection, disclosure or copying of, any of the other Party's (the "Supplying Party") Confidential Information ("Disclosure Request"), then the Receiving Party shall provide Notice and a copy of the Disclosure Request to the Supplying Party within three (3) Business Days of receipt of the Disclosure Request. Within three (3) Business Days of receipt of such Notice, the Supplying Party may provide a Notice to the Receiving Party stating that the Supplying Party believes there are reasonable legal grounds for denying or objecting to the Disclosure Request and will seek a protective order or other judicial determination to protect the confidential information, in such case, the Supplying Party will cover all of its own costs and further agrees to indemnify and pay any and all costs incurred by Receiving Party as a result of the Supplying Party's attempts to protect from disclosure the information. If the Supplying Party does not seek or does not receive a protective order or other judicial determination protecting the Confidential Information, the Receiving Party may disclose the Confidential Information, which disclosure will not be considered a violation of this Agreement. The Parties recognize that Buyer is a public entity subject to the provisions of the California Public Records Act, and that disclosures made thereunder shall not be considered a violation of this Agreement.

20.3 <u>**Remedies**</u>. Except as provided in <u>Section 20.2</u> with respect to the Parties' permitted disclosures, the Parties shall be entitled to all remedies available at Law or in equity to enforce, or seek relief in connection with, this confidentiality obligation.

20.4 <u>**Customer Information**</u>. Seller shall comply with all applicable Laws relating to the protection of Customer Information, including California Public Utilities Code Section 8380, *et seq.* and the "Rules Regarding Privacy and Security Protections for Energy Usage Data" adopted by the CPUC.

Article 21: GENERAL PROVISIONS

21.1 <u>Entire Agreement; Integration; Exhibits</u>. This Agreement constitutes the entire agreement and understanding between Seller and Buyer with respect to the subject matter hereof and supersedes all prior agreements relating to the subject matter hereof, which are of no further force or effect. The Cover Sheet and any exhibit, appendix, or other attachment hereto is an integral part hereof and is made a part of this Agreement by reference. This Agreement shall be considered for all purposes as prepared through the joint efforts of the Parties and shall not be construed against one Party or the other as a result of the preparation, substitution, submission or other event of negotiation, drafting or execution hereof.

21.2 <u>Amendments</u>. This Agreement may only be amended, modified or supplemented by an instrument in writing executed by duly authorized representatives of Seller and Buyer; *provided*, that, for the avoidance of doubt, this Agreement may not be amended by electronic mail communications.

21.3 <u>Survival</u>. Applicable provisions of this Agreement shall continue in effect after termination, to the extent necessary to enforce or complete the duties, obligations or responsibilities of the Parties arising prior to termination. Notwithstanding anything to the contrary in this Agreement, (a) all rights under <u>Sections 15.1</u> through <u>15.5</u> (Indemnities) and any other indemnity rights survive the end of the Term without limit, (b) all audit rights under <u>Sections 17.2</u> and <u>17.3</u> survive the end of the Term for an additional one (1) year, or as required by applicable Law, (c) all rights and obligations under <u>Article 20</u> (Confidentiality) survive the end of the Term without limit, and (d) all provisions relating to limitations of liability survive without limit.

21.4 <u>Waivers</u>. Waiver by a Party of any default by the other Party shall not be construed as a waiver of any other default. Any waiver of a default under this Agreement must appear in a writing signed by the waiving Party.

21.5 <u>No Agency, Partnership, Joint Venture or Lease</u>. Seller and the agents and employees of Seller shall, in the performance of this Agreement, act in an independent capacity and not as officers or employees or agents of Buyer. Under this Agreement, Seller and Buyer intend to act as Product seller and Product purchaser, respectively, and do not intend to be treated as, and shall not act as, partners in, co-venturers in or lessor/lessee with respect to the Project, the Product or any business related to the Project. This Agreement shall not impart any rights enforceable by any third party other than a permitted successor or assignee bound to this Agreement or, to the extent set forth herein, any Lender or indemnitee. In no event shall Buyer's receipt or review of any Seller submission, or Buyer's monitoring of Project data or cooperation in Project operations be construed as an assumption of any responsibility, liability or obligation of Seller for the design, construction or operation of the Project. 21.6 <u>Severability</u>. In the event that any provision of this Agreement is unenforceable or held to be unenforceable, the Parties agree that all other provisions of this Agreement have force and effect and shall not be affected thereby. The Parties shall, however, use their best endeavors to agree on the replacement of the void, illegal or unenforceable provision(s) with legally acceptable clauses which correspond as closely as possible to the sense and purpose of the affected provision and this Agreement as a whole.

21.7 <u>Mobile Sierra</u>. Notwithstanding any provision of this Agreement, neither Party shall seek, nor shall they support any third party seeking, to prospectively or retroactively revise the rates, terms or conditions of service of this Agreement through application or complaint to the FERC pursuant to the provisions of the Federal Power Act, absent prior written agreement of the Parties. Further, absent the prior written agreement in writing by both Parties, the standard of review for changes to the rates, terms or conditions of service of this Agreement proposed by a Party, a non-Party, or the FERC acting sua sponte shall be the "public interest" standard of review set forth in *United States Gas Pipe Line Co. v. Mobile Gas Service Corp., 350 U.S. 332 (1956)* and *Federal Power Commission v. Sierra Pacific Power Co., 350 U.S. 348 (1956)*, and clarified by *Morgan Stanley Capital Group, Inc. v. Public Util. Dist. No. 1 of Snohomish, 554 U.S. 527 (2008)* and *NRG Power Mrkt'g, LLC v. Maine Pub. Util. Comm'n, 558 U.S. 165 (2010).*

21.8 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which taken together shall constitute one and the same instrument and each of which shall be deemed an original.

21.9 <u>Electronic Delivery</u>. This Agreement may be duly executed and delivered by a Party by execution and electronic format (including portable document format (.pdf)) delivery of the signature page of a counterpart to the other Party, provided that any digital signatures are in compliance with California Code of Regulations, Title 2, Division 7, Chapter 10, Sections 22000 -22005.

21.10 <u>No Recourse to Members of Buyer</u>. Buyer is organized as a joint powers authority in accordance with the Joint Exercise of Powers Act of the State of California (Government Code Section 6500, et seq.) pursuant to its Joint Powers Agreement and is a public entity separate from its constituent members. Buyer shall solely be responsible for all debts, obligations and liabilities accruing and arising out of this Agreement. Seller shall have no rights and shall not make any claims, take any actions or assert any remedies against any of Buyer's constituent members, or the employees, directors, officers, consultants or advisors or Buyer or its constituent members, in connection with this Agreement.

21.11 **Forward Contract**. The Parties acknowledge and agree that this Agreement constitutes a "forward contract" within the meaning of the U.S. Bankruptcy Code, and Buyer and Seller are "forward contract merchants" within the meaning of the U.S. Bankruptcy Code. Each Party further agrees that, for all purposes of this Agreement, each Party waives and agrees not to assert the applicability of the provisions of 11 U.S.C. § 366 in any bankruptcy proceeding wherein such Party is a debtor. In any such proceeding, each Party further waives the right to assert that the other Party is a provider of last resort to the extent such term relates to 11 U.S.C. §366 or another provision of 11 U.S.C. § 101-1532.

21.12 **<u>Further Assurances</u>**. Each of the Parties hereto agree to provide such information, execute and deliver any instruments and documents and to take such other actions as may be necessary or reasonably requested by the other Party which are not inconsistent with the provisions of this Agreement and which do not involve the assumptions of obligations other than those provided for in this Agreement, to give full effect to this Agreement and to carry out the intent of this Agreement.

Article 22: NOTICES

22.1 <u>Addresses for the Delivery of Notices</u>. Any Notice required, permitted, or contemplated hereunder shall be in writing, shall be addressed to the Party to be notified at the address set forth on <u>Appendix VIII</u> or at such other address or addresses as a Party may designate for itself from time to time by Notice hereunder.

22.2 <u>Time of Delivery</u>. Each Notice required, permitted, or contemplated hereunder shall be deemed to have been validly served, given or delivered as follows: (a) if sent by United States mail with proper first class postage prepaid, three (3) Business Days following the date of the postmark on the envelope in which such Notice was deposited in the United States mail; (b) if sent by a regularly scheduled overnight delivery carrier with delivery fees either prepaid or an arrangement with such carrier made for the payment of such fees, the next Business Day after the same is delivered by the sending Party to such carrier; (c) if sent by electronic communication (including electronic mail or other electronic means) and if concurrently with the transmittal of such electronic communication the sending Party provides a copy of such electronic Notice by hand delivery; or (d) if delivered in person, upon receipt by the receiving Party. Notwithstanding the foregoing, Notices of outages or other scheduling or dispatch information or requests, may be sent by electronic communication and shall be considered delivered upon successful completion of such transmission.

SIGNATURES

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed as of the Execution Date.

SUNRUN INC.

EAST BAY COMMUNITY ENERGY AUTHORITY

By:	By:
Name:	Name:
Title:	Title:

APPENDIX I

DESCRIPTION OF PORTFOLIO

The following describes the Portfolio of Projects to be constructed, operated and maintained by Seller through the Term in accordance with the Agreement.

A. PORTFOLIO DESCRIPTION

Project name: East Bay Resilient Capacity Resource

Energy storage technology of Units: Li-ion

Interconnection:

CAISO transmission access charge area (e.g. PG&E): PG&E

B. PORTFOLIO SIZE

Nameplate capacity (aggregate): _ Minimum: 3 MW; Maximum: 12.5 MW

APPENDIX II

OPERATIONAL CHARACTERISTICS

The following describes the Operational Characteristics to determine the amount of Product (Load Modification and Capacity Attributes) it can provide.

Physical Location and Point of Interconnection

Shall be as set forth in <u>Appendix I-A</u>.

Discharging and Charging

Maximum continuous discharge power (Dmax): MW

Minimum continuous discharge power (Dmin): ____ MW

Maximum discharge duration at constant Dmax : ____ (hours)

Maximum continuous charge power (Cmax): MW

Minimum continuous charge power (Cmin): _____ MW

Maximum charge duration at constant Cmax: ____ (hours)

Amount of Energy released to fully discharge: MWh

Amount of Energy required to fully charge: ____ MWh

Round-trip efficiency: %

Ramp Rates

[Describe ramp rates. If ramp rates vary with loading level, please provide a ramp rate for each segment within the operational range in which it differs. If ramp rates vary based on the operating mode (e.g. regulation), please provide separately.]

Dmin to Dmax: ____ MW/second Cmin to Cmax: ____ MW/second Dmax to Dmin: ____ MW/second Cmax to Cmin: ____ MW/second

System Response Time

[Timing should commence with system at a steady-state starting value and end when system has reached a steady-state ending value. Idle means that the system is neither charging nor discharging, but is online and available for immediate operation. Time should include time from notification.]

Idle to Dmax: _____ seconds

Idle to Cmax: seconds

Dmax to Cmax: _____ seconds

Cmax to Dmax: _____ seconds

Dmin to Cmin: _____ seconds

Cmin to Dmin: _____ seconds

[For the purpose of filling out this Appendix, Discharge (Charge) Start-up Time is the amount of time needed to go from non-operation to Dmin (Cmin). (The state of non-operation includes but is not limited to being unsynchronized to the grid.) Provide in seconds if appropriate.]

Discharge Start-up time (from notification to Dmin): ______ seconds

Discharge Start-up Fuel: ____ MMBtu

Starts and other Run Time Limitations

[Describe start limitations. Include any daily or annual start limitations. Insert constraints, if any, on run hours and a brief description of the reason for such constraint(s).]

Start limitations: _____

Run hour limitations:

[Describe minimum times.]

The minimum run time after a Discharge Start-up is _____ seconds

The minimum run time after a Charge Start-up is _____ seconds

The minimum down time after a shutdown is _____ seconds

APPENDIX III PROGRESS REPORTING FORM

Each Progress Report must include the following items:

- 1. Executive Summary.
- 2. Project description.
- 3. Indicative site plans representative of Projects to be included in Portfolio. Individual Project site plans for 10% or fewer of total Projects in Portfolio to be provided to EBCE at EBCE's discretion.
- 4. Description of any material planned changes to the Project or the Site.
- 5. Schedule showing progress on Project construction generally and achieving each of the Critical Milestones, the Commercial Operation Milestones, and the Initial Delivery Dates.
- 6. Summary of activities during the previous month, including any OSHA labor hour reports.
- 7. Forecast of activities scheduled for the current calendar quarter.
- 8. Written description about the progress relative to the Critical Milestones, the Commercial Operation Milestones, and the Initial Delivery Dates, including whether Seller is on schedule with respect to the same.
- 9. List of issues that are likely to potentially affect achievement of the Critical Milestones, the Commercial Operation Milestones, and the Initial Delivery Dates.
- 10. Progress and schedule of the EPC Contract, all major equipment supply agreements, Governmental Approvals, technical studies, and financing arrangements.
- 11. Indicative pictures, in sufficient quantity and of appropriate detail, in order to document construction and interconnection progress of a representative portion of Projects to be included in Portfolio. Pictures pertaining to 10% or fewer of individual Projects included in Portfolio to be provided to EBCE at EBCE's discretion.
- 12. Compliance with workforce and prevailing wage requirements.
- 13. Any other documentation reasonably requested by Buyer.

SELLER'S PORTFOLIO LIST

as of [____]

Customer Service Account Number	Customer name	Physical address of Site	installed to meet capacity	installed in excess of	and model number(s)	type (i.e., residential, commercial, municipal)	Project descr (e.g., stand storage, so storage, family, disadvantage community,

[By submitting this Seller's Portfolio List to Buyer, Seller attests as of the date signed below, that all of the information is accurate and that the Customers and Projects comprising Seller's Portfolio List are in compliance with the terms of the Agreement.]

Signature:	
Name:	
Title:	
Date:	

APPENDIX IV

CERTIFICATION FOR COMMERCIAL OPERATION

This certification of commercial operation ("Certification") is delivered by ______ ("Seller") to East Bay Community Energy Authority ("Buyer") in accordance with the terms of that certain Energy Storage Resource Adequacy Agreement dated ______ ("Agreement") by and between Seller and Buyer. All capitalized terms used in this Certification but not otherwise defined herein shall have the respective meanings assigned to such terms in the Agreement.

Seller hereby certifies and represents to Buyer the following:

- (1) The Portfolio has been completed and commissioned and it became commercially operable on [].
- (2) The Portfolio has been constructed in accordance with Appendix I of the Agreement.
- (3) The Portfolio has been constructed in accordance with the Portfolio Safety Plan.
- (4) The Portfolio is capable of producing and delivering Product in the Contract Amounts, and a performance test was conducted to confirm this capability.
- (5) Seller has designed and built the Portfolio to have a design life for the Delivery Term in accordance with Prudent Operating Practices.
- (6) The design and construction of the Portfolio was carried out by the original equipment manufacturer or other qualified organization in accordance with the designs and requirements of the original equipment manufacturer.



APPENDIX V

FORM OF LETTER OF CREDIT

[Issuing Bank Letterhead and Address]

IRREVOCABLE STANDBY LETTER OF CREDIT NO. [XXXXXX]

Date: Bank Ref.: Amount: US\$[XXXXXXX] Expiry Date:

Beneficiary:

East Bay Community Energy Authority, a California joint powers authority 1111 Broadway, Suite 300 Oakland, CA 94607

Ladies and Gentlemen:

By the order of ______ ("Applicant"), we, [insert bank name and address] ("Issuer") hereby issue our Irrevocable Standby Letter of Credit No. [XXXXXX] (the "Letter of Credit") in favor of East Bay Community Energy Authority, a California joint powers authority ("Beneficiary"), 1111 Broadway, Suite 300, Oakland, CA 94607, for an amount not to exceed the aggregate sum of U.S. \$[XXXXXX] (United States Dollars [XXXXX] and 00/100), pursuant to that certain Energy Storage Resource Adequacy Agreement dated as of ______ and as amended (the "Agreement") between Applicant and Beneficiary. This Letter of Credit shall become effective immediately and shall expire on ______, 201_.

Funds under this Letter of Credit are available to you against your draft(s) drawn on us at sight, referencing thereon our Letter of Credit No. [XXXXXX] accompanied by your dated statement purportedly signed by your duly authorized representative, in the form attached hereto as Exhibit A.

We hereby agree with the Beneficiary that all drafts drawn under and in compliance with the terms of this Letter of Credit, that such drafts will be duly honored upon presentation to the drawee at [insert bank address]. Payment shall be made by Issuer in U.S. dollars with Issuer's own immediately available funds promptly and no later than the business day immediately following our receipt of a drawing certificate from Beneficiary in the form attached hereto as Exhibit A.

Partial draws are permitted under this Letter of Credit.

It is a condition of this Letter of Credit that it shall be deemed automatically extended without an amendment for a one year period beginning on the present expiry date hereof and upon each

anniversary for such date, unless at least one hundred twenty (120) days prior to any such expiry date we have sent to you written notice by overnight courier service that we elect not to extend this Letter of Credit, in which case it will expire on its the date specified in such notice. No presentation made under this Letter of Credit after such expiry date will be honored.

Notwithstanding any reference in this Letter of Credit to any other documents, instruments or agreements, this Letter of Credit contains the entire agreement between Beneficiary and Issuer relating to the obligations of Issuer hereunder.

This Letter of Credit is subject to the Uniform Customs and Practice for Documentary Credits (2007 Revision) International Chamber of Commerce Publication No. 600 (the "UCP"), except to the extent that the terms hereof are inconsistent with the provisions of the UCP, including but not limited to Articles 14(b) and 36 of the UCP, in which case the terms of this Letter of Credit shall govern. In the event of an act of God, riot, civil commotion, insurrection, war or any other cause beyond Issuer's control (as defined in Article 36 of the UCP) that interrupts Issuer's business and causes the place for presentation of the Letter of Credit to be closed for business on the last day for presentation, the expiry date of the Letter of Credit will be automatically extended without amendment to a date thirty (30) calendar days after the place for presentation reopens for business.

Please address all correspondence regarding this Letter of Credit to the attention of the Letter of Credit Department at [insert bank address information], referring specifically to Issuer's Letter of Credit No. [XXXXXX]. For telephone assistance, please contact Issuer's Standby Letter of Credit Department at [XXX-XXXX] and have this Letter of Credit available.

[Bank Name]

[Insert officer name] [Insert officer title]

(DRAW REQUEST SHOULD BE ON BENEFICIARY'S LETTERHEAD)

Drawing Certificate

[Insert Bank Name and Address]

Ladies and Gentlemen:

The undersigned, a duly authorized representative of East Bay Community Energy Authority, a California joint powers authority, 1111 Broadway, Suite 300, Oakland, CA 94607, as beneficiary (the "Beneficiary") of the Irrevocable Letter of Credit No. [XXXXXX] (the "Letter of Credit") issued by [insert bank name] (the "Bank") by order of ______ (the "Applicant"), hereby certifies to the Bank as follows:

1. Applicant and Beneficiary are party to that certain Energy Storage Resource Adequacy Agreement dated as of ______, 20__ (the "Agreement").

2. Beneficiary is making a drawing under this Letter of Credit in the amount of U.S. \$________ because an event for which Beneficiary is entitled to make a drawing under this Letter of Credit pursuant to the Agreement has occurred.

3. The undersigned is a duly authorized representative of East Bay Community Energy Authority, a California joint powers authority and is authorized to execute and deliver this Drawing Certificate on behalf of Beneficiary.

You are hereby directed to make payment of the requested amount to East Bay Community Energy Authority, a California joint powers authority by wire transfer in immediately available funds to the following account:

[Specify account information]

East Bay Community Energy Authority

Name and Title of Authorized Representative

Date_____

APPENDIX VI

PORTFOLIO SAFETY PLAN AND DOCUMENTATION

Portfolio Safety Plan Elements:

Part One: Safety Requirements and Safety Programs

Identify the applicable safety-related codes, standards, and regulations (CSR) which govern the design, construction, operation, maintenance of the Project using the proposed technology.

Describe the Seller's and the Seller's Contractor(s)' safety programs and policies. Describe Seller's compliance with any safety-related industry standards or any industry certifications (American National Standards Institute (ANSI), International Organization for Standardization (ISO), etc.), if applicable.

Part Two: Portfolio Design and Description

Describe Seller's safety engineering approach to select equipment and design systems and the Project to reduce risks and mitigate the impacts of safety-related incidents, including cascading failures, excessive temperatures, thermal runaways, fires, explosions, disk fractures, hazardous chemical releases.

Describe the results of any failure mode effects analyses (FMEA) or similar safety engineering evaluations. In the case of lithium ion batteries, describe the safety-related reasons, including design features and historical safety records, for selecting particular anode and cathode materials and a particular manufacturer.

Provide a list of major facility components, systems, materials, and associated equipment, which includes but is not limited to, the following information:

- a) Equipment manufacturer's datasheet, model numbers, etc.,
- b) Technical specifications,
- c) Equipment safety-related certifications (e.g. UL),
- d) Safety-related systems, and
- e) Approximate volumes and types of hazardous materials expected to be on Site.

Part Three: Portfolio Safety Management

Identify and describe any hazards and risks to life, safety, public health, property, or the environment due to or arising from the Project. Describe the Seller's applicable site-specific safety plans, risk mitigation, Safeguards and layers of protection, including but not necessarily limited to:

- a) Engineering controls,
- b) Work practices,
- c) Administrative controls,
- d) Personal protective equipment and procedures,
- e) Incident response and recovery plans,
- f) Contractor pre-qualification and management,

VI - 1

- Operating procedures, Emergency plans, g)
- h)
- i)
- Training and qualification programs, Disposal, recycle, transportation and reuse procedures, and j)
- Physical security measures. k)

APPENDIX VII

MUA DECISION ATTESTATION

This attestation is delivered by _____("Seller") to East Bay Community Energy Authority ("Buyer") in accordance with the terms of that certain Energy Storage Resource Adequacy Agreement dated ______("Agreement") by and between Seller and Buyer. All capitalized terms used in this attestation but not otherwise defined herein shall have the respective meanings assigned to such terms in the Agreement.

Seller hereby certifies and represents to Buyer, as of the date set forth below, that with regard to the Project, Seller is following all the rules set forth in Exhibit A of CPUC Decision 18-01-003 on Multiple-Use Application Issues, issued January 17, 2018, regarding the multiple-use applications of energy storage facilities.

EXECUTED by SELLER this _____ day of _____, 20__.

Signature:	
N	
Name:	
Title:	

APPENDIX VIII

NOTICES

SELLER	BUYER ⁴
All Notices:	All Notices:
Street:	Street: 1111 Broadway, Suite 300
City:	City: Oakland, CA 94607
Attn:	Attn: Howard Chang, Chief Operating Officer Phone: (510) 809-7458
Phone:	Email: hchang@ebce.org
Facsimile:	
Email:	
	With a copy to:
	[]
Invoices:	Invoices:
Attn:	Attn: Marie Fontenot, Senior Director of
Phone:	Power Resources
E-mail:	Phone: (510) 361-6247
	E-mail:mfontenot@ebce.org
Scheduling:	Scheduling:
Attn:	Attn: NCPA c/o Ken Goeke, Manager, Portfolio
Phone:	and Administration
Email:	Phone: (916) 781-4290
	Email: ken.goeke@ncpa.com
Payments:	Payments:
Attn:	Attn: Howard Chang, Chief Operating Officer
Phone:	Phone: (510) 809-7458
E-mail:	E-mail: hchang@ebce.org
Wire Transfer:	Wire Transfer:
BNK:	BNK: River City Bank
ABA:	ABA: 121133416
ACCT:	ACCT: *****3199
Emergency Contact:	Emergency Contact:
Attn:	Attn: Howard Chang, Chief Operating Officer
Phone:	Phone: (510) 809-7458
E-mail:	E-mail: hchang@ebce.org

 $[\]frac{4}{2}$ Note to EBCE: Please update as needed.

APPENDIX IX

PORTFOLIO MODIFICATION CERTIFICATION

This certification of commercial operation for a Portfolio Modification ("Portfolio Modification Certification") is delivered by ______ ("Seller") to East Bay Community Energy Authority ("Buyer") in accordance with the terms of that certain Energy Storage Resource Adequacy Agreement dated ______ ("Agreement") by and between Seller and Buyer. All capitalized terms used in this Portfolio Modification Certification but not otherwise defined herein shall have the respective meanings assigned to such terms in the Agreement.

Seller each hereby certifies and represents to Buyer the following, severally and not jointly, per terms specified in Appendix X:

- (1) All parts of the Portfolio affected by the Portfolio Modification became commercially operational on [].
- (2) All parts of the Portfolio affected by the Portfolio Modification have been constructed in accordance with Appendix I of the Agreement.
- (3) The Portfolio is capable of producing and delivering Product in the Contract Amounts, and a performance test was conducted to confirm this capability.
- (4) Seller has designed and built the parts of the Portfolio affected by the Portfolio Modification to have a design life for the Delivery Term in accordance with Prudent Operating Practices.
- (5) The design and construction of the parts of the Portfolio affected by the Portfolio Modification was carried out by the original equipment manufacturer or other qualified organization.
- (6) The Portfolio as modified under the Portfolio Modification is able to operate in a manner consistent with the Safety Requirements.

SELLER:	:	
Signature:		
Name:		
Title:		
Date:		
	_	

APPENDIX X

FORM OF QUARTERLY REPORT

APPENDIX XI

[RESERVED]

APPENDIX XII

RA-SPECIFIC CONDITIONS AND TERMS

PART ONE: CAPACITY ATTRIBUTE CONDITIONS PRECEDENT

1. Seller shall have provided to Buyer a certification of Seller substantially in the form attached hereto as Appendix IV, demonstrating that the Project(s) comprising the Project as set forth in Seller's Portfolio List, can deliver, in aggregate, the Capacity Attributes of Product in the applicable Contract Amounts.

2. Seller shall have secured all CAISO and Governmental Approvals as are necessary for the safe and lawful operation and maintenance of the Project and to enable Seller to deliver the Capacity Attributes of the Product to Buyer at the Contract Amounts (and including as a Proxy Demand Resource or as a Reliability Demand Response Resource).

3. Seller shall have provided to Buyer all documentation reasonably acceptable to Buyer demonstrating that the Project successfully completed all applicable testing and registration procedures required by CAISO to Bid into the CAISO Markets, including as a Proxy Demand Resource or Reliability Demand Response Resource, in a manner sufficient to enable delivery of the Contract Amounts of Product to Buyer.

4. Seller shall have obtained certification of Product in accordance with the CAISO Tariff and CPUC requirements applicable to Product, including as a Proxy Demand Resource or as a Reliability Demand Response Resource, (i) resulting in certifications of not less than the Contract Amounts and (ii) so as to ensure the applicable Stage of the Project is fully deliverable such that Seller is able to deliver Product in the Contract Amounts to Buyer for purposes of counting towards Buyer's Compliance Obligations.

5. Seller shall have provided documentation demonstrating Seller's calculations related to and attendant fulfillment of the CAISO NQC criteria, or any revised Resource Adequacy criteria, for the applicable Stage of the Project.

6. Seller shall have submitted, or shall have caused its SC to have submitted, a Notice to Buyer including Seller's proposed Supply Plan for the first Showing Month, and Seller shall have properly submitted, or shall have caused its SC to have properly submitted, a Supply Plan to CAISO.

PART TWO: RA-SPECIFIC PAYMENT TERMS

1. No later than fifteen (15) Business Days prior to the applicable Compliance Showing deadlines for each Showing Month, Seller shall submit, or cause its SC to submit, a Notice to Buyer which includes Seller's proposed Supply Plan for such Showing Month in a format and to a platform as Notified by Buyer to Seller prior to such deadline. No later than ten (10) Business Days before the applicable Compliance Showing deadlines for each Showing Month, Buyer may Notify Seller of any administrative or typographical corrections to the Supply Plan and Seller shall implement any such corrections in the Supply Plan that it submits, or causes to be submitted, to

the CAISO. In the event that Buyer does not Notify Seller of any such corrections to the proposed Supply Plan, Seller shall submit the Supply Plan to CAISO as it was proposed by Notice to Buyer.

2. After following the foregoing procedure, Seller shall submit, or cause to be submitted, a Supply Plan to CAISO, in accordance with the applicable Compliance Showing deadlines for each Showing Month, to identify and confirm the Product to be delivered to Buyer (or, with regard to Resold Product, Buyer's designee) for each day within the applicable Showing Month. For each of the Capacity Attributes of Product that Seller submits in its Supply Plan in the applicable Showing Month, Seller shall not submit an amount greater than the Contract Amount of each of the respective Capacity Attributes. The lowest daily quantity that Seller submits for each of the Capacity Attributes in a Supply Plan shall be deemed to be the amount of the respective Capacity Attributes that Seller for such Showing Month ("Delivered Quantities").

3. For all Capacity Attributes of the Product that Seller delivers during the Delivery Term in accordance with this Agreement, Buyer shall, in accordance with Article 9, pay Seller a monthly payment ("Monthly Payment" or "MP") as follows:

 $MP = DQ \times CP$

where,

DQ = The sum of the Delivered Quantities of all RA Attributes, Local RA Attributes, and Flexible Attributes; and

CP = The contract price set forth on the Cover Sheet ("Contract Price").

PART THREE: RA-SPECIFIC OPERATIONAL COVENANTS

1. Seller shall have a DRP Agreement in place with the CAISO and is required to satisfy registration requirements and to provide information to EBCE and the CAISO to allow the CAISO to establish performance evaluation methodologies in accordance with the Tariff and the applicable Business Practice Manuals.

2. Seller shall be the SC or shall designate a qualified third party to fulfill such role for the Project in order to deliver Product to Buyer during the Delivery Term in accordance with the terms of this Agreement. Seller shall be solely responsible for all costs associated with the SC. Seller shall take, or cause its SC to take, all necessary steps to qualify itself and the Project in such other manner identified and approved by the CAISO and CPUC that permits Seller to provide Product to Buyer .

3. Seller shall comply, and shall cause SC each Customer, and each Project owner and operator to comply, with all applicable CAISO Tariff provisions, CPUC Decisions and all other applicable rules, requirements or Laws, including any Bidding of the Project to meet any Must Offer Obligations, in order to deliver the Product to Buyer and allow Buyer to use the Product to satisfy Buyer's Compliance Obligations.

4. Seller shall not accept, and shall cause the Project's SC to not accept, any proposed CPM or RMR designation by the CAISO unless and until Buyer has agreed to accept such designation in a Notice. In addition, Seller shall promptly Notify, or cause the Project's SC to promptly Notify,

Buyer within one (1) Business Day of the time Seller or the SC receives a proposal from CAISO to designate any portion of the Product as CPM Capacity or RMR Generation.

5. Seller shall provide buyer access to the Project's CAISO Portal in order to view market awards or send Buyer a copy of the Project's market awards no later than 16:00 each day the Project is bid in to the CAISO Day-Ahead or Hour-Ahead market.

APPENDIX XIII

FORM OF LOAD MODIFYING RESOURCE DISPATCH PLAN

END OF AGREEMENT
Attachment Item 15C



East Bay Community Energy, Peninsula Clean Energy, Silicon Valley Clean Energy & Silicon Valley Power

Distributed Resource Adequacy Capacity

Request for Proposal (RFP)

Issuance Date: November 5, 2019

Due Date: December 23, 2019 (5 p.m. PST)

Purpose

As Bay Area residents and businesses are suffering from unprecedented power outages due to Pacific Gas and Electric Public Safety Power Shutoffs ("PSPS"), we must invest in increased resilience to keep our communities safe and livable. The following Bay Area Load Serving Entities, including; East Bay Community Energy ("EBCE"), Peninsula Clean Energy ("PCE"), Silicon Valley Clean Energy ("SVCE") and Silicon Valley Power ("SVP") (collectively "the LSEs") are seeking proposals from qualified developers or vendors ("Proposers or Respondents") for a comprehensive offering to provide Resource Adequacy ("RA") capacity and resilience to the LSEs' residential and commercial customers through the development of customer-sited Distributed Energy Resources ("DERs") (described on page 16). Through this solicitation, EBCE, SVCE and PCE are seeking to procure a minimum of 10 megawatts ("MW") each of RA capacity and SVP will be procuring 2.7 MW of RA capacity, proportional to its geography and customer account size relative to the other LSEs, for a total of approximately 32.7 MW. Every LSE will partner with the relevant selected Proposer(s) to develop a program offering to LSEs' customers to facilitate the development of projects to provide the desired RA capacity. For each LSE, a minimum capacity will be sited on residential sites with the remaining to be installed on commercial or residential sites (see Mandatory Proposal Requirements: Target Customer Categories for definition of "residential" and "commercial" and minimum requirements). A portion of the capacity shall be be deployed by September 2020 with the remaining capacity targeted for deployment by either June 2021 or September 2021. The LSEs have an extensive need for economically-viable RA, which they seek to address, at least partially, through DER aggregation mechanisms, such as Proxy Demand Response (PDR).

This solicitation is being issued by the LSEs in a joint manner, due to the alignment of purpose in responding to the PSPS and increasing community resilience. However, Proposers will submit a separate proposal to each LSE that they are interested in proposing to, but are not required to propose to each LSE, as detailed in the Submittal Requirements section. Each LSE will evaluate proposals separately, with substantial communication occurring between LSEs to discuss proposed strategies and pricing. Each LSE will select and contract with winning Proposer(s) separately.

The bulk of this solicitation is the same for each LSE - differences between LSEs have been organized into tables throughout the solicitation to assist prospective Proposers in structuring their response to each LSE.

Background & Objectives

The LSEs have a shared commitment to serving their communities' needs and delivering lowcost, clean electricity. LSE-specific background information is provided in the table below:

	Date Formed	Service Start Date	Description and Context
--	----------------	--------------------------	-------------------------

LSE-Specific Background Information

EBCE	2016	2018	
	2010	2010	EBCE is a Joint Powers Authority formed on December 1, 2016, and, on August 10, 2017, EBCE filed its Community Choice Aggregator ("CCA") Implementation Plan with the California Public Utilities Commission ("CPUC"). On September 7, 2017, the CPUC provided Pacific Gas and Electric ("PG&E") notice of the receipt of EBCE's Implementation Plan, and on November 8, 2017, the CPUC certified EBCE's Implementation Plan. The County provided initial start-up funds to EBCE for the launch period. EBCE has additionally closed a \$50,000,000 unsecured credit facility with Barclays Bank in March of 2018. The maximum debt outstanding reached approximately \$28 million in 2018. In February 2019, EBCE repaid the County in full for the start-up loan costs of approximately \$4.8 million, and in August 2019 EBCE fully repaid the funds drawn on the Barclays credit facility, eliminating all outstanding debt. As of August 1, 2019, EBCE has approximately \$100 million in unrestricted cash and available credit lines, which translates to over 90 Adjusted Days of Liquidity on Hand. On June 1, 2018, EBCE launched Phase I of its service implementation with commercial, industrial, and municipal accounts. Phase II for residential accounts was launched November 1, 2018. Phase III was launched in April of 2019 and will enroll all Net Energy Meter ("NEM") customers. Phase III is a rolling enrollment through December 2019 based on NEM billing cycles. EBCE's ~30,000 NEM customers are a small fraction of total accounts and load. EBCE serves over 540,000 customer accounts representing a population of over 1.3 million people, 50,000 businesses, and approximately 6,000 gigawatt hours ("GWh") of annual load. EBCE is currently the second largest CCA in California and the largest in PG&E territory.
PCE	2016	2016	 Peninsula Clean Energy (PCE), a community choice energy (CCE) aggregator, is San Mateo County's official electricity provider. Formed in February 2016, PCE is a joint power authority, consisting of the County of San Mateo and all twenty of its towns and cities. PCE provides cleaner and greener electricity, and at lower rates, than the incumbent investor-owned utility (IOU), Pacific Gas & Electric Company (PG&E). PCE was the fifth CCA to launch and started serving customers in October 2016 and as of May 1, 2017, PCE service was rolled out to all eligible customers in San Mateo County. PCE serves

			 approximately 300,000 customer accounts representing a population of over 700,000 people with 3,600 GWh of clean electricity annually. PCE plans for and secures commitments from a diverse portfolio of energy-generating resources to reliably serve the electric energy requirements of its customers over the near-, mid-, and long-term planning horizons. PCE's programs include transportation electrification and advancing the expansion of low-carbon electricity. In May 2019, Peninsula Clean Energy received an investment grade credit rating from Moody's. As of June 30, 2019, PCE had an unaudited cash balance of \$127.2 million, of which \$114.1 million was unrestricted. The unrestricted cash balance represented 198 days of cash on hand, well in excess of PCE's Board policy requirement of 120 days. PCE expects to publish its FY2018-2019 audited financials at the end of November. Additional information and PCE's financial statements are available on its website: https://www.peninsulacleanenergy.com/budget-and-finances/. For more information on PCE, please go to www.peninsulacleanenergy.com.
SVCE	2016	2017	Silicon Valley Clean Energy (SVCE), a Community Choice Energy agency, is redefining the local electricity market and providing our residents and businesses with new clean energy choices—renewable and carbon—free electricity at competitive rates. SVCE was formed as a Joint Powers Authority in 2016, and now serves approximately 270,000 residential and commercial electricity customers across its service area. 97% of electricity customers in SVCE's service area receive their electricity from SVCE. For more information on SVCE, please visit: <u>https://www.svcleanenergy.org/</u> . SVCE is also in the process of launching a broad set of programs supporting decarbonization efforts, including a virtual power plant initiative to further monetize distributed energy assets. For more information on SVCE's overarching program strategy and planned programs, please visit: <u>https://www.svcleanenergy.org/programs/</u> .

SVP	1896	1896	Silicon Valley Power (SVP) is the trademark adopted for use by the not-for-profit electric municipal utility of Santa Clara, CA, serving residents and businesses for over 120 years.
			SVP serves the City of Santa Clara with a service area of approximately 19 square miles. SVP is dedicated to its community, customers, and employees. SVP provides safe, reliable, affordable, and sustainable energy solutions while deploying and scheduling resources that optimize the dispatch of SVP's generation fleet and complying with statutory and regulatory requirements. SVP meets its electricity needs in a safe, reliable, cost-effective and environmentally responsible manner. SVP is the only full service, vertically integrated publicly owned utility in Silicon Valley owning generation, transmission and distribution assets. Please see more at: http://www.siliconvalleypower.com.

The objective of this RFP is twofold. The LSEs seek to identify new and existing DER systems capable of providing capacity to satisfy RA requirements while also supporting the development of DERs to increase resilience in each service territory. The increasing threat of PSPS events has intensified the urgency and need for resilience region-wide. DERs deployed through this program will be capable of providing resiliency benefits to the LSEs' customers and the broader community in advance of the 2020 and 2021 fire seasons.

In July 2019, EBCE approved a 10-year contract for the provision of 500 kilowatts ("kW") of RA capacity through aggregated behind-the-meter battery energy storage assets. This RFP builds on the intent of that project and aims to increase the amount of distributed RA capacity procured by the LSEs, grow the market for distributed RA capacity and create a program to empower LSE customers to invest in their own energy needs while supporting the needs of the electricity grid and enhancing community resilience.

The LSEs will provide resources throughout the bid development process (See "Data Provision Process on page 5) and program implementation phase (See "*Marketing Assets to be Made Available to Awardee(s) by LSE*" on page 19). The LSEs seek qualified Proposers capable of working with each LSE to develop customer-facing programs to facilitate efficient development of the required RA capacity and potentially scale the program in future years.

RFP Schedule

The table below summarizes the RFP schedule, with details of each event provided further below, as necessary. **Proposers should note that this is a preliminary schedule and the timeline is subject to change at the discretion of the LSEs.**

Proposed Calendar of Events

Event	Date

Issuance of RFP	November 5, 2019
Representative Data Available	November 7, 2019
Informational Webinar	November 12, 2019, 1:30-3:00pm
Deadline for Questions	5pm PST, November 22, 2019
Deadline for Data Requests	5pm PST, December 4th, 2019
Addendum Issuance	December 4, 2019
Final Proposals Due	5pm PST, December 23, 2019
Proposal Review	December 23, 2019 - January 17, 2020
Interviews	January 20-24, 2020
Selection of Preferred Awardee(s)	February, 2020
Contract Negotiation Period	February, 2020
Notice of Intent to Award	March 2020
Board Meeting for Approval	April 2020

Informational Webinar & Q&A Process

LSEs will hold a **mandatory** informational webinar about this solicitation on November 12, 2019 at 1:30-3:00pm. This webinar will provide an opportunity for Proposers to learn more about the solicitation and ask clarifying questions. Register for the Webinar <u>here</u>.

Questions that arise after the date of the webinar shall be sent to <u>iross@ebce.org</u>. Questions sent via email must be received by November 22, 2019 at 5pm PST. **Questions for a specific LSE should be so labeled.** LSEs reserve the right to combine similar questions, rephrase questions, or decline to answer questions, at their sole discretion.

Data Provision Process

In order to facilitate the inclusion of accurate pricing of RA capacity in proposals, the LSEs will provide interested parties with anonymized 15-minute interval load data for a sample of commercial customers in each LSE's service territory. This data will be available on the "Representative Data Available" date (see Proposed Calendar of Events).

By providing Proposers with a sample of actual customer load profiles of representative commercial customers, this data will enable Proposers to create estimated valuations of given DER systems to those customers. In turn, those estimated valuations can be used to develop RA pricing based on actual projects, thereby de-risking proposals and mitigating the chance of inflated or deflated RA pricing. **Interested Proposers must contact JP Ross (**jross@ebce.org)

with an expression of interest by December 4, 2019 (see Deadline for Data Requests in Proposed Calendar of Events) in order to receive access to the secure data room prior to proposal submissions.

Following Proposer selection ("Awardee"), an extended set of data shall be provided with the goal of refining proposed capacity pricing during contract negotiations. Upon contract execution, the contracted Awardee(s) shall indicate to its client LSE(s) specific load profiles to be pursued for project development. Each LSE and its Awardee(s) will then initiate outreach to the chosen residential and commercial customers. Proposers shall consider this process in creation of their Customer Engagement & Go-to-Market Strategy. In order for Awardees to be provided confidential customer information by the LSEs, Awardees shall execute separate confidentiality agreements with the individual parties.

Data Request Process

Data will be made available through a Secure FTP (SFTP) to potential bidders upon request. **LSEs reserve the right to request documentation to substantiate intent and capability to submit a proposal.** Access to the data will be provided to bidders solely upon request to jross@ebce.org. Data requests can be submitted before November 7 and access to the site will be provided by 5PM on November 7. Data requests made after November 7 will be given access to the site within 1 business day of the request being submitted and LSEs satisfaction of requestor's intent to bid.

Submittal Requirements

It is the Proposer's responsibility alone to ensure that its proposals are received by each LSE's Authorized Representative, as listed in the table below, at the time and place identified in this RFP (see Proposed Calendar of Events). Proposers shall submit, in a non-alterable format such as a PDF, a proposal with the title "[LSE Name] - Proposal for Distributed RA Capacity - [Respondent Firm Name]." Proposers may choose to submit separate proposals to any subset of the LSEs, but only an LSE-specific proposal received by that LSE's Authorized Representative will be considered by that LSE. Do not submit a single joint proposal for all LSEs, and do not submit an LSE-specific proposal to a different LSE's Authorized Representative.

LSE Name	Name	Role	Email
EBCE	JP Ross	Senior Director, Local Development, Electrification and Innovation	jross@ebce.org
PCE	Peter Levitt	Associate Manager, Distributed Energy Resources Strategy	plevitt@peninsulacleanenergy.com with CC to procurement@peninsulacleanenerg y.com

Authorized Representative for Each LSE

SVCE	Aimee Bailey	Director, Decarbonization and Grid Innovation	aimee.bailey@svcleanenergy.org
SVP	Erica van Dyck	Electric Program Manager, Innovation and Distributed Generation/DER	evandyck@svpower.com

The proposal must be submitted electronically to the Authorized Representative by 5:00 p.m. PST on December 23, 2019. Proposals will be received only at the email addresses shown above and any proposal received after the due date/time or at a place other than the stated email address may not be considered and may be returned to the respondent unopened. Any attempt to contact LSE staff to discuss this RFP outside of the Q&A process described in this RFP will result in disqualification.

All proposals shall be prepared in a clear and concise manner. Unnecessarily elaborate proposals are neither expected nor desired. The emphasis of the proposal should be on responding to the parameters set forth in this RFP. Generally, each proposal shall be submitted with a cover letter and three (3) sections, including: (A) Statement of Developer Information & Qualifications, (B) Proposed Capacity & Program Approach and (C) Pricing Proposal. While Developers shall follow the general proposal format and proposal requirements specified below, the LSEs encourage proposals that satisfy the mandatory proposal parameters while presenting thoughtful, creative solutions for the provision of RA capacity and community resilience.

Proposals must include the following sections:

1. Cover Letter

An authorized representative of the Proposer shall sign the proposal on a cover letter which: (1) identifies this project by name; and (2) identifies the full legal name of the Proposer, along with name of contact person, address, phone number, fax number, email address and contractor's license number; and (3) indicates Proposer's willingness to comply with the procedures identified in this RFP, including an acknowledgement of inclusion of the three (3) sections identified above and described below.

2. Statement of Developer Information & Qualifications

Proposers shall include the following elements outlining their firm information and qualifications.

- A brief description of the Proposing firm, including legal form and ownership, and an identification of the size, stability, and capacity of the Proposing firm, including, at a minimum, an identification of the following:
 - a. Total number of years in operation
 - b. Total current number of employees
 - c. Number of offices & locations in California, or relevant locations if the Proposer does not have any California offices
 - d. Number of employees and associated office locations where proposed

services are intended to be rendered

- An identification of the Proposer's experience developing, building, financing, and operating projects of a similar size, scope, and complexity as required by this RFP, including all of the following:
 - a. Number of years Proposer has been developing relevant projects
 - b. Most recent projects for which the Proposer has performed similar services
 - The list of recent projects shall include the name, contact person, address, and phone number of each party for whom the service was provided, as well as a description of the service performed, the dollar amount of the contract, and the date of performance.
 - The LSEs appreciate that the provision of aggregated DER for RA capacity is a relatively new market. Thus, relevant project examples demonstrating experience in efforts related to, or component pieces of, the services required in this solicitation will be accepted and considered.
 - c. Description of the Proposers experience with or participation in the proposed mechanism for providing the proposed RA capacity. This should focus on experience with Proxy Demand Response ("PDR") or, if Proposer is including an alternative mechanism, experience with the relevant, equivalent, alternative RA provision mechanism.
 - d. Total capacity of solar photovoltaic and/or battery energy storage systems installed to date in kW and kW/kWh, respectively.
 - e. Total installed capacity of energy projects capable of islanding from the grid
- List of the Proposer's principals, employees and agents, including brief bios and extended resumes, which the Proposer intends to assign to this project. Proposers shall include their anticipated roles on the project and their experience with the proposed mechanism for providing the proposed RA capacity and experience developing DER projects in California.
- For anticipated subcontractors (if any), Proposers shall provide a description of the subcontractors' role on the project as well as a description of the subcontractor firm, experience developing, building, financing, and operating projects of a similar size, scope, and complexity as required by this RFP and bios and resumes of the key subcontractor team members.
- Insurance information
 - a. EBCE & SVCE: [1] Insurance information meeting the following minimum requirements: Comprehensive General Liability (\$1,000,000 per occurrence, \$2,000,000 aggregate for bodily injury and property damage), Motor Vehicle Liability Insurance (\$1,000,000), and Professional Liability Insurance (\$1,000,000).
 - b. PCE: [1] Insurance information meeting the following minimum requirements: Comprehensive General Liability (\$1,000,000), Motor Vehicle Liability Insurance (\$1,000,000), and Professional Liability

Insurance (\$1,000,000).

c. SVP: [1] Commercial General Liability no less than the following: \$1,000,000 each occurrence; \$1,000,000 general aggregate; \$1,000,000 products/completed operations aggregate; \$1,000,000 personal injury. [2] Business automobile liability insurance policy which provides coverage at least as broad as ISO form CA 00 01, with minimum policy limits of not less than one million dollars (\$1,000,000) each accident using, or providing coverage at least as broad as, Insurance Services Office form CA 00 01. [3] Workers' Compensation Insurance Policy as required by statute and employer's liability with the following limits: at least one million dollars (\$1,000,000) policy limit Illness/Injury by disease, and one million dollars (\$1,000,000) for each Accident/Bodily Injury.

3. Proposed Capacity & Program Approach

Proposers must provide a comprehensive description of the distributed RA capacity and accompanying program being offered. This section shall address and/or reflect all mandatory solicitation parameters described in the "Mandatory Proposal Parameters" in this document. As discussed under Submittal Requirements, while Proposers must submit proposals reflective of the mandatory proposal parameters, LSEs are interested in reviewing creative approaches to satisfying these parameters. And while the program approach outlined in each proposal should be specific to the individual LSE territory, Proposers are encouraged to indicate how the program approach could incorporate a regional approach (and impact cost) if the Proposer were selected by multiple LSEs.

4. Pricing

Proposing Firms are required to provide a complete offer pricing package in the form of \$[]/kw-month for each year the proposed capacity shall be providing RA. **Pricing must be presented in Attachment A: Pricing & Capacity Worksheet.** Proposers are encouraged to provide uniform pricing across the entire offered capacity but may provide tiered pricing aligned with achieving the target capacity distribution across customer types discussed in *Mandatory Proposal Parameters: Target Customer Categories.* Instructions for completing Attachment A are included therein.

Evaluation of Proposals

The LSEs reserve the right to reject any and all proposals or waive any irregularities in any proposal or the proposal process. The LSEs reserve the right to negotiate the terms of each LSE's Service Agreement with one or more Awardees. In reviewing the proposals, LSEs will consider the following:

Evaluation Categories by LSE

Evaluation Category	EBCE	PCE	SVCE	SVP	
Completeness of RFP Response	Pass/Fail				
Capacity Requirements & Deployment Timeline		F	Pass/Fail		
Technology Safety Certification & Standards		F	Pass/Fail		
Installed Storage projects have islanding capability and provide backup power		F	Pass/Fail		
Pricing	40%	40%	Weighting TBD	Considered	
Experience Developing Similar Projects & Providing RA Capacity	15%	25%	Weighting TBD	Considered	
Experience and Qualifications of Personnel	10%		Weighting TBD	Considered	
Quality and Detail of Go-to-Market & Customer Engagement Plan	15%	25%	Weighting TBD	Considered	
Proposed Fire Safety Measures	10%	Pass/Fail	Weighting TBD	Considered	
Local Hiring/Workforce Development	10%	Pass/Fail	Considered but not mandatory	Considered but not mandatory	
Innovation	N/A	10%	Weighting TBD	Considered	

Evaluation of proposals for residential and commercial RA capacity will be performed separately and compared to equivalent proposals. If a Proposer wishes to propose both residential and commercial capacity offerings, they may be included in a single proposal, but should be priced and described distinctly to enable comparison to other proposals focused on the same customer category.

Each LSE expects to invite a shortlist of its high scoring Proposer(s) for in-person interviews. Interviews will be scheduled shortly after each LSE has completed its preliminary proposal evaluations and pursuant to the timeline listed in this RFP.

Proposer(s) Award

Upon selection of preferred Awardee(s), each LSE will enter into separate contract negotiation.

Initial selection for negotiation does not imply guarantee of award. If an LSE is unable to negotiate mutually acceptable contract terms with selected Awardee(s), alternate Proposers from the shortlist may be substituted (see below).

Each LSE may award multiple Awardees for each sector, residential and commercial. Contracted Awardees will be required to provide performance bonding for delivery of contracted capacity amounts.

Proposers with qualified and/or high scoring proposals NOT selected may be placed on a shortlist by the relevant LSE and contacted to provide an updated proposal in the event that the selected Awardee(s) are not able to fulfill their proposed RA capacity. Any RA capacity contracted via this mechanism will be subject to re-verification of proposal eligibility and a request for an updated and revised pricing proposal.

Mandatory Proposal Parameters

The following parameters shall be considered by Proposers and used to structure proposals that align with the LSEs desired goals and objectives. The characteristics of each mandatory parameter are described below, followed by a table delineating the differences in requirements across each LSE. The *Summary of Differences in Mandatory Proposal Parameters by LSE* (see page 24) includes a master table that summarizes the differences in all parameter requirements as they vary across each LSE.

1. Proposed Capacity: Amount, RA Type & RA Mechanism

Through this solicitation, EBCE, SVCE and PCE expect to procure a minimum of 10 MW of RA capacity to satisfy local RA requirements through DERs located at customer accounts. Due to its smaller geography, customer size and RA obligation, SVP will procure a minimum of 2.7 MW of RA capacity. To the extent viable, LSEs are also interested in procuring capacity that satisfies flexible RA and/or system RA requirements. Proposers should consider strategies to satisfy flexible RA requirements with DERs and are encouraged to propose flexible RA capacity solutions. Proposers are not required to fully provide the minimum of 10MW of desired capacity but are encouraged to include as much capacity as can be realistically achieved given the stated timelines.

Proposers must describe the mechanism through which RA capacity will be provided and controlled. The LSEs anticipate the primary mechanism to be PDR. If an alternative mechanism is proposed, the Proposer shall provide relevant case studies/examples of RA capacity provided via the proposed mechanism or other evidence that such a mechanism is capable of meeting the LSEs' RA obligations.

For each category of RA capacity proposed, the Proposer shall note the total capacity, deployment date and term being proposed. Proposers are encouraged to include varied term lengths (from 2-10 years) but the LSEs will not exclude responses that include proposed terms of a shorter length or longer length. Capacity amounts and pricing shall be presented in Attachment A: Pricing &

Capacity Worksheet. Each proposal submitted to each separate LSE should include pricing in a separate version of Attachment A.

Outside of Attachment A, specific attributes of local and flex RA should be noted in this section of the proposal narrative. If a Proposer is including local RA, the Local Area (see "Eligible System Locations") and corresponding capacity should be included. If a Proposer is including flexible RA, the capacity amount and associated flex category (1-4) shall also be noted in this section. Proposers are encouraged to use the narrative in this section of the proposal to explain any nuance in their pricing proposal that is not captured in Attachment A.

To accompany the stated amount of RA capacity provided, Proposers shall provide a description of the Net Qualifying Capacity (NQC), Qualifying Capacity (QC), and if applicable, effective flexible capacity (EFC) calculation methodologies used. Proposers should reference the California Independent System Operator ("CAISO") criteria for calculating the NQC of distributed energy resources. In addition to describing the methodologies, Proposers shall provide the anticipated amount of additional capacity to be acquired in order to reliably and consistently deliver the proposed contracted capacity of RA. Proposers shall also provide a description of their approach, methodology and experience for completing CPUC Load Impact Protocols, adopted in D.08-04-050, and all CPUC rules for RA from DER aggregations.

If the Proposer intends to rely on a significant amount of Self Generation Incentive Program ("SGIP") funds or other incentives/programs in the project development process (see section on *Go-to-Market Strategy*) the Proposer should describe how they intend to comply with the applicable program/incentive rules while consistently and reliably meeting the proposed RA capacity. For example, if the proposal relies on SGIP funding, the Proposer should describe how they will meet any and all operational requirements, including, but not limited to greenhouse gas reduction, battery cycling, and round-trip efficiency requirements.

Proposers are required to provide a consistent amount of RA capacity throughout the entire year. The only exception to this requirement is the potential offer of increased RA capacity in August and/or September of a given year.

2. Deployment Timeline

In order to align with possible PSPS events, IRP requirements, and RA filing requirements, delivery of a portion of the Proposers promised capacity should be achieved by September 2020 and the remainder by either June 2021 or September 2021.

The deployment targets of September 2020 and September 2021 are intended to align with the respective fire seasons and RA filing requirements, while the June 2021 deployment date is intended to align with IRP requirements. The LSEs do not have specific requirements for capacity amounts by the aforementioned dates, only that the Proposer

deliver the promised capacity on that schedule. Across the entire solicitation, the LSEs expect that a portion will be deployed by September 2020 and the remainder deployed by June and/or September 2021. The target RA delivery targets and deadlines by LSE are shown below. Proposals shall include a discussion of key milestones to deliver the proposed capacity on this timeline and how the Proposer will report such milestones to the LSE throughout the process. This timeline should also include expected periods of marketing and outreach activities, system sales/financing and system design, permitting and installation.

	EBCE	PCE	SVCE	SVP
Target for RA Delivered by September 2020 (MW)	3-4	3-4	3-4	No specific capacity for this delivery date
Target for RA Delivered by June 2021 (MW)	No specific capacity for this delivery date	Preference for Remainder	Preference for Remainder	No specific capacity for this delivery date
Required RA Delivered by September 2021 (MW)	Remainder	Remainder	Remainder	Remainder

RA Delivery Timeline Targets and Deadlines by LSE

3. Target Site Types & Priority Customer Categories

Capacity procured through this solicitation shall have LSE-specific limitations on the type of sites on which it can be located.

For the purposes of this solicitation, for the proposed RA to count as LSE-specific, at least 80% of the capacity comprising the DER aggregation providing RA to the given LSE must be from that LSE's customers. Furthermore, for SVP specifically, all capacity comprising the DER aggregation must be located within the service territories included in SVP's metered subsystem agreement. However, it is important to note that, while the 80% requirement is intended to provide flexibility for Proposers, it is a minimum floor and should be treated as such. The CCAs prefer aggregations that only include projects on their customer's sites. Proposers are encouraged to set a target well above the 80% minimum for participating LSE to ensure that, in the event of program participant customers returning to bundled utility service, the capacity total will not risk approaching the 80% requirement. Additionally, if a Proposer needs to put systems located on non-LSE customer sites into an aggregation (e.g. to meet PDR participation minimums) they need to justify why those systems cannot be placed on LSE customers' sites.

The residential and commercial site categories are defined as including the following site types:

- "**Residential**": Residential single or multi-family dwellings four (4) units and below and;
- "Commercial": Commercial sites and multi-family dwellings five (5) units and above

Municipal sites, defined as City and County accounts in each LSE service territory, are considered separately. For LSEs excluding municipal accounts, a separate solicitation may be issued in the future to cover these accounts. For LSEs allowing for capacity to be installed on municipal sites, any capacity will fall into the "commercial" category. Proposers may offer residential capacity, commercial capacity, or both. The eligible sites types, by LSE, can be found in the table below.

In order to support vulnerable communities, each LSE also has different requirements for capacity deployment on sites supporting low-income residents, medical baseline customers, and residents and businesses located in DACs and low income communities. The Proposer shall ensure that proposed RA capacity includes the distribution of that capacity across priority customer categories, as noted below for each LSE. Proposers shall also include in their proposal a plan to reach the required customer categories (See Customer Engagement & Go-to-Market Strategy).

- For capacity in the Residential site category, a minimum of proposed RA capacity may be required to be sited on residential properties in DACs, low-income communities, medical baseline customer properties and/or CARE/FERA customer properties. LSEs will work with Contracted Awardees to develop an outreach strategy for medical baseline and CARE/FERA customers.
- For capacity in the Commercial site category, a **minimum** of the proposed RA capacity may be required to be sited on commercial sites in DACs, low-income communities and/or multi-family properties above four (4) units.
- RA capacity above any minimums set may be sited inside or outside of DACs or low-income communities.

As discussed above, the Proposer is allowed, but not required, to propose RA capacity across the Commercial and Residential sectors, provided that a sufficient implementation process is described in the Customer Engagement & Go-to-Market Strategy. **However, if** a **Proposer includes both residential and commercial capacity the Proposer must** demonstrate that the minimum requirement of the total residential capacity is located in/on a DAC, low-income community, medical baseline customer property and/or CARE/FERA customer property, and that the minimum requirement of the

total commercial capacity is located in a DAC and/or on multi-family properties above four (4) units.

Pursuant to SB 535, CalEPA designated a list of DACs throughout the state for targeted investments from California's cap-and-trade program. For the purposes of this solicitation, DACs are those in the top 25% of scoring on the CalEnviroScreen 3.0 tool, available <u>here</u>. "Low-income communities" are census tracts with median household incomes at or below 80 percent of the county median income of the primary county of a given LSE or with median household incomes at or below the threshold designated as low income by the Department of Housing and Community Development's list of state income limits adopted pursuant to Health and Safety Code (H&SC) Section 50093. Please note, EBCE, SVCE and PCE have provided a list of qualifying low-income census tracts that should be used by Proposers. Expanded definitions are included in the "Definitions & Supporting Information" section.

Full definitions of DACs and low-income communities, as used in this RFP, are included in the "Definitions & Supporting Information" section.

	EBCE	PCE	SVCE	SVP
Minimum RA Required to be Sited in Residential Sites (MW)	5	5	5	0.7
Eligibility of Municipal Sites within the Commercial Category	Not eligible - dedicated solicitation expected at a later date	Eligible only in low- income and DAC communities - dedicated solicitation expected at a later date	Eligible, with a dedicated solicitation expected at a later date	Eligible with a preference in DAC or BAAQMD CARE Communiti es
Minimum Residential RA Required to be Sited in DACs, Low-Income Communities, Medical Baseline Customer Properties and/or CARE/FERA Customers	20%	5%	Not required, but encouraged	10%
Minimum Commercial RA Required to be Sited in DACs, Low-Income Communities and/or Multi-Family Properties above our units	20%	5%	Not required, but encouraged	10%

Priority Customer Type Minimums & Requirements by LSE

4. Eligible DER System Types

The following DER system types are eligible for the total RA capacity proposed by Proposers, provided that operational requirements for the provision of RA can be met:

- New battery energy storage (BES) systems
- New combined solar photovoltaic (PV) + BES systems
- BES retrofits on existing PV systems
- Existing BES systems (provided that dispatchable capacity can be contracted by the LSE and is not otherwise contracted)
- All systems must be able to island from the grid to provide resilience to participating customers

The LSEs reserve the right to review the developer - customer agreement in order to approve the contractual obligation for back-up power/resilience agreed upon by the Proposer and a customer at a given site.

For both new and retrofit BES systems in the proposed capacity the Proposer shall describe the process for managing site control and ensuring that necessary site access exists to maintain and dispatch the system.

Regardless of eligible DER system type, all DER systems included in proposed capacity must meet all CAISO PDR participation requirements. If Proposers include a mechanism other than PDR, Proposers shall include a description of how eligible DER systems meet the relevant requirements of the alternative mechanism.

While the eligible DER system types listed above are anticipated to account for the majority of the capacity procured by each LSE, Proposers may submit additional creative and innovative DER solutions for consideration.

5. Eligible Project Locations

Eligible DER systems shall be sited on customer premises as outlined in Parameter 3: *Target Site Types & Priority Customer Categories*. Each LSE's systems shall also be installed within communities served by that LSE to support the resiliency of those communities. The full list of eligible communities for each LSE is shown below. The LSEs appreciate that Proposers may be able to offer DER systems that satisfy local and/or system RA requirements outside of an LSE's service area. The LSEs will not automatically exclude proposals that include geographies outside of its service area, provided that the RA capacity originating outside of its service territory is clearly indicated, but the LSEs also reserve the right to exclude any of these non-conforming bids from consideration. The price of RA capacity located outside of an LSE's service area, relative to capacity located within the LSE's service territory, shall also be noted for consideration.

Eligible Communities within Service Territory by LSE

Eligible Communities

EBCE	Currently Eligible: Cities of Albany, Berkeley, Dublin, Emeryville, Fremont, Hayward, Livermore, Oakland, Piedmont, San Leandro, Union City, and Unincorporated Alameda County
	Potentially Eligible - cities joining EBCE pending Board and CPUC approval in 2021: Tracy, Pleasanton and Newark
PCE	Anywhere in San Mateo County, including Atherton, Belmont, Brisbane, Burlingame, Colma, Daly City, East Palo Alto, Foster City, Half Moon Bay, Hillsborough, Menlo Park, Millbrae, Pacifica, Portola Valley, Redwood City, San Bruno, San Carlos, San Mateo, South San Francisco, Woodside, and unincorporated San Mateo County
SVCE	13 Communities in Santa Clara County: Campbell, Cupertino, Gilroy, Los Altos, Los Altos Hills, Los Gatos, Milpitas, Monte Sereno, Morgan Hill, Mountain View, Saratoga, Sunnyvale and Unincorporated Santa Clara County
SVP	City of Santa Clara

6. Go-to-Market & Customer Engagement Strategy

The LSEs each seek a robust go-to-market and customer engagement strategy to drive the deployment of the aggregated portfolio of DER systems intended to achieve the proposed RA capacity and increase customer awareness of LSEs as innovative electricity providers delivering benefits to the community. To that end, Awardee(s) will be expected to act as a customer marketing and program engagement partner with the selecting LSE. All Proposers shall include a detailed description of their go-to-market and customer engagement strategy for each priority customer category that Proposer intends to pursue. These strategies should be tailored to the needs of each LSE's community and should be comprehensively explained in each proposal; Proposers are also encouraged to indicate how the program approach could incorporate a regional approach (and impact cost) if the Proposer were selected by multiple LSEs.

An example of go-to-market strategy options is as follows; does the Proposer intend to continue existing marketing efforts in an LSE's service area or does the Proposer intend to create an entirely new product offering and marketing plan to deliver resilience and RA in connection with this solicitation? In short, Proposers should address their desired level of collaboration and co-branding with the LSE to develop and market the program to achieve enrollment targets.

The customer engagement strategy shall describe the creative approach Proposers will take to market the program to ensure customer enrollment. Examples could include but are not limited to collaboration with the LSE, its local government partners, external affinity groups, and/or creation special offer to encourage customer uptake. Within the larger customer engagement strategy, the Proposer shall include a specific description of its comprehensive approach to reaching customers in DACs, medical baseline and/or CARE/FERA customers, and/or low-income communities and multi-family properties.

Proposers shall also include a discussion of their approach to reaching customers with existing forms of DER, such as solar PV and/or BES. This discussion shall include details of how the Proposer would manage retrofits of existing systems, including any barriers associated with existing power purchase agreements, lease contracts, incentive program rules, interconnection requirements, and CPUC and/or CAISO rules or regulations. Proposers shall also address their sales team's experience/conversion rates for this segment of customers.

The LSEs can and will provide a variety of marketing assets to assist with customer engagement and Proposers shall discuss in their proposal how each asset would be leveraged in a successful customer engagement strategy. These assets are listed below for each LSE.

Marketing Assets to be Made Available to Awardee(s)

Available Marketing Assets*

- Anonymized interval load data for initial evaluation and targeting (provided through the Data Provision Process), followed by additional data provision and collaborative outreach
- Municipal (City/County) communication channels for marketing to residents and businesses and other collaborative outreach strategies
- Community based organizations and other affinity group communication channels
- Rooftop solar PV capacity
- CARE customer outreach
- EBCE initiated customer outreach
- Medical Baseline customer outreach
- DAC/Financial Rate Assistance Program (FRAP) outreach*
- Medical Rate Assistance Program (MRAP) outreach*
- BAAQMD Community Air Risk Evaluation Boundary*

*Indicates marketing assets specific to SVP.

Finally, Proposers shall provide, as a key component of their customer engagement strategy, an overview of how the financial benefits and financing options of prospective DER systems will be presented to customers. A line-item explanation/example shall be provided of how the DER "value stack" can improve the program proposition to prospective customers and show the value of LSE service (e.g. RA value provided by LSE as a "discount" to customer, applicable SGIP incentives to customer segments, and/or other incentives).

The Proposer should outline how system operations will be optimized between the value streams of backup power/economic benefits in the form of bill savings to the customer and RA/wholesale market price exposure to the LSE.

Various incentives, financing options and other mechanisms offered to customers should be described in detail in addition to the impacts those could have on the feasibility of the Proposer's offered RA capacity. For example, if a customer project relies on SGIP funds the Proposer should describe how a failure to secure, or a step down in those funds, would impact the Proposer's offered RA capacity, price and/or deployment timeline.

7. Workforce Development Requirements

Workforce development is an approach to economic development that seeks to enhance a region's economic stability and prosperity by focusing on human capital—developing people and their marketable work skills—as opposed to business development. A coherent workforce development system consists of supply side or "push" strategies and demand side or "pull" strategies. This solicitation can be used by LSEs as a "pull" strategy. Proposals are required to describe their workforce plan for project development, construction or maintenance activities, pursuant to the following requirements by LSE.

	Workforce Requirements
EBCE	90% of all workers employed on or in support of the Project(s) by Seller or through Contractors shall be paid not less than the prevailing rate of wages for the appropriate craft, classification, type of worker and locality as determined by the Director of the State Department of Industrial Relations in accordance with Division 2, Part 7, Chapter 1 of the California Labor Code, or as set out in the wage determination of the U.S. Secretary of Labor, whichever is higher. Proposers shall make commercially reasonable efforts to hire Alameda County residents to provide at least 50% of the work hours associated with the construction, operation, and maintenance of the project.
	Additionally, Proposers shall include descriptions of how they intend to partner with workforce development agencies in Alameda County to further enhance opportunities for local hiring. An example of this is prioritization of hiring graduates of state-certified apprenticeship training programs, such as Cypress Mandela and Rising Sun Energy Center, in support of the construction, operation, and maintenance of the project.
PCE	PCE's workforce development requirements are publicly stated here: peninsulacleanenergy.com/wp-content/uploads/2018/10/Policy-10- Inclusive-and-Sustainable-Workforce-revised-10-25-18.pdf.
	PCE therefore desires to facilitate and accomplish the following objectives: (1) Support for and direct use of local businesses; (2)

Workforce Requirements by LSE

	Support for and direct use of union members from multiple trades; (3) Support for and use of training and State of California approved apprenticeship programs, and pre-apprenticeship programs from within PCE's service territory; and (4) Support for and direct use of green and sustainable businesses.
SVCE	SVCE does not have specific workforce requirements for the Proposer's response to this solicitation. However, the proposal shall include the workforce plan as outlined above, and should indicate how requirements for prevailing wages, hiring local residents, and partnering with workforce development agencies would impact the proposed cost structure and timeline.
SVP	SVP does not have specific workforce requirements for the Proposer's response to this solicitation. However, the proposal shall include the workforce plan as outlined above, and should indicate how requirements for prevailing wages, hiring local residents, and partnering with workforce development agencies would impact the proposed cost structure and timeline.

8. Safety and Fire Prevention

Proposals are required to include a description of the Proposer's approach to safety and fire prevention that will be applied across projects developed under this solicitation. Proposers should detail how fire safety is prioritized in each project through factors including system design, equipment supply and location of installed units. If applicable, Proposers should also note how the detailed fire safety approach impacts system pricing. Proposers should note fall protection safety management plans if solar is included in the program.

NFPA 855

The new NFPA 855 aims to mitigate risks associated with energy storage systems and ensure that all installations are done in a way that takes fire and life safety into consideration. In addition to looking at where the technology is located, how it is separated from other components, and what fire suppression systems are in place, NFPA 855 considers the ventilation, detection, signage, listings, and emergency operations associated with energy storage systems.

Proposals must be fully compliant with all applicable safety and fire codes, certifications and standards (see Certifications & Standards below), including, but not limited to NFPA 855. As part of their approach to fire safety and prevention, Proposers should describe how they intend to meet new NFPA 855 requirements, and how they plan to coordinate with Authorities Having Jurisdiction (AHJs) on permitting for BES. To the extent meeting these standards will require additional time, expense, or may limit eligible customer locations, Proposers should describe any potential impacts and mitigations that may be

necessary.

Shortlisted Proposers will be required to provide a detailed Safety Plan as part of the contracting process that will be followed during project implementation.

9. Certifications and Standards

Proposers shall provide confirmation that the project, including all equipment, installation and interconnection standards and certifications will be met, including but not limited to the following:

Certifications and Standards	Purpose
UL 1642 Standard for Lithium Batteries (Cells)	Sets the bar for basic safety incorporated into each individual cell
UL 1741 Standard for Inverters, Converters, Controllers and Interconnection System Equipment for Use with Distributed Energy Resources	Ensures all equipment can safely interact with, and in some cases even support the utility grid.
UL 1973 Standard for Batteries for Use in Light Electric Rail and Stationary Applications	Includes requirements for propagation resistance to single cell thermal runaway; includes an internal fire test that ensures that propagation to neighboring cells does not occur
UL 9540 Standard for Energy Storage Systems and Equipment	Combines applicable standards for individual components assembled into a fully integrated system.
IEC 62109-1 Safety of power converters for use in photovoltaic power systems	Defines the minimum requirements for the design and manufacture of power conversion electronics for protection against electric shock, energy, fire, mechanical and other hazards
IEC 62619 Safety requirements for large format secondary lithium cells and batteries for stationary and motive applications	Internationally recognized standard includes both cell and battery tests for failure related to abnormal electrical and environmental conditions.
AC 156 Seismic Acceptance Criteria	Equipment must maintain integrity during earthquakes

Note: Proposers are welcome and encouraged to include examples of additional certs/standards that address greenhouse gas emissions and/or other environmental considerations.

10. Summary of Differences in Mandatory Proposal Parameters by LSE

Mandatory Proposal Parameter # and Requirement		EBCE	PCE	SVCE	SVP
1	Total Target MW Target	10 MW	10 MW	10 MW	2.7 MW
2	Target RA Delivery Dates (1. Sept 2020, 2. June 2021, 3. Sept 2021)	1. 3-4 MW, 2. no preference for remainder, 3.remainder	 3-4 MW, preference for remainder, remainder 	 3-4 MW, preference for remainder, remainder 	No specific delivery date
3	Eligibility of Municipal Sites	Ineligible	Eligible for DAC communities	Eligible	Eligible with preference in DAC/ Low- income /BAAQMD CARE boundaries
3	Minimum Residential Site Category Capacity	5 MW	5MW	5 MW	0.7 MW
3	Priority Customer Type Requirements (Res)	20%	5%	Not required, but encouraged	10%
3	Priority Customer Type Requirements (Com)	20%	5%	Not required, but encouraged	10%
4	Eligible DER System Types	No differences.			
5	Eligible Project Locations	Customers in relevant LSE service territory.			
6	Go-to-Market & Marketing Strategy Content	No Differences.			

6	Marketing Assets Available	See Marketing Assets to be Made Available to Awardee(s) by LSE table on page 19.
7	Workforce Development Requirements	See Workforce Requirements by LSE table on page 21. Also see weighting criteria for differences in scoring.
8	Safety & Fire Protection	No Differences in submittal. See weighting criteria for differences in scoring.
9	Certifications & Standards	No Differences.

Definitions & Supporting Information

To provide additional clarity and guidance for Proposers, this section includes definitions of key terms that appear throughout the RFP and additional detail on the customer load data to be provided to Proposers in order to inform accurate customers.

Definitions

For the purposes of this RFP, the following terms are defined as such:

"Commercial Customer": Refers to a category of customer sites types consisting of commercial sites and multi-family dwellings five (5) units and above.

"Disadvantaged Communities (DACs)": Refers to CalEPA's list of designated communities, pursuant to SB 355 throughout the state for targeted investments from California's cap-and-trade program. For the purposes of this solicitation, DACs are those in the top 25% of scoring on the CalEnviroScreen 3.0 tool, available here.

"Low-income communities": Refers to census tracts with median household incomes at or below 80 percent of the median income for the primary county of each LSE, listed as follows;

EBCE: 80% of Alameda County median income (\$81,700) PCE: 80% of San Mateo County median income (\$99,540) SVCE: 80% of Santa Clara County median income (\$101,284) SVP: 80% of Santa Clara County median income (\$101,284)

Or census tracts with median household incomes at or below the threshold designated as low income by the Department of Housing and Community Development's list of state income limits adopted pursuant to Health and Safety Code (H&SC) Section 50093.

Please note, EBCE, SVCE and PCE have provided a list of qualifying low-income census tracts that should be used by Proposers to remove any uncertainty.

"Priority Customer Categories": Refers to the following property types:

- residential properties in disadvantaged communities (DACs), low-income communities, medical baseline customer properties and/or CARE/FERA customer properties and;
- commercial sites in DACs, low-income communities and/or multi-family properties above four (4) units.

"Residential Customer": Refers to a category of customer sites types consisting of residential single or multi-family dwellings four (4) units and below.

Detail of Anonymous Customer Data

As discussed in the "Data Provision Process" section, a sample of customer data will be provided to interested parties in order to inform accurate pricing of RA capacity. The key characteristics of the data to be released by each LSE are as follows.

	EBCE	PCE	SVCE	SVP
Quantity of Commercial Customers	5% of customers	5% of customers	5% of customers	5% of customers
Commercial Rate Types	A1, A6, A10 and E19	A1, A6, A10 and E19	A1, A6, A10 and E19	C-1, CB-1
Customer Flags Included	Rate class (e.g. NEM), DAC, Low-Income Area	Rate class (e.g. NEM), DAC, Low-Income Area	Rate class (e.g. NEM), DAC, Low-Income Area	Rate class (e.g. NEM), DAC, Low-Income Area, FRAP, MRAP
Residential Customer Data	Not available for RFP, but may be shared with Awardees	Not available for RFP, but may be shared with Awardees	Not available for RFP, but may be shared with Awardees	Not available for RFP, but may be shared with Awardees
Solar Data	Estimated solar potential (kW) of roof and tilt and orientation of arrays from Google SunRoof	Estimated solar potential (kW) of roof and tilt and orientation of arrays from Google SunRoof	Estimated solar potential (kW) of roof and tilt and orientation of arrays from Google SunRoof	Estimated solar potential (kW) of roof and tilt and orientation of arrays from Google SunRoof
Available Data Time Range	1/1/2018 to 12/31/2018	1/1/2018 to 12/31/2018	1/1/2017 to 12/31/2017	7/1/2018 to 6/31/2019

Anonymous Customer Data Provided to Bidders by LSE

Terms & Conditions

GENERAL

Incurring Cost: This RFP does not commit the LSEs to award or pay any cost incurred in the submission of the proposal, or in making necessary studies or designs for the preparation thereof, nor procure or contract for services or supplies. Further, no reimbursable cost may be incurred in anticipation of a contract award.

Claims Against the LSEs. Neither your organization nor any of your representatives shall have any claims whatsoever against the LSEs or any of its respective officials, agents, or employees arising out of or relating to this RFP or these RFP procedures, or the award of any contract pursuant to this RFP.

Guarantee of Proposal. Responses to this RFP, including proposal prices, will be considered firm and irrevocable for one hundred eighty (180) days after the due date for receipt of proposals and/or one hundred eighty (180) days after receipt of a best and final offer, if one is submitted.

Basis for Proposal. Only information supplied by the LSEs in writing by jross@ebce.org in connection with this RFP should be used as the basis for the preparation of proposals.

Form of Proposals. Proposals must be submitted electronically by e-mail.

Amended Proposals. Respondent may submit amended proposals before the Deadline to Submit Proposals. Such amended proposals must be complete replacements for previously submitted proposals and must be clearly identified in a written format. The LSEs will not merge, collate, or assemble proposal materials.

Withdrawal of Proposal. Respondents may withdraw their proposals at any time prior to the Deadline to Submit Proposals. Respondents must submit a written withdrawal request signed by the respondent's duly authorized representative addressed to and submitted to the LSE's Authorized Representative.

Late Responses. To be considered, proposals must be received electronically by email by December 23rd, 2019, 5:00 PM PST.

No Public Proposal Opening. There will be no public opening for this RFP.

California Public Records Act (CPRA). All proposals become the property of the LSEs, which are public agencies subject to the disclosure requirements of the California Public Records Act ("CPRA"). If a respondent's proprietary information is contained in documents submitted to any of the LSEs, and respondent claims that such information falls within one or more CPRA exemptions, the respondent must clearly mark such information "CONFIDENTIAL AND PROPRIETARY," and identify the specific lines containing the information. In the event of a

request for such information to any LSE, the LSE in question will make best efforts to provide notice to respondent prior to such disclosure. If respondent contends that any documents are exempt from the CPRA and wishes to prevent disclosure, it is required to obtain a protective order, injunctive relief, or other appropriate remedy from a court of law in the individual LSE's county, before the given LSE's deadline for responding to the CPRA request. If respondent fails to obtain such remedy within the LSE's deadline for responding to the CPRA request, the LSE may disclose the requested information. Respondents and the selected Awardee further agrees that it shall defend, indemnify, and hold each LSE or its agents, harmless against any claim, action, or litigation (including, but not limited to, all judgments, costs, fees, and attorney's fees) that may result from the LSE's assertion of an exemption or privilege as a basis for withholding any information marked confidential by the respondent.

Confidentiality. All data and information obtained from or on behalf of the LSEs by the respondent and its agents in this RFP process, including reports, recommendations, specifications, and data, shall be treated by the respondent and its agents as confidential. The respondent, selected Proposer(s) and its agents shall not disclose or communicate this information to a third party or use it in advertising, publicity, propaganda, or in another job or jobs, unless written consent is obtained from the LSEs. Generally, each proposal and all documentation, including financial information, submitted by a respondent to the LSEs is confidential until a contract is awarded, when such documents become public records, unless exempted under the CPRA.

Response is Genuine: By submitting a response pursuant to this RFP, Respondent certifies that this submission is genuine, and not sham or collusive, nor made in the interest or on behalf of any person not named therein; the submitting firm has not directly or indirectly induced or solicited any other submitting firm to put in a sham bid, or any other person, firm or corporation to refrain from submitting a submission, and the submitting firm has not in any manner sought by collusion to secure for themselves an advantage over any other submitting firm.

Electronic Mail Address. Most of the communication regarding this procurement will be conducted by electronic mail (e-mail). Respondents agree to provide jross@ebce.org with a valid e-mail address to receive this communication.

LSE Rights. The LSEs reserve the right to do any of the following at any time:

- a. Reject any or all proposal(s), without indicating any reason for such rejection;
- b. Waive or correct any minor or inadvertent defect, irregularity, or technical error in a proposal or the RFP process, or as part of any subsequent contract negotiation;
- c. Request that respondents supplement or modify all or certain aspects of their proposals or other documents or materials submitted;
- d. Terminate the RFP, and at its option, issue a new RFP;
- e. Procure any equipment or services specified in this RFP by other means;
- f. Modify the selection process, the specifications or requirements for materials or services, or the contents or format of the proposals;
- g. Extend a deadline specified in this RFP, including deadlines for accepting proposals;
- h. Negotiate with any or none of the respondents;

- i. Modify in the final agreement any terms and/or conditions described in this RFP;
- j. Terminate failed negotiations with any Proposer without liability, and negotiate with other Proposer(s);
- k. Disqualify any respondent, including a selected Proposer, on the basis of a real or apparent conflict of interest, or evidence of collusion that is disclosed by the proposal or other data available to the LSEs;; and/or
- I. Accept all or a portion of a Proposer's proposal



East Bay Community Energy, Peninsula Clean Energy, Silicon Valley Clean Energy & Silicon Valley Power

Distributed Resource Adequacy Capacity Request for Proposal (RFP)

Addendum #1

Issuance Date: November 19, 2019

The following Addendum #1 includes all questions received during the pre-proposal webinar

and via electronic communication to this point. Similar questions have been combined for brevity.

1. Confirmation on the Energy Products being procured. Page 9. Is this solicitation exclusively for the rights to Resource Adequacy (RA) or does this solicitation envision other energy products to be eligible to the buyer (i.e. energy product revenues, ancillary services, buyer dispatch)?

While the LSEs are focused on procuring RA they will certainly entertain and appreciate proposals that include options for dispatch that would enable benefits to the given LSE from various services.

In Section 6 the RFP states: "The Proposer should outline how system operations will be optimized between the value streams of backup power/economic benefits in the form of bill savings to the customer and RA/wholesale market price exposure to the LSE".These wholesale benefits could be expanded to include other services that benefit the LSE.

- 2. Residential types. Page 16. Is there any benefit in exceeding the thresholds listed for DACs? For example, if there were two bids with similar pricing but one bid had 20% DACs and the other had 60% DACs, would the latter be preferred? If so, where does that show up in the Evaluation rubric Quality and Detail of Go-to-Market & Customer Engagement Plan?
 - a. Is there a minimum requirement for commercial customers? There appears to be a minimum RA required to be sited on residential sites on Page 15.

A higher proportion of Low Income/SAC customers enrolled would be positively viewed by the LSEs.

There is no minimum for commercial customers but there is a minimum for residential customers, as stated on RFP page 15.

3. If not Proxy Demand Response (PDR) what other RA mechanisms are eligible?

The LSEs are assuming PDR is the main RA mechanism, but if there are others that Proposers are aware of then they are welcome to include them.

4. Does each LSE perform their evaluation independently?

Yes, proposal submittals, evaluations and the ensuing contracting will be done independently but the LSEs reserve the right to discuss proposals internally with the goal of establishing regional best practices.

5. Will there be a list of webinar participants circulated to support teaming?

Yes, it was emailed to webinar attendees following the webinar.

6. How, if at all, must the DERs be aggregated?

DERs must be aggregated in a way that they can meet LSE RA obligations through the relevant (likely PDR) mechanism for providing RA. If multiple Proposers chosen, each one will have to aggregate their own resources – the LSEs don't expect awardees to coordination aggregation.

Are you going to be viewing proposals with gas any differently than solar + battery proposals? Can solar + battery + gas proposals count for full RA Capacity under this RFP.

Fossil-fuel fired generation is not eligible for inclusion in Proposals submitted under this RFP.

8. Do LSEs have forecast of RA capacities they are going to procure in 2020 and 2021?

The LSEs know their overall RA requirements, and this solicitation is for a small subset of that requirement. The LSEs are not making their RA obligations public.

9. Are there more details on LSE marketing assets other than in RFP?

In addition to the resources explained in the proposal, will LSE's have designated people/hours they can commit to helping with the GTM/customer engagement strategy once a proposer is selected?

No, the marketing assets listed in the RFP are the assets that will be provided. If there are other assets that a Proposer wants to request that could improve the program, the LSEs are open to considering other options.

Designated staff/hours for GTM/customer engagement strategy will be worked out through negotiation process w/ shortlisted proposers, depending on what was proposed in the GTM/customer engagement strategy.

10. Must the aggregated DER be dedicated for CCA use or is each individual site's capacity to be dedicated for each LSE?

Each DER aggregation will be dedicated to the LSE that it is proposed to. Proposers will be contracting with each LSE separately.

11. Will CCAs consider CAISO Metered-Generator-Output (MGO) as the performance evaluation methodology instead of the traditional DR baseline?

Proposers are asked to disclose how they are calculating the NQC of their

aggregation. Provided that the calculations and methodologies included are aligned with relevant CAISO and CPUC requirements the LSEs will consider whichever performance evaluation methodology is included by the Proposer.

12. Can you please repeat the minimum RA capacity required per proposer? Can it be aggregated over multiple sites? in KW and kWh

I was not able to see the minimum proposal requirements. Could you repeat the minimum project size?

Is the 500kW minimum, 500kW of commercial OR residential (with a minimum 1MW for a mixed commercial/residential proposal?)

This minimum proposal size is 500 kW. This minimum proposal size is driven by the minimum aggregated capacity for PDR participation. This may be aggregated over multiple sites. For example, the 500kw can be comm, res or both (250kW on each).

13. Is there a standard form of contract across the LSE's or will each LSE be using their own contract?

Yes, the LSEs are working on a standard form for use, but it isTBD by each LSE legal team as to whether the standard form will be used as is, or whether changes will be made.

14. Will LSEs share the list of their customers that we can market our solar/storage/microgrid solutions including LSE's RA capacity?

Will the solar capacity per site be made available in the data room?

The LSEs are providing anonymized info for commercial customers now through the data room to facilitate initial pricing. There is also the potential to share more customer data later during the contracting and implementation process. The details of that process will be LSE specific.

Estimated solar capacity and annual production from Google SunRoof is available along with corresponding anonymized interval data in the data room. Instructions for accessing this data can be found in the RFP.

15. Difference in other assumptions can make the bids quite different. Will the following be provided:

- a. -assumptions of electrical infrastructure that would guide costs needed for islanding
- b. value that needs to be given to the customer, as the remaining value will guide the RA price

Assumptions of electrical infrastructure needed for islanding will not be provided,

up to each Proposer.

The LSEs expect that the RA revenue stream will lower cost of ownership/operation for customer – there will be a flow through of the RA payment from LSE to customer. The lowered cost should increase participation rates. The LSEs expect proposers to know what pricing is necessary to achieve this increased participation and will price RA payments in accordance with what you know is necessary to drive adoption.

16. Given there may be multiple awardees, is there a breakdown of the individual parts of the proposal than an individual proposer can respond to? Is that based primarily on capacity?

RA Capacity must be a part of all proposals. The LSEs understand that this proposed capacity may change based on # of awardees selected (e.g. a Proposer includes res and comm sited capacity but is only chosen to provide res capacity). This will be discussed through negotiation process.

The quality of GTM approach, price and capacity will all be evaluated.

17. How are LSEs balancing the workforce requirements with their rapid proposal and deployment timelines and cost competitiveness requirements?

Is the prevailing wages requirement be forced for residential installations? Is the union personnel required?

The LSEs are relying on Proposers to balance between meeting workforce requirements and meeting specified deployment timelines. Concerns or restrictions can be noted in proposals and will be weighed during the evaluation of various proposal sections.

Prevailing wage and union requirements differ by LSE and are included on RFP p. 19.

18. Can the CCA's facilitate providing bill credits to consumers for participation?

This is an interesting proposal, the LSEs are interested in seeing how it fits into a Go-to-Market strategy

19. What are the resiliency requirements per facility/project?

Is there a minimum length of time that capacity needs to be supplied by the battery system? (is there a minimum energy capacity preferred or specified?)

The LSEs do not plan to create resilience minimums – because minimums are customer specific and should be discussed between Proposers and customers. The duration needed for resilience is for Proposers to evaluate and offer to customer.

Capacity for RA does have specific durations for local and flex RA and the balance of these needs should be optimized by Proposer.

20. Do the LSEs expect that the chosen proposer and the LSE will coordinate for customer acquisition even for the Sept. 2020 timeline or do you need the proposer to have actual projects that they can commit for Sept. 2020 delivery in order to participate in the RFO?

Is the timeline based on commercial operation/permission to operate or will completed installation be considered for the 2020 deployment

The LSEs do not require that proposer has existing projects and plan to coordinate with Proposer on customer acquisition for first timeline. Targeted and focused Go-to-Market strategies are very important.

The September 2020 deadline coincides with an RA filing and is for systems that will be online, operational in January 2021 for RA purposes. The September 2020 first delivery allows for 3 months of cushion to finalize projects. As long as the LSEs can count those assets in 2021 RA filing, there could be some flexibility on that Sept 2020 timeline. The LSEs recognize that there is a CPUC registration process for RA that takes time as well.

The target delivery dates in the RFP do NOT have mandatory capacity amounts associated with them (just target amounts). Proposers should indicate how much capacity is realistic for delivery by each target date based on their existing customer pipelines, customer acquisition rates, etc.

21. Are proposers liable for any penalties due to a failure to meet RA requirements?

Yes, penalties will be in commercial contract. This RA will be like any other resource with damages for failure to show.

22. Are the LSEs prepared to provide any credit security for the proposer-LSE RA contracts?

This will be dealt with on an LSE specific basis during contracting.

23. Is the SunRun - EBCE contract announced earlier this year providing RA? If so, is that using PDR as the RA mechanism?

Yes, the SunRun contract provides RA and used PDR as the mechanism.

24. Will the LSEs be providing / publishing a list of CAISO Scheduling Coordinators to help developers find SCs for CAISO bidding as PDR resources?

The LSEs are not planning to release such a list at this point. The LSEs refer Propoers to the CAISO list of schedule coordinators.

25. What are the integration requirements for aggregated assets for dispatch and M&V?

The LSEs do not have specific dispatch and metering protocols or requirements at this time. Please see Question 1 and Question 32 for more information on dispatch.

26. Is there a list of storage hardware vendors that the projects should focus on? Is there any guidance on technologies?

If a resource can provide RA but does not provide any resilience benefit, is it eligible to include in an aggregation?

There is no list of preferred hardware vendors.

Due to the dual goals of RA and resilience contained in this RFP, the LSEs are only looking for resources that provide both. Only solar +storage (both new and storage retrofits), storage only and any other DER solution capable of providing RA and resilience that a Proposer may be aware of will be considered.

Generally, the LSEs are also procuring RA at this time, separate of this RFP. Thus, if there are other technologies and providers that are in the market, they are encouraged to monitor for other solicitations.

27. Part of the evaluation criteria includes "Fire Safety Measures". Is there an expectation that proposers should be going above and beyond current building and electrical code requirements? Does storage technology (e.g. Lithium Iron Phosphate vs. NMC) come into play here?

There is not an expectation that Proposers will go beyond code requirements, but fire safety is extremely important to each LSE. This is why it is included as a weighting criteria. Proposers are encouraged to include any additional fire safety measures (beyond code) they are willing to take.

Storage technology is relevant to the extent that some chemistries have higher fire risk. In their fire safety measures Proposers should describe the measures relevant to their proposed technologies.

28. Does the "at least 80% of the capacity comprising the DER aggregation must be from that LSE's customers" requirement mean that 80% of
capacity must be local RA? Or that 100% of capacity must be local RA, but 20% can be from local resources who are not CCA customers?

If a customer who is providing RA elects to shift back to utility service, will they no longer be eligible to participate in this program? Or only if they would increase the portfolio to below 80% of CCA customers?

This sentence refers to the latter example provided in the question. The LSEs realize some customers in their service territories may still be PG&E (or other) customers and that sub-LAPs may cross LSE service territories, so a minimum of 80% of the customers in a given aggregation proposed to an LSE must be LSE customers. But 100% of the capacity proposed needs to fit the local RA requirements. If a Proposer bids local RA, it should be local.

The CCAs recognize that CCA customers have choice, and if they were to opt out, they would still be eligible for this program. The CCAs have not specified that customers have to remain in the CCA to participate in the program.

However, it is important to note that, while the 80% requirement is intended to provide flexibility for Proposers, it is a minimum floor and should be treated as such. Proposers are encouraged to set a target well above the 80% minimum for participating LSE to ensure that, in the event of program participant customers returning to bundled utility service, the capacity total will not risk approaching the 80% requirement. Additionally, if a Proposer needs to put systems located on non-LSE customers into an aggregation (to meet PDR minimums, etc) they need to justify why those systems cannot be placed on LSE customers.

29. Do you have any site guidelines that must be taken into consideration (for commercial storage only), besides grid access efficiency?

There are no specific site guidelines, only customer cohort guidelines (see Target Site Types & Priority Customer Categories on RFP p. 14).

30. Where do I find Attachment A: Pricing & Capacity Worksheet.?

It is available for download at Ebce.org/solicitations

31. What are the priority target markets? You mentioned San Jose and Palo Alto, but these cities are served by different LSEs. Would SJCE and PA Utility participate in this RFP?

Are the LSEs able to provide more accurate definitions of the boundaries of their service territories, through zip codes, census tracts, or other designations? Or is the best approach to use the county-level approach you mentioned earlier, and just remove those certain cities?

The eligible and priority markets for this RFP are the service territories of the

issuing LSEs. The simplest way to delineate this area is that it includes Alameda, San Mateo, Santa Clara counties, excluding the cities specified in the webinar slides that do not receive service from the LSEs issuing this RFP. The excluded cities are the City of Alameda, San Jose, Palo Alto, Pleasanton and Newark.

However, Pleasanton, Newark and Tracy, which is not in Alameda County, will be served by EBCE in January of 2021 so will become eligible for the second milestone year for Proposers proving proposals to EBCE.

32. Are CCA's requiring provisions in customer agreements as part of their participation in this program? Any requirements that the CCA's have on T's and C's for agreements with customers?

Will there be a requirement for developers to document customers to "opt in" to this program or will the dispatch of the resource effectively be "invisible" to the customer?

Generally, the customer will need to opt-in to this program, and if the customer opts-out then the RA payments will cease and the Proposer will need to replace with other capacity. This opt-in make take place between the customer and the LSE, or be included in the customer - Proposer agreement, but this will be dealt with on an LSE specific basis as part of the GTM strategy. It is currently being considered and various LSEs may require a dedicated agreement between the customer and the LSE in addition to the Proposer - customer agreement.

Pending the handling of an "opt-in" clause, the LSEs are not currently specifying any requirements for the Proposer - customers agreement but various LSEs may require review/approval of the Proposer - customer agreement. This is currently being considered and will be dealt with during the contracting process.

The Proposer should be prepared to have the dispatch of the resource be invisible to the customer and the LSEs expect bidders to know how batteries can be optimized across RA requirements (e.g. must offer obligation, if applicable) and customer value streams (e.g. Demand charges and RA periods for commercial customers will most likely be different, as the peak demand may not overlap with RA time periods). However, if Proposers are able to work with customers who may desire more visibility into the process as part of the GTM strategy that is desirable.

33. The CPUC has dual participation restrictions on PDR and other DR programs with the IOUs. e.g. Can't do PDR and CPP. Are you aware of similar restrictions for CCA customers?

The LSEs are cognizant of these restrictions for IOUs but not aware of any additional restrictions for CCA customers. Proposers must make sure that assets can be aggregated for RA and successfully participating in PDR or other

proposed mechanism.

34. Are Silicon Valley Power customers eligible for the SGIP incentive for energy storage?

Yes, if they receive gas service from PG&E

35. The RFP details terms of 2-10 years with flexibility to propose shorter or longer terms. Is there any weighting on longer terms (within the 10 years)?

There is no weighting for longer terms. Term preference is LSE specific.

36. Would the Solar+Storage or standalone ESS be required to be capable of operating in island mode? How are you determining that aggregated resources are island capable?

Yes, isanding capability is a requirement across this program and solicitation. Systems need to be individually capable of islanding for the site. Confirming that assets will provide back-up power will be a part of go to market strategy.

37. BTM solar "kills" traditional PDR baselines. Can you determine what aspect of PDR the SunRun contract is using to avoid this problem and add that info to the RFP docs? Is it Permanent Load Shift?

The Sunrun contract is confidential. Proposers will need to confirm how much RA they can provide based on the DER systems they can develop.



LICON VAI



East Bay Community Energy, Peninsula Clean Energy, Silicon Valley Clean Energy & Silicon Valley Power

Distributed Resource Adequacy Capacity Request for Proposal (RFP)

Addendum #2

Issuance Date: December 5, 2019

The following Addendum #2 includes all questions received via electronic communication and not included in Addendum #1. Similar questions have been combined for brevity.

1. How will the parties work together to provide or integrate with the LSE for dispatch signals? There is no information in the RFP on the DERMS API integration or platform between the parties to manage DERs by aggregating data and enabling hardware dispatch, back-end infrastructure, etc.

See Addendum #1, Questions #1, #25 and #32.

RA provided through PDR will coordinate dispatch through CAISO PDR rules. If Proposers expect to use an RA mechanism other than PDR, there may be a need for further integration. In that case, Proposers should understand the integration needed with a given LSE to ensure that their proposed mechanism can satisfy RA requirements. LSEs will work with Proposers on a case by case basis during the legal and commercial contracting processes.

2. What additional data will be provided to support customer outreach? Will actual customer data be provided?

See Addendum #1, Question #16. The LSEs are already providing actual anonymized customer data. The potential to share additional customer load profiles during the contracting process and un-anonymized data during the implementation process (post contract execution) will be on an LSE specific basis. Proposers should indicate how possibility for the sharing of additional data after contracting would be incorporated into their Go-To-Market strategy.

3. What is the ability for a 3rd party to reach Low Income and DAC customers? Will these customers be specifically identified?

Low-income & DAC customers have already been identified in the anonymized data provided. See RFP pages 17-19 and Addendum #1, Question #9 for additional discussion of marketing assets provided to Proposers. See Question #11 for Low Income Census tracts.

4. Are special permits required to implement residential projects?

All residential projects will be subject to the permitting requirements of the Authority Having Jurisdiction (AHJ) over a given site.

5. Will the LSEs provide a residential customer 'hotline' for customer questions?

The LSEs have not decided what specific marketing activities they will be undertaking. Proposers are asked to create a Go-To-Market and Marketing Strategy proposing such activities.

6. Is there a preference of which interconnection process is used?

No. Proposers should factor interconnection timelines into their proposed deployment timeline.

7. Do any of the LSEs have a position on specific technologies offered outside of the list provided? Is there anything they do NOT want to consider?

See Addendum #1, Questions #7, #26 and Question #27 in this addendum.

8. Are all CAISO mechanisms options for consideration (e.g., PDR, PGA and DERA)

Yes. See Addendum #1, Question #3.

9. Are there any specific metering requirements for the projects?

See Addendum #1, Question #25.

10. The typical understanding of PDR resources is that resources don't receive any credit for exports beyond site load. However, we're curious whether there may be some nuance to this, specifically that PDR resources don't receive energy market payments for exports, but do receive capacity/RA credit for exports. Can the LSEs confirm whether this is the case, or that PDR resources actually are limited by site load for both energy and capacity credit?

The LSEs are open to any proposals that would allow for the full value of aggregated BTM storage and solar+storage to be fully counted for Resource Adequacy. A proposal to fully value the capacity of BTM batteries for RA would be considered.

11. Are the income thresholds listed for "low-income communities" on pg 24 the median incomes, or 80% of the median incomes? In other words, should we do the math and multiply those income levels by 80%, or have you already done it?

The thresholds listed on RFP p. 24 are 80% of the median household income for the given county, as recorded in the 2018 American Community Survey. To avoid confusion around median income year, each LSE has provided the census tracts in their territory that qualify as "low-income" for the purposes of this RFP. These census tracts are the definitive list of census tracts that qualify as low-income for each LSE.

EBCE	SVCE	PCE	SVP*
6001400700	6085500100	06081610800	5049.01
6001400800	6085500800	06081610900	5050.01
6001400900	6085501700	06081601300	5050.07
6001401000	6085502001	06081610302	5052.02
6001401300	6085502002	06081612000	5053.01
6001401400	6085502201	06081601502	5053.02
6001401500	6085502910	06081602200	5053.03
6001401600	6085503113	06081611900	5053.04
6001401800	6085503116	06081610202	5053.05

Low income Census Tracts by LSE

		(
6001402200	6085503210	06081610400	5054.01
6001402400	6085503214	06081611800	5054.02
6001402500	6085503218	06081610500	5054.03
6001402600	6085503331	06081602100	5055
6001402700	6085503332	06081610203	5056
6001402800	6085503802	06081610201	5057
6001402900	6085503902	06081612100	5059
6001403000	6085503903	06081600800	5060
6001403300	6085504101	06081984300	5061.01
6001403400	6085504102		5061.02
6001403501	6085504412		5061.03
6001403502	6085504422		
6001405301	6085504506		
6001405302	6085504601		
6001405401	6085504602		
6001405402	6085504802		
6001405500	6085505009		
6001405600	6085505100		
6001405700	6085505202		
6001405800	6085505600		
6001405901	6085506501		
6001405902	6085509000		
6001406000	6085509303		
6001406100	6085509403		
6001406201	6085511608		
6001406202	6085511915		
·			

6001406300	6085512026	
6001406400	6085512310	
6001406500	6085512314	
6001406601	6085512506	
6001406602	6085512508	
6001407000	6085512509	
6001407101	6085512602	
6001407102	6085512603	
6001407200	6085512604	
6001407300	6085513000	
6001407400		
6001407500		
6001407600		
6001408200		
6001408400		
6001408500		
6001408600		
6001408700		
6001408800		
6001408900		
6001409000		
6001409100		
6001409200		
6001409300		
6001409400		
6001409500		

6001409600		
6001409700		
6001410100		
6001410200		
6001410300		
6001410500		
6001420400		
6001422100		
6001422300		
6001422400		
6001422500		
6001422700		
6001422800		
6001422900		
6001423100		
6001423200		
6001423500		
6001423602		
6001425104		
6001430900		
6001431000		
6001432200		
6001432400		
6001432501		
6001432502		
6001432600		
	-	-

6001433103		
6001433104		
6001433200		
6001433600		
6001433800		
6001433900		
6001434000		
6001435104		
6001435400		
6001435500		
6001435601		
6001435602		
6001436200		
6001436300		
6001436500		
6001436602		
6001436900		
6001437500		
6001437701		
6001437702		
6001438201		
6001440200		
6001440307		
6001451404		

- * SVP list of Census tracts are all census tracts in SVP territory
- 12. To be crystal clear on the '80% of capacity' rule, please confirm that this interpretation is correct: 80% of capacity must be from resources on LSE customer

sites (with SVP as the exception). The only locational constraint on the remaining 20% of capacity is that it must be 'local RA', meaning that it is in a sub-LAP that overlaps the CCA's territory. If this is correct, can you please provide the sub-LAPs that are eligible for each CCA? Based on our understanding, they are as follows, but would be good to confirm:

EBCE	SVCE	PCE	SVP
PGE_PGEB,	PGE_PGCC,	PGE_PGP2,	PGE_PGP2,
PGE_PGSB	PGE_PGSF	PGE_PGSF	PGE_PGSB

This interpretation is correct. Proposers should also consider, however, that per the "Eligible Project Locations" RFP section, the CCA's are strongly prioritizing projects within their service territory. Locational requirements specific to the proposed RA mechanism may also apply (e.g. all PDR resources in a given aggregation must be in the same sub-lap).

Additionally, proposers should treat this rule as a minimum floor and aim to set a target for 100% of the capacity coming from LSE customer sites in a given aggregation. The CCAs prefer aggregations that consist of projects on their customer's sites. See Addendum #1 Question #28.

Generally, Proposers should refer to CAISO for confirmation of sub-lap boundaries, as CAISO is the entity that defines the sub-lap boundaries, requires PDR resources to be comprised of DERs from within a given sub-lap and is the entity that would be enforcing that. The LSEs do not have any additional information besides the map and zip-code reference available at from PGE (here).

As a municipal-owned utility, SVP is not assigned sublaps, however if this will be used for SLAP nodal pricing reference, these sublaps would be nearest in proximity to SVP. This assignment is subject to change at any time without prior notice. SVP is seeking DER RA within the City of Santa Clara's boundary.

13. The vagueness of the current resiliency requirements are problematic, because the RFP presents 'resilience' as a true/false condition. However, it is really more of a sliding scale, where a minimal amount of investment can create a minimal amount of resilience. This is especially difficult on the C&I side, where there are almost unlimited degrees of resilience for a project, from short-duration, critical load-only backup, to full off-grid microgrids. If all parts of the spectrum qualify as resilience, this box can be checked by adding minimal functionality to a given site, meaning that this requirement would greatly increase the complexity of the RFP, but without providing significant additional value for customers (essentially 'compliance resiliency').

In Addendum #1 Question #32 the LSEs indicated they may reserve the right to review the developer - customer agreement used by the Proposer during implementation. The LSEs will exercise this right in order to review and approve of the contractual obligation for back-up power/resilience that the Proposer includes in their customer contract for a given site. This will give LSEs the ability to ensure that the systems developed for RA are providing some level of meaningful resilience to the customer while enabling flexibility for systems to meet the needs of a given customer and avoiding the creation of a "resilience requirement" that creates an undue burden on developers in cases where a prescribed level of resilience might not meet a customers needs.

The robustness of the Proposer's approach to determining the resilience needs of customers and a requisite system will be considered by each LSE when scoring the Go-to-Market Strategy.

14. We understand that PCE has been rated investment grade. What credit documentation will the other LSEs have available?

SVP can provide an investment grade rating. SVCE and EBCE plan to pursue investment grade ratings.

15. If the winning bidder(s) commit to a certain capacity, and cannot deploy that capacity because of lack of customer interest or other factors, will that bidder be penalized?

Awardees must reach the capacity that they have contracted to supply. Specific penalties will be determined in the contracting process.

16. Beyond resilience, will host customers be expecting to receive any benefits from their systems (peak shaving, load shifting, etc) when those systems are not being called on to meet RA needs?

Yes. Proposers are expected to know how to optimize the interplay between customer benefits and system benefits. See Webinar Questions Addendum #30.

17. Will the sites need to be open for demonstration, STEM learning or other public access, during certain hours of operation?

There is no requirement to this effect.

18. Are the LSEs subject, in any way, to the upcoming California Consumer Privacy Act (CCPA), either as a covered business or a service provider? Have the LSEs implemented all necessary requirements for compliance with the CCPA, including individual rights processes and right to opt-out of data selling? Do the services covered under this RFP constitute "selling" as defined in the CCPA?

All LSEs in this program are public agencies, and therefore are not considered "businesses" or "service providers" subject to the CCPA.

It is expected that awardees in this program will be private businesses that will sell solar+storage solutions to customers. Those businesses and the services they provide may be subject to the CCPA.

19. Are there any restrictions (contractual or otherwise) on sharing customer personal information with service providers for marketing products or services? For example, would the LSE share the names, addresses, meter numbers, account numbers, and rates of their customers?

Additional data sharing beyond the anonymized data shared via the Data Room will be determined on an LSE specific basis during contracting and will likely only be shared after a contract, or other data sharing agreement, is executed. Each LSE may have specific contractual requirements and processes, such as requiring permission from the customer in question, to enable the sharing of customer specific data with Awardee(s).

As discussed in the RFP on p. 17, Awardee(s) are "expected to act as a customer marketing and program engagement partner" LSE in question. Proposers should use the Go-to-Market & Customer Engagement Strategy section of the RFP response to discuss what additional customer information would be useful and how it would be used in a successful Go-to-Market & Customer Engagement Strategy. The LSEs will work with Awardee(s) to execute their proposed strategies.

20. Do the LSE have email addresses for the customers?

Yes, however, see previous previous question and refer to Addendum #1 Question #14 for more information on the data sharing process.

21. Can the LSEs provide total residential accounts, number of existing NEM accounts, number of existing storage accounts, total small commercial customers (excluding industrial/large commercial), number of existing NEM accounts, number of existing storage accounts?

	EBCE	SVCE	PCE	SVP
Total Residential Accounts	501,424	243,251 SVCE SA IDs / 252,579 total SA IDs (as of 11/16/19)	261,000	47,007
	Rate Schedules: E1, E6, ETOU, EV rates	Rate schedules: E1, E6, ETOU, EV rates		
# of Existing Residential NEM Accounts	28,084	19,659 SVCE SA IDs / 20,941 total SA IDs (as of 11/16/19) Rate schedules: E1, E6, ETOU, EV rates	~12,331	~800
# of Existing Residential Storage Accounts	571 total interconnections++	449 total interconnections (as of 8/22/19)	~383	1
Total small commercial customers	44,340	20,596 SVCE SA IDs / 21,553 total SA IDs (as	21,860	6,168

	Rate schedules: A1, A6	of 11/16/19) Rate schedules: A1, A6		
# of existing small commercial NEM accounts	695	343 SVCE SA IDs / 390 total SA IDs (as of 11/16/19) Rate schedules: A1, A6	~320^	~90
# of existing small commercial storage accounts	571 total interconnections++	3 total interconnections (as of 8/22/19)* Rate schedules: A1, A6	~8	<5

*Please note that SVCE has additional storage deployed by commercial customers, but they are not "small commercial" customers, as was asked in the question, so they are not reflected in the numbers above.

++EBCE storage accounts are not broken down by customer class

^PCE calculated this number to include commercial/ag/industrial NEM with solar under 200kW.

22. As a Solar Structure Manufacturer/Supplier, we are looking to provide a proposals for our structure for all of the LSE's. Would we be required to partner with a Developer to provide a proposal that encompasses the entire scope of the project, or can we supply proposals that cover just the structural aspect of the outlined projects?

All proposals are required to include RA capacity.

23. At what stage does hard pricing need to be included in the bid? In our initial submission or during the subsequent negotiating stage?Can the bidders provide pricing that is varied or scaled based upon the quantity/capacity that is negotiated and finalized with the LSEs?

Firm pricing must be provided in the bid. All pricing will be finalized during contracting.

If a bidder expects pricing to change based on volume and chooses to submit multiple pricing offers based on volume increments, that is acceptable (within reason).

24. Can the bidders provide different pricing for each of the LSEs? If so, how would you evaluate/score that proposed pricing arrangement?

Pricing can vary for each LSE. Proposers should submit a different version of Attachment A: Pricing & Capacity Worksheet for each LSE that they are proposing too.

All LSEs will evaluate their Proposals independently. See Addendum #1 Question #4.

25. Regarding fire safety certification, should it be assumed that the P3 investor/owner

will be responsible for this (i.e., that the risk for this will be transferred to the P3)?

The LSEs are not defining the ownership/financing structure of the projects, which will be left to the Proposer to determine. The LSEs anticipate that the owner of the asset will assume the fire risk.

26. There are dependencies that will only be guaranteed in the future, e.g. actually being able to identify and contract community load counterparts, and the ability to reach the promised price point. Will there be off-ramps in the agreement, or does signing an agreement with an LBE require the delivery of RA at the agreed-upon price point, no matter what.

A signed contract requires the Awardee to deliver on the agreed upon RA capacity.

27. Our microgrids utilize solar PV, battery storage and dispatchable generation. We would be maximizing the use of the solar and batteries, with the natural gas only for the backup generation that, from our research, we believe is essential for C&I customers to adopt distributed energy solutions. I've attached a brochure to help give you more details about our system.

We would be using the batteries to bid the RA, and the natural gas would only be used for servicing the C&I customers. The natural gas would not be related to the services provided to the LSE.

So, what I'm still trying to figure out is, are the LSE's not going to seriously consider battery, solar, and dispatchable generation C&I microgrid proposals for this RFP, even if the natural gas will not participate in the RA bid?

As stated in the Addendum #1, the LSEs will NOT procure RA capacity from fossil fuel technology. A primary intent of this RFP is to catalyze the market for behind-the-meter energy storage in the LSEs' service territories.

However, the LSEs will consider an overall customer solution that includes natural gasbased technology if a natural gas based system is required to enable a sufficient level of resilience that creates value to the host customer, provided that the following is true:

-The natural gas based technology is not used to provide RA -The natural gas based technology still results in GHG emissions reductions. i.e. it is only activated rarely for back-up power purposes during long power outages

These requirements will be enforced via review of the developer - customer agreement by the LSE in guestion.

28. On page 12 of the RFP, bidders are instructed to submit a description of the Net Qualifying Capacity (NQC), QC, and EFC according to the CAISO methodologies. We understand that from CAISO that they haven't finalized an NQC calculation methodology for solar+storage hybrid (one meter) projects. If this is the case, would the Joint CCA team provide a standard methodology for all bidders to use so our proposals use the same basis calculations?

Current CPUC PD: http://docs.cpuc.ca.gov/PublishedDocs/Efile/G000/M320/K714/320714614.PDF

The LSEs recognize that the accepted methodologies are changing. There will be provisions to adapt to any regulatory changes throughout the legal and commercial contracting processes. It is likely that specific provisions to address possible regulatory change will also be included in the contract itself.



EBCE Customer Resilience Program

JP Ross June 17, 2020

Outline

- Background
- Proposed Program
- Solicitation
- Load Modification
- Board Request



PSPS and Resilience

- 2018 PSPS resulted in >50,000 EBCE accounts losing power at least once
- PG&E expects PSPS to continue for the next decade
- EBCE is working to deliver resilience to our customers through multiple program areas
 - Critical Facilities
 - Medical Baseline Resilience Program
 - EBCE Resilience Program for Residential and Commercial Customers
- This program will deliver resilience to thousands of customers while also procuring Load Modification at a competitive rate



Program Outline

- EBCE will partner with multiple vendors to deploy solar and storage on homes and businesses throughout EBCE service territory
- Systems will be able to island from the grid to provide resilience
- Battery capacity will be aggregated and sold to EBCE for Load Modification
- Vendor payments will be shared with customers to increase participation by providing an ongoing revenue stream for battery capacity
- A minimum of 20% of system capacity will be installed in low income/disadvantaged communities or CARE/FERA/Medical Baseline homes
- Vendors have agreed to workforce development requirements similar to OCEI
- As first major call to action EBCE is using this program to raise awareness with customers



Solicitation Summary

- Issued Nov, 19 2020 with 3 other LSEs; PCE, SVCE, PCE
- EBCE goal of delivering 10+MW of RA from behind the meter solar + storage
- Received 19 bids 12/23/19
- Interviewed 8 vendors
- Negotiating with four finalists

Category Weighting	Proposal Section(s)
A. Completeness of Response to RFP	Pass/Fail
B. Capacity Requirements & Deployment Timeline	Pass/Fail
C. Technology Safety Certifications & Standards	Pass/Fail
D. Islanding Capability	Pass/Fail
E. Pricing	40
F. Experience Developing Similar Projects & Providing RA Capacity	15
G. Experience & Qualifications of Personnel	10
H. Quality & Detail of Go-to-Market & Customer Engagement Plan	15
I. Proposed Fire Safety Measures	10
J. Local Hiring/Workforce Development Plan	10
TOTAL SCORE	100

Table 1: Scoring Criteria



Final Vendor Selection

- EBCE has selected 2 finalists
- Requesting Board Approve contract with Sunrun and provide Authority for CEO to negotiate and execute with EnelX for a total of 10.75MW of Load Modification
- Time is of the essence as EBCE works to launch the program in time to deliver solutions by Summer 2020





Load Modification

- Load Modification reduces EBCE Resource Adequacy obligation by reducing peak load
- EBCE will be able to change hours of dispatch over the term as load shape changes
- Load reduction happens every day
- Annual dispatch plan filed with the CEC





Action Item

- Approve the selection of two vendors in response to the Distributed Resource Adequacy (RA) Request for Proposals to deliver RA in the form of load reduction from behind-the-meter solar and storage installations delivering resilience to EBCE commercial and residential customers.
- Authorize the CEO or his designee to negotiate and execute a Load Modifying Agreement with Sunrun Inc. attached in Exhibit A, for Load Modification up to 5.75MW/23MWh through December 31, 2032.
- Authorize the CEO or his designee to negotiate and execute a Load Modifying Agreement with Enel X in substantially the same form as the Sunrun agreement attached in Exhibit A, for Load Modification up to 5MW/20MWh through December 31, 2032.



THANK YOU!

EBCE.org

- f /EastBayCommunityEnergy
- @PoweredbyEBCE



- customer-support@ebce.org
- 833.699.EBCE (3223)







Staff Report Item 16

то:	East Bay Community Energy Board of Directors
FROM:	Marie Fontenot; Senior Director of Power Resources
SUBJECT:	Update on 2020 Integrated Resource Plan (Informational Item)
DATE:	June 17, 2020

Recommendation

Receive the informational update on process and status of Integrated Resource Planning (IRP) analysis.

Background and Discussion

EBCE is currently engaged in its 2020 IRP study. This analysis is being undertaken with support from consultants from Siemens Energy Business Advisory and Ascend Analytics. Staff is undertaking analysis in two phases: phase one is focused on performing analysis and completing documentation as required to comply with the California Public Utility Commission's (CPUC) IRP filing requirements; phase two is focused on supplemental analysis that will inform a future Staff recommendation to the Board on a procurement strategy to reduce greenhouse gas (GHG) emissions as a greater rate than is required by the CPUC and State of California.

Results of phase one of the IRP Analysis will be presented to the Board in July. At that time, Staff will seek Board approval on the compliance filing as the filing must be completed by September 1, 2020. Staff will present the results of phase two and will request Board input and eventual approval of more aggressive GHG emissions reduction targets and associated procurement strategy at a later date.

In this June 17 Board meeting, Staff is providing an update to the Board and the public, for feedback and discussion as an informational item, on the analysis underway toward phase one, achieving compliance with the CPUC's IRP study and filing.

Fiscal Impact

This update has no fiscal impact on current operations.

Attachments

A. Presentation of initial results of phase one of IRP analysis



Integrated Resource Plan Update

PRESENTED BY: Marie Fontenot

DATE: June 17, 2020

Deliverables

Phase 1: CPUC IRP Compliance Filing

- Analysis based on prescriptive assumptions
- Narrative analysis, process, results, lessons learned
- Resource Data conforming & "preferred" portfolios
- Clean System Power Calculator

Phase 2: Establish EBCE Organizational Goals

- Additional analysis
- Identify reliability needs
- Define trade-offs between organizational objectives
- Inform procurement recommendations
- Develop path to expedited GHG reduction



Revised CPUC Requirements

• 46 MMT *and* 38 MMT scenarios

LSE	2030 Load (GWh)	Share of 2030 load in <u>IOU territory</u>	2030 GHG emissions benchmark – 46 MMT scenario	2030 GHG emissions benchmark – 38 MMT scenario
PG&E Bundled	26,777	35.2%	5.479	4.526
EBCE	6,910 ¹	9.08%	1.23 ²	.984 ²
SCE Bundled	54,393	63.49%	9.687	8.003
SDG&E Bundled	5,366	29.46	1.198	0.990

¹ Load represents CPUC approved load forecast as of 4/15/20; subject to revision in final IRP analyses and filing. ² Reflects requirement after behind the meter Combined Heat & Power emissions are removed from target.

- Specific Input Requirements
 - Filing date: September 1, 2020

Context / Comparison

2020 Load Forecast (GWh)	Estimated 2020 GHG emissions for 39.5% RPS ¹	Estimated 2030 GHG emissions for 60% RPS ²	2030 Load (GWh)	2030 GHG emissions benchmark – 46 MMT scenario	2030 GHG emissions benchmark – 38 MMT scenario
Total: 5,900	Total: n/a	Total: n/a		1.414	1.168
Bright Choice: 4,889	Bright Choice: ~970,000 MT + CHP	Bright Choice: ~537,000 MT + CHP ³	6,910	(or ~900,000MT 1.23MMT)	(or ~668,000MT .984 MMT)

¹ Assumption: use of April's Board-approved PCL methodology: PG&E hydro allocation and RPS+5%. 39.5% represents 5% above PG&E's 2019 renewable energy power content forecast (per 2019 Bundled RPS Energy Sale Solicitation Advice Letter filing). ² 60% is equal to RPS compliance for 2030.



³ Estimate is high-level approximate ONLY based on Index+REC transactions. Does not represent expected outcomes associated with long-term RPS contracts (e.g. impacts of curtailment)

Scenario Analysis Will Evaluate...

Key Evaluation Metrics	Scenario 1: 46 MMT / i.e. 1.414 MMT i.e. 1.23 MMT	Scenario 2: 38 MMT / i.e. 1.168 MMT i.e984 MMT	Scenario 3: EBCE aggressive 30 MMT i.e74 MMT
Carbon Free			
Affordability (Cost)			
Resource Mix (incl. New build vs existing)			
Risk Mgmt: Spot Market vs Short-Term vs Long-Term Contracts			
Reliability			



Developing Conforming Portfolios

CPUC compliance portfolios developed based on the CPUC's "Reference System Plan".

Benefits:

- Consistent with CPUC view of reliability
- Defensible: Tied to CPUC-expectations of resource availability (defensible)
- Able to incorporate EBCE-views of availability & portfolio-fit

Limitations:

- Not directly tied to EBCE organizational goals
- Final results & comparison of <u>the 3</u> <u>scenarios</u> will not be true "apples to apples"

	2020	2022	2026	2030
RSP CAISO Load (46 or 38 MMT)	205,907	204,065	205,132	206,953
EBCE Load	7,535	6,894	6,906	6,910
EBCE % of CAISO	3.66%	3.38%	3.37%	3.34%



Draft Conforming Portfolios

46 MMT Scenario: EBCE = 1.23 MMT in 2030

	% of	2022	max				
	Pro-	Adjust	allow				
	Rata	ment	ed	Overall	EBCE Pro	Rata Por	tfolio
Resource				2020	2022	2026	2030
2-hr Battery Storage	125.0%	100%		0.0	0.0	80.0	280.8
4-hr Battery Storage	125.0%	100%		0.0	175.6	339.8	225.8
Pumped Storage (long-duration)	75.0%	0%		0.0	0.0	0.0	64.4
Large Hydro	70.0%	100%	100	0.0	167.2	166.6	100.0
Imported Hydro	70.0%	100%		0.0	67.4	67.2	66.7
Biogas	50.0%	0%		0.0	0.0	0.0	4.6
Biomass	50.0%	0%		0.0	0.0	0.0	10.4
Geothermal	100.0%	10%		0.0	12.2	75.2	74.5
Small Hydro	100.0%	100%	20	0.0	32.9	32.8	20.0
Shed DR	50.0%	100%		0.0	40.8	40.7	40.4
Candidate Wind Resources	112.0%						
Southern_CA_Desert_Southern_NV_		100%		0.0	119.4	130.7	130.7
Sacramento_River_Wind		100%		0.0	57.5	57.5	57.5
Tehachapi_Wind		100%		0.0	119.4	130.7	130.7
Generic_CA_Wind		100%		0.0	0.0	0.0	23.3
New_Mexico_Wind		100%		0.0	59.7	65.4	65.4
Candidate Solar Resources	112.5%						
Southern_PGE_Solar		100%		0.0	168.0	493.0	493.0
Southern_CA_Desert_Southern_NV_		100%		0.0	186.9	186.9	186.9
Tehachapi_Solar		100%		0.0	186.9	186.9	186.9
Generic_CA_Solar		100%		0.0	0.0	0.0	106.2

	% of	2022	max				
	Pro-	Adjust	allow				
	Rata	ment	ed	Overall	EBCE Pro	Rata Po	rtfolio
Resource				2020	2022	2026	2030
2-hr Battery Storage	125%	100%		0.0	0.0	80.0	224.0
4-hr Battery Storage	125%	100%		0.0	175.6	293.9	317.7
Pumped Storage (long-duration)	75%	0%		0.0	0.0	0.0	80.2
Large Hydro	74%	100%	100	0.0	176.8	176.1	100.0
Imported Hydro	74%	100%		0.0	71.3	71.1	70.5
Biogas	50%	0%		0.0	0.0	0.0	4.6
Biomass	50%	0%		0.0	0.0	0.0	10.4
Geothermal	100%	0%		0.0	0.0	78.3	77.7
Small Hydro	100%	100%	20	0.0	32.9	32.8	20.0
Shed DR	50%	100%		0.0	40.8	40.7	40.4
Candidate Wind Resources	115.0%						
Southern_CA_Desert_Southern_NV_		90%		0.0	124.5	151.5	151.5
Sacramento_River_Wind		100%		0.0	57.5	57.5	57.5
Tehachapi_Wind		90%		0.0	124.5	151.5	151.5
Generic_CA_Wind		90%		0.0	0.0	0.0	168.0
New_Mexico_Wind		90%		0.0	62.2	75.7	75.7
Candidate Solar Resources	113.7%						
Southern_PGE_Solar		100%		0.0	168.0	493.0	493.0
Southern_CA_Desert_Southern_NV_		100%		0.0	204.6	204.6	204.6
Tehachapi_Solar		100%		0.0	204.6	204.6	204.6
Generic_CA_Solar		100%		0.0	0.0	0.0	118.3



Draft Conforming Portfolios

46 MMT Scenario: EBCE = 1.23 MMT in 2030





Initial Costs of Conforming Portfolios

Values represented on this slide are results of early, DRAFT model runs. Information is displayed to demonstrate the **types** of results we will provide. These values do <u>not</u> represent indicative estimates.

46 MMT Scenario: EBCE = 1.23 MMT in 2030







Portfolio Reliability

Preliminary results only

46 MMT Scenario: EBCE = 1.23 MMT in 2030






Risk Mgmt: Transaction Tenors

46 MMT Scenario: EBCE = 1.23 MMT in 2030

38 MMT Scenario: EBCE = .984 MMT in 2030

Tenor	% or MWh	Tenor	% or MWh
Spot Market		Spot Market	
Short Term		Short Term	
Long Term		Long Term	

Preliminary results pending



Reliability of Portfolios

46 MMT Scenario: EBCE = 1.23 MMT in 2030

38 MMT Scenario: EBCE = .984 MMT in 2030

Susceptibility/Robustness in terms of:	Susceptibility/Robustness in terms of:
Weather Volatility in Power Prices	Weather Volatility in Power Prices
Renewable Intermittency	Renewable Intermittency

Preliminary results pending





Staff Report Item 17

TO: East Bay Community Energy Board of Supervisors

FROM: Nick Chaset, CEO

SUBJECT: EBCE Budget Fiscal Year 2020-2021 (Action Item)

DATE: June 17, 2020

Recommendation

- A. Approve one of the presented budget scenarios for fiscal year 2020-2021
- B. If Non-COVID scenario selected, modify the Bright Choice Value Proposition to a 1.0% discount as compared to PG&E Rates; otherwise modify the Bright Choice Value Proposition to a 0.5% discount as compared to PG&E rates. Either effective July 1, 2020.
- C. Make no change to Brilliant 100 value proposition (i.e. maintain parity with PG&E) for balance of 2020
- D. Close Brilliant 100 to new accounts and opt-ups, effective July 1, 2020
- E. Present 2021 Brilliant 100 options to Board in September for Board action in Q4 2020
- F. Set a renewables procurement floor at the current year RPS, while maintaining 100% carbon-free content for Brilliant 100 for fiscal year 2020-2021

Background and Discussion

EBCE's fiscal year is from July 1 through June 30. Staff is presenting value propositions and proposed budget scenarios for fiscal year 2020-2021. One scenario assumes no material impact on load from COVID-19, and the other assumes a moderate reduction in load. The value propositions align with the load scenarios to reach a balanced budget.

The presented value proposition and budget is based on feedback from the Finance, Administrative, and Procurement Committee as well as the May 20, 2020 Board meeting. These budget scenarios outline expected costs and revenues anticipated for the next 12 months under COVID influenced, and Non-COVID based, load and PCIA changes.

Fiscal Impact

This establishes the forecast of EBCE's fiscal position for the next 12 months.

Attachments

- A. Summary of EBCE Budget FY 2020-2021
- B. Summary of Public Comment of EBCE Value Proposition
- C. Presentation of EBCE Budget, Value Proposition and B100 Power Content for FY 2020-2021

EBCE DRAFT BUDGET SCENARIOS FOR JULY 1, 2020 – JUNE 30, 2021

	FY 2020-21 Non-COVID	FY 2020-21 COVID	Scenario Difference
REVENUE AND OTHER SOURCES			
Electricity Sales	388,125,000	382,758,000	(5,367,000)
Investment Income	1,231,000	1,231,000	-
Other Income	275,000	275,000	-
Uncollectables	(9,703,000)	(9,568,000)	135,000
Total Revenue and Other Sources	379,928,000	374,696,000	(5,232,000)
EXPENSES AND OTHER USES			
ENERGY OPERATIONS			
Cost of Energy	347,594,000	342,796,000	(4,798,000)
Data Management/Customer Service	7,704,000	7,704,000	-
PG&E Service Fees (Billing/Metering)	2,396,000	2,396,000	
Scheduling	660,000	660,000	-
Total Energy Operating Costs	358,354,000	353,556,000	(4,798,000)
OVERHEAD OPERATIONS			
Personnel	7,429,000	7,429,000	-
Marketing, Outreach, Communications	1,214,000	1,214,000	-
Legal, Policy, & Reglatory Affairs	1,297,000	1,297,000	-
Other Professional Services	1,345,000	1,345,000	
General & Administrative	2,146,000	2,146,000	-
Local Development Funding	6,615,000	6,615,000	
Depreciation	60,000	60,000	-
Total Overhead Operating Costs	20,106,000	20,106,000	-
INTEREST PAYMENTS			
Borrowing Interest	804,000	804,000	-
Total Interest Payments	804,000	804,000	-
TOTAL EXPENSES & INTEREST DUE	379,264,000	374,466,000	(4,798,000)
NET INCREASE (DECREASE) IN POSITION	664,000	230,000	(434,000)

NOTE: The above budget statement does not include financials related to new communities



CURRENT YEAR BUDGET AND DRAFT SCENARIOS

	FY 2019-20 BUDGET	FY 2020-21 Non-COVID	FY 2020-21 COVID
REVENUE AND OTHER SOURCES			
Electricity Sales	487,014,000	388,125,000	382,758,000
Investment Income	578,000	1,231,000	1,231,000
Other Income	-	275,000	275,000
Uncollectables	(2,447,000)	(9,703,000)	(9,568,000)
Total Revenue and Other Sources	485,145,000	379,928,000	374,696,000
EXPENSES AND OTHER USES	_		
ENERGY OPERATIONS			
Cost of Energy	386,905,000	347,594,000	342,796,000
Data Management/Customer Service	6,758,000	7,704,000	7,704,000
PG&E Service Fees (Billing/Metering)	2,253,000	2,396,000	2,396,000
Scheduling	653,000	660,000	660,000
Total Energy Operating Costs	396,569,000	358,354,000	353,556,000
OVERHEAD OPERATIONS			
Personnel	6,703,000	7,429,000	7,429,000
Marketing, Outreach, Communications	2,263,000	1,214,000	1,214,000
Legal, Policy, & Reglatory Affairs	1,586,000	1,297,000	1,297,000
Other Professional Services	1,214,000	1,345,000	1,345,000
General & Administrative	2,290,000	2,146,000	2,146,000
Local Development Funding	6,340,000	6,615,000	6,615,000
Depreciation	61,000	60,000	60,000
Total Overhead Operating Costs	20,457,000	20,106,000	20,106,000
INTEREST PAYMENTS			
Borrowing Interest	1,229,000	804,000	804,000
Total Interest Payments	1,229,000	804,000	804,000
TOTAL EXPENSES & INTEREST DUE	418,255,000	379,264,000	374,466,000
NET INCREASE (DECREASE) IN POSITION	66,890,000	664,000	230,000



UPDATES TO BASE CASE

- Non-COVID base case adjustments
 - Bright Choice at 1.0% discount to PG&E
 - No change to Brilliant 100 through December, 2020 then set to cost of service at 2.5% premium, with review and evaluation of options with the Board through the fall
 - o 90% migration of current Brilliant 100 customers to Bright Choice in January 2021
 - o Change power content to match RPS for Brilliant 100, but remain 100% carbon free
 - o Migration causes \$400k loss of revenue, and uptick in renewables by \$251k
 - Difference from previous draft in energy costs is also due to beneficial changes in renewable contracts independent of budget process and capture of basis savings with highly hedged hours
 - o Energy cost is further reduced by \$1.5M with sale of carbon-free resources
 - o Marketing is reduced by \$300k
- COVID base case adjustments
 - Load was reduced to reflect a moderate COVID scenario. Volumetric load is estimated down 5-6% in the near-term months with a peak demand reduction of 10-15% with a gradual reversion to about 1-3% volumetric load reduction and 4-6% peak reduction through the 2020 calendar year, which is returned to base line starting in 2021.
 - o Bright Choice at 0.5% discount to PG&E
 - o Non-COVID Brilliant 100 conditions are applied to this scenario as well
 - Revenue is reduced by about \$5.4M
 - o Cost of energy is reduced by about \$4.8M from reduced COVID load
 - Energy cost is further reduced by \$1.5M with sale of carbon-free resources
 - All noted overhead conditions from Non-COVID scenario apply

Additional conditions for both scenarios with the Brilliant 100 product:

- Brilliant 100 is maintained at least through December to evaluate how to appropriately manage the product, but should be locked to new accounts and opt-ups until a decision can be made
- Some options to evaluate this fall for Brilliant 100 product:
 - o Maintain product at a rate below cost of service for existing customers
 - o Discontinue product at some specified time and allow migration to other products
 - Set product to cost of service, currently at 2.5% premium to PG&E
 - If market conditions become favorable, continue Brilliant 100 as is or with some appropriate minor adjustments



QUARTERLY DISTRIBUTION

FY 2020-21 BUDGET Non-COVID	Q1	Q2	Q3	Q4
REVENUE AND OTHER SOURCES	121,785,000	97,813,000	75,065,000	85,265,000
ENERGY OPERATIONS OVERHEAD OPERATIONS	101,302,000 5,228,000	93,475,000 5,228,000	89,755,000 5,228,000	73,822,000 5,228,000
NET INCREASE (DECREASE) IN POSITION	15,255,000	(890,000)	(19,918,000)	6,215,000
FY 2020-21 BUDGET COVID	Q1	Q2	Q3	Q4

FY 2020-21 BUDGE1 COVID	ŲI	Q2	Q3	Q4
REVENUE AND OTHER SOURCES	115,277,000	97,148,000	76,262,000	86,011,000
ENERGY OPERATIONS OVERHEAD OPERATIONS	99,945,000 5,228,000	92,224,000 5,228,000	88,554,000 5,228,000	72,833,000 5,228,000
NET INCREASE (DECREASE) IN POSITION	10,104,000	(304,000)	(17,520,000)	7,950,000

In both scenarios it can be seen:

- Q1 results are generally the strongest quarter due to summer rates. Q1 includes the largest COVID load reductions
- Q2 results are flat due to a transition to winter rates
- Q3 results are negative due to winter rates and an increase in PCIA to reflect hitting the PCIA undercollection trigger
- Q4 results are positive due to a transition to summer rates and low spring energy costs

NEW COMMUNITIES DISCUSSION

	FY 2020-21 Non-COVID	FY 2020-21 COVID	Scenario Difference
REVENUE AND OTHER SOURCES			
Electricity Sales	13,280,000	13,369,000	89,000
Uncollectables	(199,000)	(201,000)	(2,000)
Total Revenue and Other Sources	13,081,000	13,168,000	87,000
EXPENSES AND OTHER USES			
ENERGY OPERATIONS			
Cost of Energy	9,470,000	9,470,000	-
Data Management/Customer Service	299,000	299,000	-
PG&E Service Fees (Billing/Metering)	87,000	87,000	-
Total Energy Operating Costs	9,856,000	9,856,000	-
OVERHEAD OPERATIONS			
Marketing, Outreach, Communications	330,000	330,000	-
Total Overhead Operating Costs	330,000	330,000	-
TOTAL ENERGY & OPERATING EXPENSES	10,186,000	10,186,000	-
NET INCREASE (DECREASE) IN POSITION	2,895,000	2,982,000	87,000

- COVID scenario expects slightly higher revenues because Bright Choice is at 0.5% discount to PG&E, whereas Non-COVID Bright Choice is at 1.0% discount to PG&E
- Because new communities are expected to join after the new year, there is very little difference between COVID and Non-COVID scenarios in terms of costs
- Inclusion of new communities is expected to have a positive lift on margins based on an April enrollment timeframe estimated at about \$3MM in net margin based on current market conditions
- Positive margin is driven by a modest increase in operating overhead related to serving these customers and softer energy prices
- Revenues are based on:
 - o 5% opt out rate
 - o All new accounts are Bright Choice in both scenarios
- Additional analysis is underway to determine the optimal timing based on different customer rates and procurement risks/opportunities



REVENUE DISCUSSION

	FY 2020-21	FY 2020-21	Scenario
	Non-COVID	COVID	Difference
REVENUE AND OTHER SOURCES			
Electricity Sales	388,125,000	382,758,000	(5,367,000)
Investment Income	1,231,000	1,231,000	-
Other Income	275,000	275,000	-
Uncollectables	(9,703,000)	(9,568,000)	135,000
Total Revenue and Other Sources	379,928,000	374,696,000	(5,232,000)

- The previous draft had Bright Choice at 1% discount to PG&E with Brilliant 100 at a 3% premium
- This draft assumes no changes to Brilliant 100 in both scenarios through the remainder of 2020, then adjusted to match the cost of service at 2.5% premium in January 2021
- Bright Choice at 1.0% discount to PG&E in the Non-COVID scenario, and at 0.5% discount in the COVID scenario
- Another key difference from the previous draft is a 90% migration of customers from Brilliant 100 to Bright Choice is assumed in January 2021 in both scenarios
 - o This change incorporates the additional risk from migrating customers
- This draft includes anticipated changes in load based on a moderate COVID scenario expectation
 - Reduction of \$5.4M in revenues due to load changes from COVID, primarily occurring in the summer months of 2020, and Bright Choice rate differences
 - o The load reductions are assumed to be 1.7% in total through the fiscal year
 - o Assumes a scenario of reduced load through the 2020 calendar year
 - The hardest hit months are in the first quarter at -5% with eventual tapering through December. Load reverts to a status quo/normal condition in Jan.
- Investment Income is based on projected bank account balances in interest bearing accounts (reserves and ICS) at a slightly lower interest rate of 1.5% than current rates (1.79%)
- Uncollectables is increased from 0.5% of sales to 2.5% in anticipation of COVID induced recessionary impacts in both scenarios
- Current budget does not reflect the inclusion of the new communities of Tracy, Newark, and Pleasanton. New Communities are separated in order to isolate the economic impacts, which will differ based on exact enrollment timing, and are previously discussed



ENERGY OPERATIONS DISCUSSION

	FY 2020-21 Non-COVID	FY 2020-21 COVID	Scenario Difference
ENERGY OPERATIONS			
Cost of Energy	347,594,000	342,796,000	(4,798,000)
Data Management/Customer Service	7,704,000	7,704,000	-
PG&E Service Fees (Billing/Metering)	2,396,000	2,396,000	-
Scheduling	660,000	660,000	-
Total Energy Operating Costs	358,354,000	353,556,000	(4,798,000)

- Decrease in cost of energy from the previous draft and the Non-COVID scenario are from:
 - Reduction in renewable contract position costs of about \$400k
 - Increase in renewable attributes due to migration in 2021 of about \$251k
 - Capture of basis costs and market purchases in highly hedged hours of about \$760k
 - o \$1.5M from sale of carbon-free assets
- Difference in energy costs between the two scenarios are due to the following impacts from the COVID reduced load:
 - o \$4.0M reduction in physical energy cost and basis costs
 - \$640k reduction in renewable attributes due to load reduction
 - o \$140k reduction in CAISO costs
 - \$1.5M from sale of carbon-free assets
- The two scenarios are not expected to have any impact on energy related service costs provided by SMUD for data management and customer service, PG&E for billing and metering, and NCPA for scheduling



OVERHEAD OPERATIONS DISCUSSION

	FY 2019-20 BUDGET	FY 2020-21 BUDGET	Fiscal Year Difference
OVERHEAD OPERATIONS			
Personnel	6,703,000	7,429,000	726,000
Marketing, Outreach, Communications	2,263,000	1,214,000	(1,049,000)
Legal, Policy, & Reglatory Affairs	1,586,000	1,297,000	(289,000)
Other Professional Services	1,214,000	1,345,000	131,000
General & Administrative	2,290,000	2,146,000	(144,000)
Local Development Funding	6,340,000	6,615,000	275,000
Depreciation	61,000	60,000	(1,000)
Total Overhead Operating Costs	20,457,000	20,106,000	(351,000)

Because there is no anticipated impact on overhead operations between COVID and Non-COVID scenarios, this discussion is related to the current year budget but some changes to the previous draft were made:

- Marketing was reduced from the previous draft budget by an additional \$300k in the areas of community grants and noticing
- Local Development was reduced by \$230k to be on parity with the current year, with \$275k added for COVID Outreach from a grant provided by EDPR. This can be seen in more detail on the next page
- Personnel costs increase due to:
 - Staff salaries covering the full fiscal year (FY 2019-2020 budget assumed partial year hires). Headcount maintained at 37FTE, flat to the 2019-2020 approved headcount
 - o Enact a policy to allow staff to monetize PTO
 - 2.5% merit-based adjustments (\$118,000)
 - o 3% discretionary promotions/retention-based compensation adjustments (\$169,000)
- Marketing, Outreach, Communications costs are reduced due to:
 - o Reduced mailing requirements and availability of lower cost notification processes
 - Reduced advertising and sponsorship based on cost cutting measures
 - New Community costs are not included in this estimate
 - Internal call center development is not included in the current budget, assuming a deferral of that investment to FY 2021-2022
- Legal, Policy, & Regulatory Affairs: Reduction in policy related costs based on cost cutting measures
- Other Professional Services: Increase in costs due to development of the CRM platform, which is primarily utilized for effective local program customer outreach
- General & Administrative: Reduced based on cost cutting measures
- Local Development Funding: Maintaining local development budget with current year's budget. Additional funds of about \$1.7M from expected grants and remaining funds rolled over from the 2019-2020 fiscal year to yield approximately \$8M in total spend.

LOCAL DEVELOPMENT DISCUSSION

LOCAL DEVELOPMENT FUND	FY 2019-20 BUDGET	FY 2020-21 BUDGET	Fiscal Year Difference
Program Funding			
Beginning Balance	-	1,398,000	1,398,000
Funding from General Fund Revenues	6,340,000	6,340,000	_
Grants/Credits	1,150,000	275,000	(875,000)
Mid-Year adjustment	500,000	-	(500,000)
Total Available Funding	7,990,000	8,013,000	23,000
Program Areas			
Demand Response	195,000	100,000	(95,000)
Energy Efficiency	100,000	740,000	640,000
Building Electrification	782,000	950,000	168,000
Vehicle Electrification	1,750,000	4,930,000	3,180,000
Collaboartive Procurement	810,000	958,000	148,000
Community Investment	250,000	-	(250,000)
Sponsorships/Events	65,000	-	(65,000)
Capital Set Aside	1,200,000	-	(1,200,000)
COVID-19 Relief Fund	1,440,000	335,000	(1,105,000)
Total Program Expenses	6,592,000	8,013,000	1,421,000
Ending Balance	1,398,000	-	

Similar to overhead, the Local Development Fund is not expected to be impacted by the two scenarios being evaluated. As such, this discussion focuses on the between year changes

- In addition to the Local Development Programs related spending shown above, there is approximately an additional \$4.7MM budgeted for locally generated utility-scale wholesale energy from projects such as EBCE's Altamont wind project, which is expected to come online during the fiscal year.
- Figures above do not reflect any local spending based on General & Administrative expenses



INTEREST PAYMENT DISCUSSION

	FY 2019-20 BUDGET	FY 2020-21 BUDGET	Fiscal Year Difference
INTEREST PAYMENTS			
Borrowing Interest	1,229,000	804,000	(425,000)
Total Interest Payments	1,229,000	804,000	(425,000)

Similar to overhead and Local Development, interest payments are not expected to be impacted by the two scenarios being evaluated. As such, this discussion focuses on the between year changes

Borrowing Interest accounts for commitment fees on undrawn debt and issuance fees for letters of credit. Assumes no outstanding debt.



RISKS DISCUSSION

EBCE is exposed to a variety of risks in this coming fiscal year. Below is a representative list of key risks, though it is not comprehensive in reflecting all risks.

Load Risk:

COVID impacts on load, collections, and energy market prices are being closely monitored and to date the Shelter in Place policy has been in effect for approximately two months. While this has provided meaningful data it is very difficult to extrapolate the insights to accurately forecast impacts on the next fiscal year.

Staff projects four potential scenarios for fiscal year 2020-2021:

- 1) **Moderate COVID load decrease:** Volumetric load is down 5-6% in the near-term months with a peak reduction of 10-15% and there is a gradual reversion to about 1-3% volumetric load reduction and 4-6% peak reduction, which is returned to base line starting in 2021.
- 2) Status quo: Quick reversion to a pre-COVID load. We anticipate continued county/city-based policies that will adjust work and personal interactions to account for public health and safety in light of COVID. With that said there is a scenario where electricity usage based on businesses partial reopening and moderately increased residential use that results in load that closely reflects EBCE load pre-COVID. This is the basis for the current budget.
- 3) Moderate COVID load increase: With the partial reopening of small businesses and large commercial space, we could see commercial load revert close to normal levels and residential load remain high based on partial families working from home. This could result in a moderate increase in volumetric load of 2-4% and a peak load similar to normal levels
- 4) Significant and sustained macro recession where volumetric load continues to be depressed. Volumetric load is down 6-10% and peak load is down 10-15% for next 12-24 months
- The base case COVID budget scenario presented assumes a moderate COVID load decrease. We believe this to be reasonable but somewhat conservative as we see some of the county and state level restrictions being lifted. However, we feel this approach is prudent in light of the uncertain trajectory of the COVID impacts. If load reverts to the status quo scenario net margin could increase between \$1-3MM.
- Changes in volumetric load due to ongoing COVID conditions are expected to reduce revenue and margin, though there is a decrease in energy costs that would help mitigate that impact. To date, EBCE has generally experienced a positive impact to load shape that has reduced the more expensive peak hours that help to mute this margin impact.
- Load changes can have material implications to EBCE as it relates to planning, procurement, and scheduling. Less certainty when it comes to load forecasting can increase risk. Generally speaking, an overall industry reduction in load directionally leads to softer energy prices. However, the uncertainty also leads to more significant day to day volatility, which can increase or decrease energy costs where EBCE is not hedged.



Energy Market Risk:

 EBCE currently has an estimated unhedged energy portfolio of between \$60-\$80 million (depending upon scenario) which is exposed to the market. Short term energy prices can see volatility of 10-20% on a regular basis based on typical supply/demand dynamics that are heavily influenced by weather and electricity infrastructure. The volatility may be exacerbated by COVID due to general uncertainty of load and the impacts from the macro oil industry on gas prices. We estimate that energy costs could increase reasonably as much as \$16-20MM, though this could also result in a similar reduction in energy costs. EBCE could consider hedging more load to reduce this risk. However, in an environment of softer energy prices, there are follow-on repercussions of an increasing PCIA the following year.

Uncollectables/Write-offs:

While EBCE has increased our expected exposure from 0.5% to 2.5% to uncollectable accounts, the current data is limited in measuring the extent of this impact. This forecast of 2.5% is at the higher end of the range compared to other CCAs based on our regional income demographics. In researching other load serving entity uncollectables through prior recessions, we have seen data ranging from non-material impacts up to 6% write-offs. Based on the recessionary impacts of COVID and suspension of disconnects, we do anticipate an increase in uncollectables that would lead to write-offs. There is a potential for the uncollectables to exceed our current forecast based on a more severe and extended recession. To mitigate this risk, we have launched an active campaign to increase CARE enrollments as a way to lower customers' bills and are continuing to create a plan to enroll customers onto longer term payment plans as necessary. While enrollment in payment plans should reduce write-offs, it would reduce near term cash inflows. EBCE has a strong cash and reserve position to support payments deferrals.

Opt-out risk:

Customer opt-outs are generally an ongoing risk to EBCE. However, we do not anticipate an
increased risk of opt-outs for the fiscal year, depending on adjustments to the value
propositions. There is currently no active discussion of DA expansion that would take effect in
2021 beyond what has already been in process.

Regulatory Risk:

- The current budget assumes a deferral of approximately \$16MM in costs, which is based on maintaining an undercollection (PUBA) of 7%. The most significant regulatory risk to revenues in the near term is the management of this PUBA cap. The CPUC could rule to fully collect on the undercollection, which would decrease fiscal year revenues or potentially elect for a larger undercollection, which could increase revenues.
- AB110 Power Content reporting is anticipated to be finalized in mid 2020. These regulations will include potential updates to the reporting treatment of PCC3 unbundled RECs and Asset Controlling Supply (ACS) procurement. Currently procurement of PCC3 RECs is limited and serves as a flexible tool to balance the portfolio. ACS is also currently procured largely as a source of clean large hydro energy, a cost-effective energy hedging tool, and a source of import RA from the pacific northwest. ACS sources are generally made up of 85-95% large hydro resources but may also include up to 10% source specified nuclear or natural gas. The new reporting regulation will likely require disclosure of all underlying energy as source specified, which could limit future procurements of this product depending on board direction.



• PCIA Working Group 3, which is focused on the disposition of PG&E's current long portfolio may benefit EBCE's energy procurement related costs. Though those benefits are not expected to be material for the fiscal year.

RISK MITIGATION

Considering the previous discussion, EBCE has the following options to further mitigate these economic risks.

- Additional changes to EBCE's value proposition:
 - Further reduction to Bright Choice discount to 0.0% = \$3.2MM
 - Increase to Brilliant 100 premium is expected to have negligible effect with 90% migration to Bright Choice
- EBCE has the opportunity to accept and resell PG&E's nuclear allocation. This could generate up to \$2 million based on market demand, which is somewhat unknown. This could present some risk to EBCE's power content label, but we can likely structure an arrangement to largely mitigate this risk.



Public Comments Summary Report

PRESENTED BY: MAS Team DATE: JUNE 17, 2020

2020 PROCESS

Webinars:

Tuesday, June 2 at noon Wednesday, June 3 at 6 PM

Audio Only:

Friday, June 5 at noon

Written Comments:

Opened on May 21 Closed at 11 pm on Sunday, June 7

Outreach:

Advertised via social media and the EBCE website starting on Friday, May 22

- Facebook 22,334 views
- Twitter over 34,000 impressions

Email notification on Tuesday, May 26

sent to over 2,500 subscribers with a 60 % open rate

Participation:

Webinar attendance: 15 Verbal comments: 6 Written comments: 7

SUMMARY OF COMMENTS

The most commonly stated value proposition and budget recommendations:

- Need for EBCE to be mindful of the **economic recession** caused by the COVID-19 pandemic and make all attempts to keep rates as low as possible to reduce the energy burden on our community members;
- Do not reduce the Local Development Business Plan (LDBP) budget;
- The Board should consider using the **reserves** to ease the shortfall caused by the Power Charge Indifference Adjustment (PCIA), and other budgetary risks;
- Consider **cost-cutting measures regarding staff**, such as freezing salaries and hiring, and allow for voluntary staff salary reductions; and,
- EBCE and CalCCA should recommend to the California Public Utilities Commission (CPUC) that the **PCIA** be reduced or removed.



ADDITIONAL COMMENTS

- Maintain a product that is priced lower than PG&E's base rate;
- Consider the required timeline for City Councils to update their default product, should Brilliant 100 become more expensive than PG&E. Do not implement this change immediately;
- Set the Renewable 100 rate so that it is consistently lower than PG&E's Solar Choice product; and,
- Set the Renewable 100 rate to the market rate for renewable energy.



REQUEST FOR BILL EXAMPLES

			Total EBCE Charges						
Bright Choice		1.5% discount	1.0% discount	0.5% discount	Proposed (1.0%) Bright Choice versus Current Bright Choice		Alternate (0.5%) Bright Choice versus Current Bright Choice		
Customer Class	Monthly Energy Use	Current Bright Choice	Proposed Bright Choice	Alternate Bright Choice	Monthly Difference	Yearly Difference	Monthly Difference	Yearly Difference	
Residential (E1)	380 kWh	\$ 43.99	\$ 44.21	\$ 44.43	\$ 0.22	\$ 2.64	\$ 0.44	\$ 5.28	
Small Commercial, Winter (A1X)	2,270 kWh	\$254.55	\$255.85	\$257.14	\$1.30	\$15.60	\$2.59	\$31.08	
Large Commercial, Summer (E19S)	1,014 kW / 294,710 kWh	\$35 232 18	\$35,411.02	\$35,589.86	\$178.84	\$2,146.08	\$357.68	\$4,292.16	



REQUEST FOR BILL EXAMPLES

					Total EB	CE Charges			
Brilliant 100		Parity with PG&E	PG&E + 3%	PG&E + 2.5%	Proposed (3%) Brilliant 100 versus Current Brilliant 100		Alternate (2.5%) Brilliant 100 versus Current Brilliant 100		PG&E + \$0.01
Customer Class	Monthly Energy Use	Current Brilliant 100	Proposed Brilliant 100	Alternate Brilliant 100	Monthly Difference	Yearly Difference	Monthly Difference	Yearly Difference	Renewable 100
Residential (E1)	380 kWh	\$44.66	\$46.00	\$45.77	\$1.34	\$16.08	\$1.11	\$13.32	\$48.46
Small Commercial, Winter (A1X)	2,270 kWh	\$258.43	\$266.18	\$264.89	\$7.75	\$93.00	\$6.46	\$77.52	\$281. <mark>1</mark> 3
Large Commercial, Summer (E19S)	1,014 kW / 294,710 kWh		\$36,841.77	\$36,662.92	\$1,073.06	\$12,876.72	\$894.21	\$10,730.52	\$38,715.81



QUESTIONS FROM THE PUBLIC Local Development Business Plan

- If staff and the Board eliminate \$2.7 million from the LDBP, would that leave only the electric vehicle (EV) program?
 - Staff Response: A reduction in LDBP/Programs budget will be applied based on direction from the Board.
- Why is 70% of the LDBP budget allocated to the EV program? EVs are not available to everyone?
 - Staff Response: EBCE's Board of Directors unanimously approved participation in the CEC's 2021 California Electric Vehicle Incentive Program (CALeVIP) EVIP) to leverage state incentive funding to enable Alameda County to meet it's share of the state target of 1.5M EVs on the road by 2025. Investment from CALeVIP is for EV charging infrastructure, not vehicles. Additionally, single family homeowners are not eligible for CALeVIP incentives for charging infrastructure.
- What percentage of the EV program will go towards fleet, municipal, and light/heavy duty trucks instead of personal vehicles?
 - Staff Response: The Local Development Budget allocations are stated in the May Board Packet. Municipal fleet electrification support, and our partnership with CALSTART are explicitly focused on light, medium and heavy-duty vehicle electrification and infrastructure deployment. Incentives through CALeVIP will support charging infrastructure for fleet managers. The exact proportion of CALeVIP incentive allocation is impossible estimate as applications are submitted on a first come first serve basis per the requirements of the CEC.



Power Procurement

- In the proposed budget for Fiscal Year (FY) 2020-21, the cost of electricity is shown as \$55.28/MWh. How was that amount calculated? How does this compare to historic pricing?
 - Staff Response: The amount is calculated based on actual and projected costs divided by projected load. This includes all energy related costs. This is a reduction in costs compared to FY 2019-2020, which is projected at approximately \$60/MWh.
- How much of EBCE's 2020-2021 power is already procured and how much remains open?
 - Staff Response: EBCE procures based on its Risk Management Policy as approved by the Board in January 2018. Hedge levels have been maintained in compliance to the established hedge matrix as detailed in the Energy Risk Management Regulations Appendix 9. Currently EBCE has hedged approximately 2/3 of the energy costs for the fiscal year.
- What does it mean that power costs are "soft?"
 - Staff Response: "Soft" power costs refers to low power costs in the market.



Power Procurement (cont'd)

- What is the average price of each type of energy fossil fuel, carbon-free, and renewable?
 - Staff Response: This table reflects the CPUC's estimates for the market price of energy (CAISO market prices), Resource Adequacy and the "RPS Adder" for 2020. The RPS Adder can be thought of as the REC value (Source: PCIA Market Price Benchmarks Forecast for 2020)
 - Additional pricing information can be found in the "Q4 2019 Report on Market Issues and <u>Performance</u>" published by CAISO on February 28, 2020 and the "2020 Padilla <u>Report</u>" published by the CPUC in May 2020

2020 Forecast Adders								
		PG&E	SDG&E	SCE				
Energy	On Peak	\$36.88	\$37.61	\$37.61				
Index ² (\$/MWh)	Off Peak	\$29.93	\$29.56	\$29.56				
RA Adder (\$/kW-month)	System RA	\$4.59						
	Local RA	\$4.11	\$4.15	\$4.30				
	Flexible RA	\$4.41						
RPS Adder (\$/MWh)			\$17.35					



Power Procurement (cont'd)

- What are the cost details for executed power contracts?
 - Staff Response: This is sensitive market information as defined by Public Utilities Code Section 454.5.
- What is the relationship between procurement costs and proposed rates for EBCE's products? Is there a direct relationship or are rates calculated using some other methodology?
 - Staff Response: There is a direct relationship between procurement costs and the value proposition. EBCE's rates are also based on PG&E's rate structure. The discount is proposed based on maximum savings while still competing with PG&E rates.



Budget/Operations

- Staff has provided a worst-case scenario for the 2020-2021 budget. What does the bestcase scenario look like?
 - Staff Response: The budget presented is a base case or best assessment of where we expect revenues and costs to be over the next year. It does not reflect a worst case scenario as actual results may be better or worse than forecasted.
- Is the financial reserve on track to cover 50% of operating expenses for 182 days?
 - Staff Response: Under the current budget, we are not forecasting significant contributions to reserves for the 2020-2021 fiscal year.
- How would EBCE's total net reserves change if reserves were used to lessen the budget impact of risks, rather than the proposed changes to the value proposition?
 - Staff Response: Total net reserves could decrease depending on whether cash that is not held in reserve is utilized or funds placed in reserves.

Value Proposition

- Why is the value proposition changing for Brilliant 100, but not for Renewable 100?
 - Staff Response: The cost of Brilliant 100 has been adjusted to reflect the cost of service. Renewable 100 already accurately reflected the cost of service.
- EBCE's value proposition appears to be weaker than other CCAs. How are other CCAs managing their 2020-2021 budgets?
 - Staff Response: This information is covered in the May 20, 2020 staff <u>presentation</u> to the Board on slides 22, 23, and 24.



Power Mix

- What are the projected greenhouse gas emissions and product mix for each product, particularly in light of the April 22, 2020 decision to anchor the Bright Choice product at a minimum of 5% higher renewable energy content than PG&E's projected renewable energy content?
 - Staff Response: For purchases made in 2019 EBCE will likely be reporting emissions through The Climate Registry which was used for calculating emissions from 2018 purchases. The emissions from 2019 will be calculated during this summer and made public near the end of the year. Bright Choice is currently estimated at 110 pounds per MWh, which is below the previously established (2018) emissions benchmark of 142 pounds per MWh. Emissions from 2020 purchases will likely be included in the Power Content Label that will be available to customers near the end of 2021. EBCE does not currently have forecasts of the emissions associated with the current year 2020 purchases, but it is anticipated that the emissions level will increase.



Public Comment Period: Friday, May 22, 2020 - Sunday, June 7, 2020

Staff hosted three public meetings the week of June 1, two webinars, and one audio-only workshop. These events were advertised via social media and the EBCE website starting on Friday, May 22 and by email on Tuesday, May 26. The sponsored posts on Facebook had 22,334 views by 10,816 users; on Twitter the advertisement had over 34,000 impressions. The email was sent to over 2,500 subscribers with a 60 percent open rate.

Six verbal comments were received at the webinars, which were attended by 15 community members.

Staff received multiple requests for an example of the bill amount under the new value proposition. Table 1 shows EBCE bills with the current, proposed, and alternate proposals to the Bright Choice. A 1% discount compared to today's value proposition is less than an additional \$0.25 for a resident, a small commercial customer \$1.30 and a large commercial customer would see a monthly increase around \$180.

Table 1: Sample bill totals with the current, and proposed value proposition, for Bright Choice

					Total EBCE Cha	rges		
		1.5% discount	1.0% discount	0.5% discount	Proposed (1.0%) Bright Choice versus Current Bright Choice		Alternate (0.5%) Bright Choice versus Current Bright Choice	
Customer Class	Monthly Energy Use	Current Bright Choice	Proposed Bright Choice	Alternate Bright Choice	Monthly Difference	Yearly Difference	Monthly Difference	Yearly Difference
Residential (E1)	380 kWh	\$ 43.99	\$ 44.21	\$ 44.43	\$ 0.22	\$ 2.64	\$ 0.44	\$ 5.28
Small Commercial, Winter (A1X)	2,270 kWh	\$254.55	\$255.85	\$257.14	\$1.30	\$15.60	\$2.59	\$31.08
Large Commercial, Summer (E19S)	1,014 kW / 294,710 kWh	\$36,232,18	\$35,411.02	\$35,589.86	\$178.84	\$2,146.08	\$357.68	\$4,292.16

Table 2 shows bills for the same residential, small commercial, and large commercial customers on Brilliant 100 - today, with the proposed value proposition, and alternate proposal. The same customer would pay more on Renewable 100 than either proposed Brilliant 100 product.

Table 2: Sample bill totals with the current and proposed value proposition for Brilliant 100 compared to Renewable 100

		Total EBCE Charges							
		Parity with PG&E	PG&E + 3%	PG&E + 2.5%	Proposed (3%) Brilliant 100 versus Current Brilliant 100		Alternate (2.5%) Brilliant 100 versus Current Brilliant 100		PG&E + \$0.01
Customer Class	Monthly Energy Use	Current Brilliant 100	Proposed Brilliant 100	Alternate Brilliant 100	Month ly Differen ce	Yearly Difference	Monthly Difference	Yearly Difference	Renewable 100
Residential (E1)	380 kWh	\$44.66	\$46.00	\$45.77	\$1.34	\$16.08	\$1.11	\$13.32	\$48.46
Small Commercial, Winter (A1X)		\$258.43	\$266.18	\$264.89	\$7.75	\$93.00	\$6.46	\$77.52	\$281.13
Large Commercial, Summer (E19S)			\$36,841.77	\$36,662.92	\$1,073.06	\$12,876.72	\$894.21	\$10,730.52	\$38,715.81

Comments

Customers and advocacy groups provided a variety of comments and questions for EBCE staff. Many individuals and groups highlighted the need for EBCE to be mindful of the economic recession caused by the COVID-19 pandemic and make all attempts to keep rates as low as possible to reduce the energy burden on our community members.

The most commonly stated value proposition and budget recommendations:

- Do not reduce the Local Development Business Plan (LDBP) budget;
- The Board should consider using the reserves to ease the shortfall caused by the Power Charge Indifference Adjustment (PCIA), and other budgetary risks;
- Consider cost-cutting measures regarding staff, such as freezing salaries and hiring, and allow for voluntary staff salary reductions; and,
- EBCE and CalCCA should recommend to the California Public Utilities Commission (CPUC) that the PCIA be reduced, or removed.

Additional comments included:

- Maintain a product that is priced lower than PG&E's base rate;
- Consider the required timeline for City Councils to update their default product, should Brilliant 100 become more expensive than PG&E. Do not implement this change immediately;
- Set the Renewable 100 rate so that it is consistently lower than PG&E's Solar Choice product; and,
- Set the Renewable 100 rate to the market rate for renewable energy.
- EBCE staff should look to supplement local development funding for efficiency measures, such as electrification for lower- and middle- income households.
- Equity metrics must be developed and utilized to measure the economic and social welfare impact of LDBP programs on low-income communities.

In addition to recommendations, staff received requests for additional information, or clarification.

Local Development Business Plan:

- If staff and the Board eliminate \$2.7 million from the LDBP, would that leave only the electric vehicle (EV) program?
 - Staff Response: A reduction in LDBP/Programs budget will be applied based on the direction from the Board.
- Why is 70% of the LDBP budget allocated to the EV program? EVs are not available to everyone?
 - Staff Response: EBCE's Board of Directors unanimously approved participation in the CEC's 2021 California Electric Vehicle Incentive Program (CALeVIP) (EViP) to leverage state incentive funding to enable Alameda County to meet its share of the state target of 1.5M EVs on the road by 2025. Investment from CALeVIP is for EV charging infrastructure, not vehicles. Additionally, single family homeowners are not eligible for CALeVIP incentives for charging infrastructure.
- What percentage of the EV program will go towards fleet, municipal, and light/heavy duty trucks instead of personal vehicles?

 Staff Response: The Local Development Budget allocations are stated in the May Board Packet. Municipal fleet electrification support, and our partnership with CALSTART are explicitly focused on light, medium and heavy-duty vehicle electrification and infrastructure deployment. Incentives through CALeVIP will support charging infrastructure for fleet managers. The exact proportion of CALeVIP incentive allocation is impossible to estimate as applications are submitted on a first come first serve basis per the requirements of the CEC.

Power Procurement:

- In the proposed budget for Fiscal Year (FY) 2020-21, the cost of electricity is shown as \$55.28/MWh. How was that amount calculated? How does this compare to historic pricing?
- How much of EBCE's 2020-2021 power is already procured and how much remains open?
- What does it mean that power costs are "soft?"
 - $\not\subset$ Staff Response: "Soft" power costs refers to low power costs in the market.
- What is the average price of each type of energy fossil fuel, carbon-free, and renewable?

2020 Forecast Adders								
		PG&E	SCE					
Energy	On Peak	\$36.88	\$37.61	\$37.61				
Index ² (\$/MWh)	Off Peak	\$29.93	\$29.56	\$29.56				
	System RA	\$4.59						
RA Adder (\$/kW-month)	Local RA \$4.11 \$4.15			\$4.30				
	Flexible RA	\$4.41						
RPS Adder (\$/MWh)			\$17.35					

 <u>Performance.pdf</u>) and the "2020 Padilla Report" published by the CPUC in May 2020

(https://www.cpuc.ca.gov/uploadedFiles/CPUCWebsite/Content/About_Us/Organ ization/Divisions/Office_of_Governmental_Affairs/Legislation/2020/2020%20Padil la%20Report.pdf?__ac_lkid=2a14-b0f6-39ef-d2f417268072d07)

- What are the cost details for executed power contracts?
- What is the relationship between procurement costs and proposed rates for EBCE's products? Is there a direct relationship or are rates calculated using some other methodology?
 - ✓ Staff Response: –There is a direct relationship between procurement costs and the value proposition. EBCE's rates are also based on PG&E's rate structure. The discount is proposed based on maximum savings while still competing with PG&E rates.

Budget/operations:

- Staff has provided a worst-case scenario for the 2020-2021 budget. What does the bestcase scenario look like?
 - Staff Response: The budget presented is a base case or best assessment of where we expect revenues and costs to be over the next year. It does not reflect a worst case scenario as actual results may be better or worse than forecasted.
- Is the financial reserve on track to cover 50% of operating expenses for 182 days?
 - Staff Response: Under the current budget, we are not forecasting significant contribution to the reserves after the 2021 fiscal year.
- How would EBCE's total net reserves change if reserves were used to lessen the budget impact of risks, rather than the proposed changes to the value proposition?
 - Staff Response: Total net reserves could decrease depending on whether cash that is not held in reserve is utilized or funds placed in reserves.

Value proposition:

- Why is the value proposition changing for Brilliant 100, but not for Renewable 100?
 - Staff Response: The cost of Brilliant 100 has been adjusted to reflect the cost of service. Renewable 100 already accurately reflected the cost of service.
- EBCE's value proposition appears to be weaker than other CCAs. How are other CCAs managing their 2020-2021 budgets?
 - Staff Response: This information is covered in the May 20, 2020 staff presentation to the Board on slides 22, 23, and 24.

Power Mix:

- What is the projected greenhouse gas emissions and product mix for each product, particularly in light of the April 22, 2020 decision to anchor the Bright Choice product at a minimum of 5% higher renewable energy content than PG&E's projected renewable energy content?
 - Staff Response: For purchases made in 2019 EBCE will likely be reporting emissions through The Climate Registry which was used for calculating emissions from 2018 purchases. The emissions from 2019 will be calculated

during this summer and made public near the end of the year. Bright Choice is currently estimated at 110 pounds per MWh, which is below the previously established (2018) emissions benchmark of 142 pounds per MWh. Emissions from 2020 purchases will likely be included in the Power Content Label that will be available to customers near the end of 2021. EBCE does not currently have forecasts of the emissions associated with the current year 2020 purchases, but it is anticipated that the emissions level will increase.

Comments and questions were received from the following individuals and/or organizations:

Verbal Comments	Written Comments
Tom Kelly, Kyoto USA	Tom Kelly, Kyoto USA
Richard Rollins, Berkeley	Richard Rollins, Berkeley
Aleta Dupree, Oakland	Michael Katz, Berkeley
Barbara Stebbins, Berkeley	Edward Dijeau, Union City
Igor Tregub, Berkeley, as a resident	Beth Weinberger, Kehilla Community Synagogue
Erik Pearson, Environmental Services Manager, City of Hayward	Jessica Tovar, East Bay Clean Power Alliance
	Igor Tregub and Minda Berbeco, Sierra Club Northern Alameda County Group and SF Bay Chapter (Late Submission)

Attachments

- 1. PDF of Presentation made during webinars
- 2. Links to Recorded Webinars
- 3. Written Comments



1

AGENDA

- Introduction to East Bay Community Energy (EBCE)
- Background
- 2020 2021 considerations
- 2020 2021 recommendation
- Public comment period



<section-header><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item>












DEFINITIONS

- Rates what EBCE charges customers for electricity, currently updated when PG&E changes their generation rates or PCIA
- (Rate) Value proposition the financial savings customers may realize as an EBCE customer
 - EBCE brings additional value to our community. This is only about the direct financial value to our customers.
- **Power Charge Indifference Adjustment** (**PCIA**) - a per kilowatt-hour fee that PG&E assesses to a customer that does not receive electric generation service from PG&E.
 - EBCE's value proposition must include the PCIA, since a customer who receives both delivery and generation from PG&E does not pay this fee.





EAST BAY









NEW VALUE PROPOSITION PRESENTATION

- New value proposition proposed as part of the pending Fiscal Year 2020 -2021 Budget
- Presented to Finance Subcommittee on Friday, May 15, 2020
- Presented to Community Advisory Committee on Monday, May 18, 2020
- Presented to Board of Directors on Wednesday, May 20, 2020
- Presented to Executive Committee on Friday, May 22, 2020
- Copy of presentation available at <u>https://ebce.org/meetings/board-of-directors-5-20-20/</u>

13



15

RISKS

The value proposition is closely tied to the budget.

EBCE is exposed to a variety of risks in this coming fiscal year. The following is a representative list of key risks, though it is not comprehensive in reflecting all risks.

- COVID/Load Risk
- Energy Market Risk
- Uncollectables/Write-off Risk
- Opt-out Risk
- Regulatory Risks





17

CONSIDERED SCENARIOS Alternative 1: Reduce Local Development budget by \$2.7MM, maintain 1. the Bright Choice discount of 1.5% Alternative 2: Increase Local Development budget by \$2.7MM, further 2. decrease Bright Choice discount to 0.5% Alternative 3: Reduce Bright Choice discount further to 0.5% and 3. increase net margin by \$2.7MM to mitigate risks related to energy markets, COVID, and undercollection and build increased reserves Note – proposed Local Development budget is currently \$6.75MM EAST BAY COMMUNITY ENERGY





HOW-TO

- Click on the chat bubble
- Enter your <u>name, city, and organization</u> into the chat box
 - Note that everyone on the call will see what you type into the chat box
- We will call on each speaker in the order we received the chat, we'll then disable the mute feature for each speaker
- Please be conscientious of the time. Every speaker will be allowed four minutes.
- Note that this webinar is being recorded and will be posted to our website



20

THANK YOU FOR PARTICIPATING

• EBCE staff's next steps:

- Post this webinar, including the audio recording of your comments, on our website at ebce.org/rates
- Summarize all public comments in the staff report to Board
- Present summary and value proposition recommendation at Board of Directors meeting on June 17 (teleconference)
- Additional comment opportunities:
 - **Online**, webinars on Tuesday, June 2 at noon and Wednesday, June 3 at 6pm
 - Phone/audio-only on Friday, June 5 at noon
 - Written comments may be emailed to <u>PublicComment@ebce.org</u>
 - Due to Shelter in Place orders, staff remains working from home and we are not able to receive physical mail delivered to our office. Therefore, written comments are only accepted via email.
- All comments are due by Sunday, June 7, 2020 at 11pm



21



Public comments were gathered at two webinars. Links to the staff presentation and the comments can be found on our website at ebce.org/rates and here:

Tuesday, June 2 at noon Wednesday, June 3 at 6pm

There were no attendees to the phone/audio-only presentation on Friday, June 5 at noon.

Public Comment to EBCE FY 2020-2021 Rate Setting

Submitted June 4, 2020 Audrey Ichinose East Bay Clean Power Alliance California Alliance for Community Energy

Rate-setting for FY 2020-2021 needs to be done in a broad context. It's

commendable that EBCE, as CEO Chaset stated, simply aims to "balance the books." EBCE is not a for-profit entity. Nor can it afford to operate in the red. Yet in today's context of the pandemic and Black Lives Matter, EBCE needs to take a wider view. It needs look beyond its laudable budget-balancing focus to uphold other important goals, i.e. its commitment to social equity and local development.

In light of the current crises, I would like to suggest the following for the FY 2020-2021 budget, EBCE's third:

- Dipping into the reserves to ease the shortfall caused by the PCIA while keeping rates largely unchanged
- Freezing salaries and allowing for 10% voluntary salary cuts for Staff out of solidarity with EBCE's hard pressed communities
- Supplementing local development funding, made possible in part by budget savings

Dipping into Reserves. The Reserves policy was established by the FY 2018-2019 budget, EBCE's first. It established three Reserves. ("Overview of Expenditures: Reserve Contributions and Proposed Reserve Policy," Attachment Item 15B, EBCE Board Meeting, June 20, 2018.) By far the largest, the Financial Operating Reserve was created to "move swiftly" to build necessary reserves in its first 3-5 years of operation in order to apply for a credit rating in its 4th or 5th year. Staff consulted with Barclay's Bank about the necessary reserve level to achieve an "A" rating or better. That target level was set at an operating reserve of 50% of operating expenses, enough to cover 182 days.

Now, as EBCE enters only its third year of operation and is faced with colliding crises, where does EBCE stand with this important goal? Is its Financial Reserve well on track to cover 50% of operating expenses for 182 days?

If it is well on track to apply for a credit rating in 2022 or 2023, EBCE should dip into its Financial Reserve or reduce its FY2020-21 reserve allocation in order to support the critical goals of the moment, social equity and local development funding. The two other initial reserves, **Rate Stabilization** and **Collateral**, should also be reexamined. Circumstances may warrant dipping into them, too. **Supplementing Local Development Funding.** The decision to fund eVIP for Alameda County's eV charging infrastructure blows a big hole in local development funding. Although many are alarmed by certain problems—the continued support of personal vehicles over the use of public transit, a reliance on incentives that may never bear fruit, vaguely defined, "passive" benefits for low-income communities, etc.—the program promises GHG emissions reductions in the transport sector and helps Alameda County overcome its lag in eV charging infrastructure with the help of state funds.

Nevertheless, the decision to fund eVIP handcuffs local development actions for the next four years and possibly longer. Once millions are invested, more money will undoubtedly be sucked in to realize the promise of this complicated program.

A criticism of eVIP is that it only vaguely outlines benefits for low-income, impacted communities of color. One would want to see more concrete benefits. Everybody benefits from improved air quality, not just the people forced to live near the Port of Oakland and along major transport routes. Impacted communities, including its churches, need to receive tangible "ownership" benefits from local development funds. The following, for example, are benefits listed for a recent BayREN Home+ program that offers rebates for middle-class homeowners for

- Efficient air conditioning and/or furnaces
- Efficient water heaters
- Attic and wall insulation
- Duct sealing and/or replacement
- Weatherization
- + Electrification Options (heat pump hot water heaters, space heating/cooling, and clothes dryer, and induction cooktop or range)..."

If eVIP cannot be modified or attenuated in some way to counter what may be a deep multi-year recession, EBCE needs to commit to supplementing the local development budget for the next few years. In addition to using reserves, the savings from freezing salaries this year could contribute toward such funding.

Personnel compensation. EBCE staff is extremely hard-working, conscientious, and technically expert in many areas. The following proposal is in no way a criticism of Staff or of the sound principle of retaining and promoting one's best employees. In light of the loss of jobs and hardships visited on local businesses, however, EBCE should undertake a freeze on salaries (bonuses, merit increases and the like) and make possible the voluntary 10% pay cuts from higher salaried employees.

The value of this action would go far beyond its limited budgetary impact. Such an action would show EBCE standing in solidarity with its community, many parts of which are suffering. It would also serve as an excellent public reminder that EBCE is a community agency serving its many diverse constituencies. Governor Newsom (salary \$210,000) and his staff, state legislators (\$115,000 plus \$200/day per diem), and the nine-campus UC system have all set the example. EBCE should do the same. These budget savings could also help to supplement local development funds.

Rethinking the rates for Renewable 100 and Brilliant 100. The demand "elasticity" of these two customer groups would strongly suggest that a better balance is needed between the rates for these products. Those who choose Renewable 100 are likely to be more firmly committed to fighting climate change and better able to absorb the added costs. They could be asked to contribute more without EBCE running too great a risk of opt-outs. These customers could, of course, always opt down. The Brilliant 100 group, on the other hand, is roughly 10 times larger and exhibit more elastic demand. Their number will undoubtedly sink under a 3% increase. They, too, can opt down—or out, depending on their shock or disgust at the new rate. A $1\frac{1}{2}$ to 2% increase—in tandem with a rate hike for Renewable 100—could probably yield just as much in the long run without an increase in opt-outs.

Finally, could Staff consider finding some way of tying the Renewable 100 rate to the market price for renewable energy? It would add an attractive marketing dimension to the product, even though a fluctuating Renewable 100 rate would hardly be a gamble in the long-term, given the steady sinking of renewable energy prices.

EBCE needs to take into account the exceptional, tragic circumstances it faces this year. It cannot carry on as though the picture has not changed. This is true for conditions not just in Alameda County, the Bay Area and California, but nationally as well.



Thank you for participating in our public comment process 2020-21 Budget

Beth Weinberger

bethw0104@mindspring.com>

To: "PublicComment@ebce.org" <PublicComment@ebce.org>

Wed, Jun 3, 2020 at 9:26 AM

I am writing on behalf of Kehilla Community Synagogue, an Alameda County congregation of 500 households.

We are especially concerned about and oppose any cuts to the Local Development Business Plan (LDBP) which is the heart and soul of our

County's community choice energy program. Preserving it and funding it well are much more important than small increases in rates. The LDBP's capacity to create local clean energy jobs and promote and support local businesses is even more important in the current economic crisis caused by the pandemic.

We strongly oppose investing a proposed 70% of the LDBP budget exclusively in an electric vehicle program as individual electric vehicles are a luxury item unaffordable to low-income communities. Unless this investment in EVs is aimed primarily at public transit vehicles, trucking fleets and the like, this proposed earmark would be discriminatory against low-income and communities of color.

Given that budget shortfalls in this budget year are largely due to the global pandemic, why isn't a portion of the estimated \$73 million in emergency reserves being used to make up these shortfalls? And, given that the other main reason for EBCE's decreases in revenue is the increase in and unwarranted longevity in the PCIA, why aren't you fighting more to eliminate or reduce this charge?

Finally, transparency in ratesetting by a public agency is fundamental and required. An important part of your process must be to fully disclose the costs involved in acquiring each form and source of energy as the basis for ratesetting.

Beth Weinberger

Oakland resident and representative of Kehilla's Greening Committee

Sent from Mail for Windows 10



Subject: Budget + Rate Setting 2020-2021

June 7, 2020

Dear EBCE Board and Staff,

East Bay Clean Power Alliance appreciates the daunting task that the agency faces in presenting a balanced budget for 2020-2021, especially at a time when a global pandemic has left our communities devastated, and Black Lives Matter uprisings have highlighted the neglect of equity and the urgency of addressing the destructive impact of racial disparity. Our comments on the proposed budget and rates fall under two broad topics, the long term health of our communities and the agency, and the need for greater transparency by staff so that both the board and the public can make informed decisions.

Long Term Health of Community and Agency

East Bay Community Energy was established with goals of prioritizing the local development of clean energy resources in order to equitably benefit our communities. As demonstrated by EBCE's Local Development Business Plan, adopted by the Board in 2018, local development can provide clean, affordable energy and increase economic, social and racial equity by addressing the needs of low income communities and people of color. EBCE has so far not delivered on this promise, contracting for utility-scale energy from remote sources that has little impact on local economies and local needs.

Prioritizing local development has become even more urgent in the shadow of utility power shut offs and a global pandemic. The first emphasizes the need for local energy resilience, and the second emphasizes the need for local job creation to address the 25% pandemic unemployment rate, a number which undoubtedly undercounts those without jobs in areas of Alameda County chronically plagued by high unemployment.

Furthermore, EBCE's emphasis on cheaper remote energy has not protected the agency from the financial challenge of the ever increasing PCIA fee our customers pay to PG&E. The long-term viability and financial health of EBCE requires a successful strategy for beating back this attack on local energy autonomy.

- The Alliance rejects any cuts to the Local Development Business Plan (LDBP), especially now because with 70% of that budget going to electric vehicle charging, only 30% remains that could significantly benefit low income communities.
- We reject any budget proposal that would substitute an average \$1/ year savings to Bright Choice customers for the remaining LDBP budget (\$2.7 million) not earmarked for electric vehicle charging.
- If \$2.7 million is required for mitigating risk or to make-up budget shortfall, we recommend that amount be taken out of the \$73 million in EBCE reserve accounts, established for times of crisis such as the one we are in.
- Equity metrics must be developed and utilized to measure the economic and social welfare impact of Local Development Business Plan programs on low income communities.
- EBCE should join with other Community Choice programs in the state to fight the PCIA, in the courts, the CPUC, and the Legislature. It is outrageous that for-profit utilities be able to charge customers on an ongoing basis when they choose public agencies for better energy services.

Increased Transparency

In order to make an informed decision about budget and rates, the EBCE Board and the public must have more complete information than they have been given by staff. As we understand it, the current projections of expenses and revenues are a worst-case scenario. What does the best case look like?

• This year, the information presented to the Community Advisory Committee, Board and the public has not been sufficient for anyone to judge the impact of staff proposals on the budget and rates.

- Staff should provide the average costs to EBCE of renewable, non-renewable carbon-free and fossil fuel based energy, the three different types of energy that serve customers, as well as the differential between the cost and the rate charged for each class of energy user. Reporting average costs would not violate individual contract confidentiality and is critical information for decisions on rates and power content.
- Staff's current proposal for rates, particularly the 3% increase in Brilliant 100, comes without any accounting to justify that increase. Staff needs to provide that information.
- Staff should always provide the impact of rate changes on the average customer's bill, residential, municipal and commercial. Using figures presented by EBCE staff in the past, we estimated the proposed ½% decrease in Bright Choice discount would result in only a \$1/ year increase on the average residential bill.
- The incomplete information that is presented tends to predispose certain decisions, without consideration of other possibilities. For instance, there are only three alternative budget scenarios presented by staff. There is no information about other ways to make up the anticipated budget shortfall. For example, there is no mention of the reserves or how the proposed 3% increase in the Brilliant 100 rate might affect revenues.
- Two of the alternate budget scenarios involve contradictory actions, pitting Bright Choice rates against a \$2.7 million increase or decrease in the LDBP budget. Given that the LDBP budget makes up only 1.7% of EBCE's total expenses this seems almost a targeted proposal to get rid of what remains in the LDBP outside of the electric vehicle charging proposal.
- None of the alternate budge scenarios will come close to mitigating the over \$60 million budget shortfall predicted for 2020-2021.
- We would like to understand what happened to the \$21 million that was supposed to go into reserves according to staff in the June 19, 2019 budget presentation, including \$9.6 million budgeted for the LDBP reserves. The actual 2019 contributions shown in the 2020 Budget presentation are \$40.5 million total and only \$3.8 million for the LDBP (see below)

From Slide 25, Budget Presentation, June 19, 2019

EBCE Net Position FY19-20 68,391,000

Reserve Funds	2018-19	2019-20	Total	Target	Remaining
Operating Credit	38,537,100	48,514,600	87,051,700	207,747,000	120,695,300
Rate Stabilization	5,780,600	1,496,590	7,277,190	7,277,190	-
Collateral	7,707,400	13,528,350	21,235,750	39,656,800	18,421,050
Local Development	9,634,300	4,851,460	14,485,760	48,514,600	34,028,840
Total Reserves	61,659,400	68,391,000	130,050,400	303,195,590	173,145,190

From Slide 21, Budget Presentation, May 20, 2020

EBCE Estimated Net Position 2019-2020 \$ 63,690,000

Reserve Fund	2018-19	2019-20*	Total	2019-20 Target	Remaining
Operating/Credit	32,768,648	23,393,450	56,162,098	200,646,000	144,483,902
Rate Stabilization	1,936,260	2,339,345	4,275,605	7,018,035	2,742,430
Collateral	1,936,260	2,339,345	4,275,605	37,742,400	33,466,795
Local Development	3,872,519	4,678,690	8,551,209	46,786,900	38,235,691
Total	40,513,687	32,750,830	73,264,517	292,193,335	218,928,818

* Expected contributions set to minimum amount

EBCPA urges the Board to reject any cuts to the Local Development Business Plan and ask for more information and alternative solutions before making any decisions on the 2020-2021 budget.

Thank you,

Permon

Jessica Tovar, Coordinator of the East Bay Clean Power Alliance



Thank you for participating in our public comment process Setting Rates without giveaways or donations that keeps rates higher than they should be. You are a public trust.

Fri, May 29, 2020 at 3:07 PM

1 message

EDWARD DIJEAU <Dijeau@msn.com> To: "PublicComment@ebce.org" <PublicComment@ebce.org>

Dear Sirs.

When setting rates, they should be just the cost of electricity, from renewable providers, and the office overhead and should not contain "rip offs" of rate payers to pay for a "special class" of rate payer or to give away money to charitable organizations. Special rates for the low income, seniors, the disabled or non-profits should only be given by PG&E and taken from the proceeds "they get" from distribution of electricity since their rates are 70% of "all rate payers bills" with the exception of solar customers who seem to be getting a "Free Ride" on every one else by not paying their fair share of the infrastructure portion of the costs including the "giveaways" with the Net Metering (NEM) they get. We have some of the highest electric rates in the country and part of that is because of California mandates for clean energy and there are a lot of programs that force rate payers to pay part of the bill for low income classes of people. As the Energy provider for Alameda and Contra Costa Counties, East Bay Energy should stay out of politics and making it's cost of energy by NOT taking on programs that shift the rate burden onto the hard working middle class from the rich "solar owning NEM customers" or poor "low income" rate payers. Let PG&E handle the CARE program through it's own energy program and suppliers and not force East Bay Energy to shift the burden onto it's clean, renewable energy programs, making them more costly. If you want to be believed that you are getting us consumers true renewable energy at the lowest possible cost, we can not keep reading that you are giving our moneys away. You are a public trust.

Thank you for your consideration on this matter.

Edward F Dijeau

50 year member IBEW Local 595



Thank you for participating in our public comment process 2020-2021 rate setting: Please lower "Renewable 100" to a competitive rate 1 message

Michael Katz <mqkatz@gmail.com>

Tue, May 26, 2020 at 6:42 PM

To: East Bay Community Energy public comments <PublicComment@ebce.org>

Dear EBCE Staff,

As you set 2020-2021 pricing, I strongly urge you to reset the "Renewable 100" tier to a rate no higher than – and ideally, lower than – PG&E's "Solar Choice" alternative.

EBCE is a sizable aggregator, negotiating with a financially hobbled utility. Something is seriously wrong if EBCE lacks the bargaining power, the purchasing skill, or the purchasing options to price an allrenewable tier competitively with the rate that PG&E charges individual ratepayers.

For my purposes, I don't care if the "Renewable 100" power mix remains 50% wind, or remains tied to locally produced wind power, or includes wind power at all. I just want an all-clean-power option that won't make ratepayers like me feel fools.

By choosing an aggregator, we should not be paying a premium over the utility's comparable retail rate. We should be seeing a discount, as ratepayers on the "Bright Choice" tier do.

Again, for my own purposes: If "Renewable 100" does not get priced competitively with PG&E's "Solar Choice," I will reluctantly opt out as soon as next year's rates are announced, and return to PG&E.

In principle, I strongly support (truly) public power. I grew up in Canada, where most provincial governments proudly run an efficient, end-to-end electrical grid for their regions – as well as selling surplus hydro power to U.S. customers. (Which of course include BC Hydro customers PG&E and EBCE.)

But on principle, I won't support a feel-good version of an aggregator – one that gives local politicians a symbolic achievement to crow about, but fails to deliver reasonable, achievable pricing discounts for its ratepayers on all tiers.

Thank you for considering this plea to set "Renewable 100" rates no higher than PG&E's "Solar Choice" tier. Two years in, this should certainly be doable.

Respectfully yours, Michael Katz Berkeley, CA

Date:	June 5, 2020
То:	East Bay Community Energy Chair Kalb, EBCE Directors, and Community Advisory Committee
From:	Richard Rollins, P.E., LEED AP Berkeley, CA
Subject:	EBCE's Proposed 2020-2021 Budget and Rates

Dear Chair Kalb and EBCE Directors,

Respectfully, I am writing to offer comments and recommendations regarding EBCE's proposed 2020-2021 Budget and Rates.

One of the benefits of CCAs is the opportunity for ratepayers and their elected representatives on the Board to participate when critical decisions are made. Thank you for this opportunity to comment.

Transparency in Budgeting and Rate Setting

Transparency in budgeting and rate setting requires more detailed information than has been provided thus far. Even though the "floor" metric for Bright Choice content was revised on April 22, 2020 (from 85% GHG-free to 39.5% Renewable) the Public is still concerned and the Board should be informed about the estimated GHG and RPS content of EBCE's three products. The implication of slide 24 of the May 20, 2020 Budget Presentation to the Board is that Bright Choice GHG-free content could be as low as 54.5%. Since this EBCE product serves most of the EBCE load, this is a significant change from the initial goals that were outlined in EBCE's formation process and are specifically called out in the JPA Agreement.

To be fully transparent, EBCE Directors should require EBCE staff to provide the Board and the Public with answers to the questions listed in Attachment A before decisions on rates and budget are made.

This is not the time for Rate Increases or Cuts to Community Programs!

The estimated unemployment rate in CA is 25%, up from about 5% in early March. The proposed rate increases will impact both ratepayers and taxpayers within EBCE member jurisdictions. EBCE's proposed rate increase will disproportionately impact the following:

- unemployed and low-income rate payers in all EBCE jurisdictions, particularly in communities that are being disproportionately impacted by decades-long injustices and now the pandemic.
- customers in the cities of Albany and Hayward who were automatically enrolled in Brilliant 100

With so many people affected by the COVID-19 pandemic, EBCE's priorities for the 2020-2021 budget should be to find ways to cut costs so that programs and low carbon content can be maintained without increasing rates. The combined impacts of lost jobs, shuttered businesses, and reduced revenue to member jurisdictions cannot be ignored. EBCE must recognize that low-income ratepayers are most severely affected. EBCE must maintain its commitment to the LDBP, which can be a job creator for many disadvantaged community members.

Any rate increase at this time will be seen as regressive and insensitive to those ratepayers impacted by the COVID-19 pandemic, unemployment, and the pending economic recession.

EBCE's Reputation and Trust with the Public is at Risk

If approved, EBCE's proposed rate increases (combined with eroded goals for GHG-free content) will reduce the products' value proposition and undermine the trust that the public has extended to EBCE.

Recommendations

I strongly suggest that EBCE staff provide additional information as noted above followed by a robust public process that allows the public to weigh in before the budget and rates are finalized for 2020-2021.

Sincerely, Richard Rollins, P.E., LEED AP Berkeley, CA

ATTACHMENT A: Comments and Questions regarding Transparency and Scenario Analysis

Transparency

- What is the projected Renewable content of Bright Choice and Brilliant 100? How is this different than previous year actual performance?
- What is the projected cost of renewables, GHG-free power, low carbon power (ACS) and "system" power? How does this compare with previous year costs and product power content?
- What is the projected GHG free content of Bright Choice? (slide 6 of the May 20, 2020 Board presentation indicates projected energy cost reduction due to revised power content targets) How do the new targets compare with PG&E on CO2 intensity and/or GHG-free content; with previous EBCE procurement targets and actual CO2 intensity and/or GHG-free content?
- With the proposed 3% increase to Brilliant 100 rates, what is the resultant difference between Brilliant 100 and Renewable 100?
- What are the actual costs for the "soft" energy (slides 4 and 6 of the May 20, 2020 Board Presentation on budget) and what are the sources?
- What is relationship between procurement cost and proposed rates for EBCE's three offerings? Is there a direct relationship or are rates subjective and arbitrary?

More scenario analysis is warranted

- The only scenario presented is a LDBP vs Rate analysis
- A reduction to the LDBP budget was essentially the only alternative scenario presented by EBCE staff. The Board should request that staff prepare other scenario options and share with the Board and Public prior to rate and budget decisions. Options should include:
 - reduce the reserve allocation
 - use the rate stabilization reserve
 - freeze EBCE hiring and salaries
- Slide 33 of the May 20, 2020 Board presentation indicates a possibility of selling "excess large hydro related carbon free attributes". This suggested possibility needs more explanation regarding its impact on rates, budget, and GHG-free content as well as details about the decision-making process that determines "excess".
- What is the impact of the proposed rate increases on member jurisdiction budgets? Since EBCE knows the load for each member jurisdiction, taxpayers and member jurisdiction Board members deserve an estimate of financial impact on their jurisdiction's budget for 2020-2021.
 - How can this impact be reduced?
 - What short-term financial solutions have been considered?
- Are other CCAs increasing rates and reducing carbon free content? If not, how are they managing the combined impacts of the increased PCIA and the COVID-19 pandemic?



June 7, 2020

Serving Alameda, Contra Costa, Marin and San Francisco counties

EBCE Board of Directors 1999 Harrison Street, Suite 800 Oakland, CA 94612

SUBMITTED VIA E-MAIL

Dear EBCE Board of Directors,

We hope this letter finds you well. Thank you for all your work, especially in these trying times.

One of the benefits of CCAs is the opportunity for ratepayers and their elected representatives on the Board to participate when critical decisions are made. Thank you for this opportunity to comment on the 2020-2021 Budget and EBCE Rates.

Transparency in Budgeting and Rate Setting

Transparency in budgeting and rate setting requires more detailed information than has been provided thus far. Even though the "floor" metric for Bright Choice content was revised on April 22, 2020 (from 85% GHG-free to 39.5% Renewable) the Public is still concerned and the Board should be informed about the estimated GHG and RPS content of EBCE's three products. The implication of slide 24 of the May 20, 2020 Budget Presentation to the Board is that Bright Choice GHG-free content could be as low as 54.5%. Since this EBCE product serves most of the EBCE load, this is a significant change from the initial goals that were outlined in EBCE's formation process and are specifically called out in the JPA Agreement.

To be fully transparent, EBCE Directors should require that EBCE staff provide the Board and the Public with answers to the questions listed in Attachment A before decisions on rates and budget are made.

This is not the time for Rate Increases or Cuts to Community Programs!

The estimated unemployment rate in CA is 25%, up from about 5% in early March. The proposed rate increases will impact both ratepayers and taxpayers within EBCE member jurisdictions. EBCE's proposed rate increase will disproportionately impact

- unemployed and low-income rate payers in all EBCE jurisdictions, particularly in communities that are being disproportionately impacted by decades-long injustices and now the pandemic.
- customers in the cities of Albany and Hayward who were automatically enrolled in Brilliant 100

With so many people affected by the COVID-19 pandemic, EBCE's priorities for the 2020-2021 budget should be to find ways to cut costs so that programs and low carbon content can be maintained without increasing rates. The combined impacts of lost jobs, shuttered businesses, and reduced revenue to member jurisdictions cannot be ignored. EBCE must recognize that low-income ratepayers are most severely affected. EBCE must maintain its commitment to the Local Development Business Plan (LDBP), which is a job creator for many disadvantaged community members. More scenario analysis is warranted and options should include: reducing the reserve allocation, using the rate stabilization reserve, and/or freezing EBCE hiring and salaries

Any rate increase at this time will be seen as regressive and insensitive to those ratepayers impacted by the COVID-19 pandemic and the pending economic recession.

EBCE's Reputation and Trust with the Public is at Risk

If approved, EBCE's proposed rate increases (combined with reduced goals for GHG-free content) will reduce the products' value proposition and undermine the trust that the public has extended to EBCE.

Recommendations

We strongly suggest that EBCE staff provide additional information as noted above followed by a robust public process that allows the public to weigh in before the budget and rates are finalized for 2020-2021.

Respectfully,

hpot Stelfal

Igor Tregub, Chair, Northern Alameda County Group

Miller

Minda Berbeco, SF Bay Chapter, Chapter Director

ATTACHMENT A: Questions regarding Transparency and Scenario Analysis

Transparency

- What is the projected Renewable content of Bright Choice and Brilliant 100? How is this different than previous year actual performance?
- What is the projected cost of renewables, GHG-free power, low carbon power (ACS) and "system" power? How does this compare with previous year costs and product power content?
- What is the projected GHG free content of Bright Choice? (slide 6 of the May 20, 2020 Board presentation indicates projected energy cost reduction due to revised power content targets) How do the new targets compare with PG&E on CO2 intensity and/or GHG-free content; with previous EBCE procurement targets and actual CO2 intensity and/or GHG-free content?
- With the proposed 3% increase to Brilliant 100 rates, what is the resultant difference between Brilliant 100 and Renewable 100?
- What are the actual costs for the "soft" energy (slides 4 and 6 of the May 20, 2020 Board Presentation on budget) and what are the sources?
- What is relationship between procurement cost and proposed rates for EBCE's three offerings? Is there a direct relationship or are rates subjective and arbitrary?

More scenario analysis is warranted

- The only scenario presented is a LDBP vs Rate analysis
- A reduction to the LDBP budget was essentially the only alternative scenario presented by EBCE staff. The Board should request that other scenario options be prepared and shared with the Board and Public prior to rate and budget decisions. Options should include:
 - reduce the reserve allocation
 - use the rate stabilization reserve
 - freeze EBCE hiring and salaries
- Slide 33 of the May 20, 2020 Board presentation indicates a possibility of selling "excess large hydro related carbon free attributes". This suggested possibility needs more explanation regarding its impact on rates, budget, and GHG-free content as well as the decision-making process that determines "excess".
- What is the impact of the proposed rate increases on member jurisdiction budgets? Since EBCE knows the load for each member jurisdiction, taxpayers and member jurisdiction Board members deserve an estimate of financial impact on their jurisdiction's budget for 2020-2021.
 - How can this impact be reduced?
 - What short-term financial solutions have been considered?
- Are other CCAs increasing rates and reducing carbon free content? If not, how are they managing the combined impacts of the increased PCIA and the COVID-19 pandemic?



Thank you for participating in our public comment process comments on EBCE rate changes 1 message

Tom Kelly <tkelly@kyotousa.org>

Sun, Jun 7, 2020 at 7:53 PM

To: PublicComment@ebce.org

Cc: Berkeley Mayor's Office <mayor@cityofberkeley.info>, Kate Harrison <kharrison@cityofberkeley.info>, Richard Rollins <rrollins@rollinscs.com>, Chris Gilbert <chris@gilbertbiz.com>

Thank you for the opportunity to comment on the proposed EBCE rate changes.

As others have undoubtedly mentioned, the timing on the proposed changes – raising the rates on all but the Renewable 100 customers – does feel insensitive to the economic hardships facing EBCE ratepayers. It seems especially so since other CCAs are addressing the reduction in electricity sales and the increase in the PCIA by absorbing those costs – rather than passing the increases on to their ratepayers. I am attaching an updated (and corrected) spreadsheet based on slide 24 (Comparison of CCA Products and Value) that EBCE staff provided to the Board on May 24, 2020 (FY2020-2021 Draft Budget). Note that changes and corrections are highlighted in red. This slide is based upon information provided on the CCA websites that in most cases reflect rates as of May 1, 2020. Please note that most CCAs provide a much cleaner basic product at higher discounts than EBCE offers.

1) Please provide the Board and public with an analysis of how other CCAs are handling this situation including a comparison of rates, power content, and impact on reserves. When showing the impact on reserves, please do so for EBCE's total net reserves and not just the impact on net reserves for FY 2020-2021.

Staff has recently reported that power costs are "soft".

2) What does it mean that power costs are "soft"?

3) Please provide cost details for executed power contracts. Staff have mentioned that some contract details need to be kept confidential, but in a review of the California Public Records Act, I am unable to identify an exemption to the disclosure of all contract details.

4) The draft budget that staff provided to the Board indicated that the cost of electricity per MWh in 2019-20 was \$60.42. In the proposed budget for FY 2020-21, the cost of electricity is shown as \$55.28/MWh. Please describe how the figure of \$55.28/MWh was calculated. Please provide a chart or some other visual that describes the history of power procurement prices (renewable, hydro, Asset Controlling Supplier, system power, etc.) for the current fiscal year and estimated power costs for the next fiscal year.

Here is a link to the slide deck from the May 28, 2020 Peninsula Clean Energy Board meeting and discussion on the PG&E hydro/nuclear allocation issue. CEO Jan Pepper points out that due to the COVID-19 crisis, load estimates and procurement needs have fallen. She also points out that estimated power costs for GHG free power have <u>fallen dramatically from about \$8.00/MWh in January to \$3.25/MWh in May</u>.

5) How much of EBCE's 2020-2021 power is already procured and how much remains open? Please provide a similar analysis to the one provided the PCE Board of Directors.

In October 2018, the EBCE Board abandoned the 85% carbon free standard that had been set earlier that year, and set a benchmark that the GHGs associated with EBCE's Bright Choice total retail sales would not exceed 142 lbs CO2/MWh. Once the CEC had completed the Power Content Label for 2018, EBCE reported that they had exceeded the benchmark by producing only 101 lbs CO2/MWh.

6) What do you estimate the 2020 carbon free power mix to be and what are the estimated carbon emissions in lbs/MWh for Bright Choice based on the Board's decision to compete with PG&E on renewables only (PG&E plus 5%)?

I would also like to state that I think cutting the budget for the LDBP is short-sighted. Those funds provide economic opportunities to residents and businesses in the region that is more important than ever. I would suggest looking at other cost cutting measures. I know that others have suggested a freeze on hiring and a delay in raises, but I would also suggest – as a demonstration of solidarity with those who have lost their jobs and have received cuts in income – that EBCE staff also take a 10% reduction in salaries. Doing so would show that EBCE is in tune with the situation in which many of its customers find themselves and would make a significant contribution to assuring EBCE's ability to balance its budget without raising rates or reducing the carbon free content of Bright Choice.

Sincerely,

Tom Kelly

KyotoUSA

Power content comparison (6-2-20).xlsx 12K

5. Comparison of CCA Products & Value

	Basic Product	% Renewable	% Carbon-Free	Total Carbon Free	Discount to PG&E (denotes a premium)	Upgrade 2	% Renewable	Premium to Basic Product
EBCE	Bright Choice	39.5%	10%-15%	54.50%	1.00% (proposed)	Renewable 100	100% RE	Premium to PG&E rate - 0.01/kWh
SVCE	Green Start	50%	50%	100%	4.00%	Green Prime	100% RE	+\$0.008/kWh
MCE	Light Green	61%	13%	74%	-1.00%	Deep Green	100% RE	+\$0.01/kWh
SCP	Clean Start	49%	42%	91%	4%	Evergreen	100% RE	+\$0.025/kWh
SJCE	Green Source	45%	41%	86%	1.00%	Total Green	100% RE	+\$0.005 - \$0.01/kWh
МВСР	MB Choice	34%	66%	100%	2%-7%	MB Prime	100% RE	+\$0.01/kWh
PCE	Eco Plus	50%	40%	90%	5.00%	Eco 100	100% RE	+\$0.01/kWh
CPSF	Green	50%	40%	90%	1.00%	SuperGreen	100% RE	

*Yellow cells denote discounts/premiums relative to last year; not yet confirmed if these CCAs have made adjustments to these values/



FY 2020-2021 Draft Budget

PRESENTED BY: Nick Chaset

DATE: June 17, 2020

FY 2020-21 Draft Budget

(Informational)

- 1. Budget Overview
- 2. Updates to Base Case 3. Overview of Revenue
- 4. Overview of Expenses
 - a. Energy Operations b. Personnel

 - c. Legal, Policy, & Regulatory Affairs
 d. Marketing Services
 e. General and Administration

 - f. Other Professional Services
 - g. Local De h. Interest Local Development
- 5. New Communities
- Overview of EBCE Net Position and Reserve Allocation
 Review of EBCE Value Proposition and Discount
 Risks and Mitigants

- 9. Alternative Budget Scenarios



FY 2020-21 Draft Budget

- 2020-21 Budget reflects staff's current projections for both electricity sales and energy costs under two scenarios
 - COVID impacted load (reduction of load in 2020 of 5-6% in near months, with tapering through December)
 - Non-COVID impacted load
- The FY 2019-20 Budget presented is the budget approved by the Board in June of 2019
- Local Development will be represented as a Fund with a carry over balance of FY 2019-20 unspent dollars in addition to Overhead allocation
- New Communities Revenue and Expenses are not included in the table to the right, but are discussed separately in this presentation

	FY 2019-20 BUDGET	FY 2020-21 Non-COVID	FY 2020-21 COVID
REVENUE AND OTHER SOURCES	_		
Electricity Sales	487,014,000	388,125,000	382,758,000
Investment Income	578,000	1,231,000	1,231,000
Other Income	-	275,000	275,000
Uncollectables	(2,447,000)	(9,703,000)	(9,568,000)
Total Revenue and Other Sources	485,145,000	379,928,000	374,696,000
EXPENSES AND OTHER USES	_		
ENERGY OPERATIONS			
Cost of Energy	386,905,000	347,594,000	342,796,000
Data Management/Customer Service	6,758,000	7,704,000	7,704,000
PG&E Service Fees (Billing/Metering)	2,253,000	2,396,000	2,396,000
Scheduling	653,000	660,000	660,000
Total Energy Operating Costs	396,569,000	358,354,000	353,556,000
OVERHEAD OPERATIONS			
Personnel	6,703,000	7,429,000	7,429,000
Marketing, Outreach, Communications	2,263,000	1,214,000	1,214,000
Legal, Policy, & Reglatory Affairs	1,586,000	1,297,000	1,297,000
Other Professional Services	1,214,000	1,345,000	1,345,000
General & Administrative	2,290,000	2,146,000	2,146,000
Local Development Funding	6,340,000	6,615,000	6,615,000
Depreciation	61,000	60,000	60,000
Total Overhead Operating Costs	20,457,000	20,106,000	20,106,000
INTEREST PAYMENTS			
Borrowing Interest	1,229,000	804,000	804,000
Total Interest Payments	1,229,000	804,000	804,000
2 cm and or a ny mento	1,225,000	004,000	004,000
TOTAL EXPENSES & INTEREST DUE	418,255,000	379,264,000	374,466,000
NET INCREASE (DECREASE) IN POSITION	66,890,000	664,000	230,000



FY 2019-20 Budget to Actuals

- Actuals are unaudited results through Q3 (July, 2019 March, 2020) plus Q4 projections
- Electricity sales are lower than expected due to reduced load based on weather conditions and COVID impacts and PCIA increases, effective May 1, 2020
- Cost of energy is lower than expected due to reduced load and softer energy prices
- Overhead is lower than projected primarily due to some deferred allocation of local development spend into the next fiscal year, and general cost saving measures across functional areas

	FY 2019-20	FY 2019-20	
	BUDGET	ACTUALS	Difference
REVENUE AND OTHER SOURCES	407 04 4 000	400.004.000	140.052.000
Electricity Sales	487,014,000	468,961,000	(18,053,000
Uncollectables	(2,447,000)	(2,346,000)	101,000
Investment Income	578,000	1,141,000	563,000
Other Income	-	113,000	113,000
Total Revenue and Other Sources	485,145,000	467,869,000	(17,276,000
EXPENSES AND OTHER USES			
ENERGY OPERATIONS			
Cost of Energy	386,905,000	377,424,000	(9,481,000
Scheduling/Broker Fees	653,000	758,000	105,00
Data Management/Customer Service	6,758,000	7,057,000	299,00
PG&E Service Fees (Billing/Metering)	2,253,000	2,326,000	73,00
Total Energy Operating Costs	396,569,000	387,565,000	(9,004,00
OVERHEAD OPERATIONS			
Personnel	6,703,000	5,871,000	(832,00
Marketing, Outreach, Communications	2,263,000	1,410,000	(853,00
Legal, Policy, & Reglatory Affairs	1,586,000	1,250,000	(336,00
Local Development	6,340,000	5,002,000	(1,338,00
Other Professional Services	1,214,000	685,000	(529,00
General & Administrative	2,290,000	1,607,000	(683,00
Depreciation	61,000	46,000	(15,00
	20 457 000	15 071 000	(4 596 00)
Total Overhead Operating Costs	20,457,000	15,871,000	(4,586,00
FOTAL ENERGY & OPERATING EXPENSES	417,026,000	403,436,000	(13,590,000
INTEREST PAYMENTS			
Borrowing Interest	1,229,000	743,000	(486,00
Total Interest Paym ents	1,229,000	743,000	(486,00
FOTAL EXPENSES & INTEREST DUE	418,255,000	404,179,000	(14,076,00



Updates to the Base Case

Both scenarios include the following updates

- No change to Brilliant 100 pricing through December 31, 2020 then set to cost of service at 2.5% premium to PG&E, with Board review and evaluation in summer/fall to determine the best path forward
- 90% migration of current Brilliant 100 customers to Bright Choice in January, 2021
- Power content for Brilliant 100 set to match RPS on renewables (33% in 2020, 35.8% in 2021) while maintaining 100% carbon free
- Energy costs are further reduced by sale of excess carbon-free assets
- Marketing is reduced by \$300k in community grants and noticings

The Non-COVID scenario sets Bright Choice at 1% discount to PG&E

The COVID scenario sets Bright Choice at 0.5% discount to PG&E



Brilliant 100 Considerations

- Brilliant 100 is maintained at parity through December 31, 2020 to allow for additional dialogue and planning with the board and community
- Rate is locked to new accounts and opt-ups as of July 1, 2020

Options for future consideration:

- Maintain product at a rate below cost of service for existing customers
- Discontinue product at some specified time and allow migration to other products or a new product
- Set product to cost of service, currently at 2.5% premium to PG&E
- Re-assessment of rate if market conditions become favorable
- Determine proper marketing/communication plan and customer

New Communities: Newark, Pleasanton, and Tracy

- Regardless of load scenario, inclusion of new communities is expected to yield positive margins at about \$3.0M based on an April enrollment timeframe estimated and current market conditions.
 - April-June timeframe represents higher margin months. Full calendar year net margin is lower.
 - COVID scenario yields slightly higher revenue because of Bright Choice discount
 - Because new communities are expected to join after the new year, there is no COVID load impact on costs
- Positive margin is driven by a modest increase in operating overhead related to serving these customers and softer energy prices.
- Additional analysis is underway to determine the optimal timing based on different customer rates
 Add procurement risks/opportunities

	FY 2020-21 Non-COVID	FY 2020-21 COVID	Scenario Difference
REVENUE AND OTHER SOURCES			
Electricity Sales	13,280,000	13,369,000	89,000
Uncollectables	(199,000)	(201,000)	(2,000)
Total Revenue and Other Sources	13,081,000	13,168,000	87,000
EXPENSES AND OTHER USES			
ENERGY OPERATIONS			
Cost of Energy	9,470,000	9,470,000	-
Data Management/Customer Service	299,000	299,000	-
PG&E Service Fees (Billing/Metering)	87,000	87,000	-
Total Energy Operating Costs	9,856,000	9,856,000	-
VERHEAD OPERATIONS			
Marketing, Outreach, Communications	330,000	330,000	-
Total Overhead Operating Costs	330,000	330,000	-
OTAL ENERGY & OPERATING EXPENSES	10,186,000	10,186,000	-
NET INCREASE (DECREASE) IN POSITION	2,895,000	2,982,000	87,000

Quarterly Distribution by Scenario

FY 2020-21 BUDGET Non-COVID	Q1	Q2	Q3	Q4	FY 2020-21 BUDGET COVID	Q1	Q2	Q3	Q4
REVENUE AND OTHER SOURCES	121,785,000	97,813,000	75,065,000	85,265,000	REVENUE AND OTHER SOURCES	115,277,000	97,148,000	76,262,000	86,011,000
ENERGY OPERATIONS OVERHEAD OPERATIONS	101,302,000 5,228,000	93,475,000 5,228,000	89,755,000 5,228,000	73,822,000 5,228,000	ENERGY OPERATIONS OVERHEAD OPERATIONS	99,945,000 5,228,000	92,224,000 5,228,000	88,554,000 5,228,000	72,833,000 5,228,000
NET INCREASE (DECREASE) IN POSITION	15,255,000	(890,000)	(19,918,000)	6,215,000	NET INCREASE (DECREASE) IN POSITION	10,104,000	(304,000)	(17,520,000)	7,950,000

- Q1 results are generally the strongest due to summer rates. Includes largest COVID load reductions
- Q2 results are flat due to transition to winter rates
- Q3 results are negative due to winter rates and increase in PCIA to reflect hitting the under-collection trigger.
- Q4 results are positive due to transition to summer rates and low spring energy


Overview of Revenue

	FY 2020-21	FY 2020-21	Scenario
	Non-COVID	COVID	Difference
REVENUE AND OTHER SOURCES			
Electricity Sales	388,125,000	382,758,000	(5,367,000)
Investment Income	1,231,000	1,231,000	-
Other Income	275,000	275,000	-
Uncollectables	(9,703,000)	(9,568,000)	135,000
Total Revenue and Other Sources	379,928,000	374,696,000	(5,232,000)

- The previous draft had Bright Choice at 1% discount and Brilliant 100 at 3% premium, with no migration
- These scenarios assume no changes to Brilliant 100 through 2020, then adjustment in 2021 to 2.5%, and 90% migration of Brilliant 100 customers to Bright Choice in 2021
- Non-COVID has Bright Choice at 1% discount, COVID has Bright Choice at 0.5% discount
- Other effects on Electricity Sales revenue exogenous to COVID load impacts still being considered are:
 - PUBA trigger occurs in December and PCIA increase in January to maintain 7% under-collection
 - 2021 PCIA rate changes occur after June 30, 2021--beyond the scope of these scenarios
 - Additional rate increase assumed in Jan, 2021 per GRC
- Investment Income is due to higher bank account balances in interest bearing accounts (reserves and ICS), projected at 1.5% interest. Current interest rate is 1.79%
- Other Income is grant from EDPR and is allocated to Local Development COVID Relief grants
- Uncollectables increased from 0.5% to 2.5%, in anticipation of COVID recessionary impacts



Overview of Expenses: Energy Operations

Total Energy Operating Costs	358,354,000	353,556,000	(4,798,000)	COVID Energy Costs
				67%
Scheduling	660,000	660,000	-	Energy
PG&E Service Fees (Billing/Metering)	2,396,000	2,396,000	-	
Data Management/Customer Service	7,704,000	7,704,000	-	Capacity 19%
Cost of Energy	347,594,000	342,796,000	(4,798,000)	
ENERGY OPERATIONS				9%
	Non-COVID	COVID	Difference	REC
	FY 2020-21	FY 2020-21	Scenario	CO2-Free 2% 2%
				CO2 Free CAISO Long-Term PPA

Energy Costs are made up of four primary energy products—block or shaped energy, renewable energy, carbon free energy, resource adequacy—and CAISO fees

- Costs of Energy are estimated based on contracted energy costs and estimated open position costs
- Costs are lower due to softer energy market prices
- Additional reductions due to anticipated PG&E large hydro allocation and revised power content targets
- Includes reduced load demand due to COVID scenario

Data Management, PG&E Service Fees, and Scheduling costs

- Data Management/Customer Service are paid to SMUD at \$1.05/meter/month through Dec of 2020. EBCE is currently reviewing
 options to extend the contract with SMUD and as a result, EBCE is preparing for potentially higher Data Management/Customer
 Service costs depending on outcome of action on SMUD contract. EBCE has input a placeholder cost of \$1.20/meter/month for the
 second half of the fiscal year to reflect the current uncertainty.
- PG&E Service Fees are paid to PG&E to use their billing and metering systems at \$0.35/customer/month
- Scheduling costs are paid to NCPA to manage EBCE's energy purchases and market activity



Overview of Expenses: Personnel

Because there is no anticipated impact on overhead operations due to COVID, the discussion of these areas will be in relation to the current year budget

	FY 2019-20	FY 2020-21	Fiscal Year
	BUDGET	BUDGET	Differ ence
Personnel	6,703,000	7,429,000	726,000

FY 2019-2020 budget was established with 37 FTE

• As of June 1, 2020 staff consists of 33 FTE

Current hiring plan for remainder of FY 2019-2020 and FY 2020-2021 is 6 FTE to remain flat to the FY 2019-2020 hiring plan

- Data Engineer for Technology
- CRM implementation Manager for Local Development/Technology
- Human Resources Business Partner for Operations
- Power Contracts Manager for Power Procurement

Increase in FY 2020-21 budget is driven by:

- Staff salaries covering the full fiscal year (FY 2019-2020 budget assumed partial year hires)
- Enact a policy to allow staff to monetize PTO (\$110,000 assumes full monetization)
- 2.5% merit-based adjustments (\$118,000)
- 3% discretionary promotions/retention-based compensation adjustments (\$169,000)



Overview of Expenses: Personnel Continued

- Actuals through Q3 with projections for Q4
- Reduction in current fiscal year personnel costs reflect hiring deferrals/delays
- Current Q4 personnel costs are forecasted at \$1,747,000, which includes the addition of 3 FTE in Q4



Personnel Cost by Fiscal Quarter 2019-2020-Budget to Actuals



Overview of Expenses: Marketing, Outreach, and Communications

	FY 2019-20	FY 2020-21	Fiscal Year
	BUDGET	BUDGET	Difference
Marketing, Outreach, Communications	2,263,000	1,214,000	(1,049,000)

	2019-2020	2020-2021
Advertising/Sponsorships/Events:	587,000	289,000
Communications:	223,000	290,000
Community Grants	190,000	-
Data manager:	300,000	300,000
Noticing:	133,000	75,000
Promotional Items:	55,000	10,000
Mailings:	775,000	250,000
Total	2,263,000	1,214,000

Examples of Marketing, Outreach, and Communications expenses

- Advertising/Sponsorships/Events: Active community presence activities in local jurisdictions and operating communities
- Communications: Public relations, media, newsletters, consultants
- Data Manager: Data/Billing management system enhancements
- Noticing: New account noticing (cost shown does not include new communities)
- Promotional Items: Promotional items for outreach/marketing events
- Mailings: Joint Rate Mailer/Power Content Label



Overview of Expenses: Legal, Policy, and Regulatory Affairs

	FY 2019-20	FY 2020-21	Fiscal Year
	BUDGET	BUDGET	Difference
Legal, Policy, & Reglatory Affairs	1,586,000	1,297,000	(289,000)

	2019-2020	2020-2021
Legal Consultants	1,076,000	970,000
Legislative Consultants	132,000	132,000
Other Consultants	259,000	195,000
Sponsorships	50,000	-
Temp. Employee	69,000	-
Total	1,586,000	1,297,000

Examples of Legal, Policy, and Regulatory Affairs expenses

- Legal Consultants: Outside General Counsel and legal counsel for procurement, programs, finance, and regulatory activities
- Legislative Consultants: Funds towards legislative advocacy
- Other Consultants: Economic analysis in regulatory/legislative cases and communications consultant



Overview of Expenses: Other Professional Services

	FY 2019-20 BUDGET	FY 2020-21 BUDGET	
Other Professional Services	1,214,000	1,345,000	131,000

	2019-2020	2020-2021
Accounting	205,000	220,000
Human Resources Consulting	25,000	-
IT Consulting	36,000	45,000
Other Professional Services	948,000	730,000
Customer Relationship Management	-	350,000
Total	1,214,000	1,345,000

Examples of Other Professional Services

- Accounting and auditing for financial compliance
- Technical consultants related to power procurement, risks analysis, and planning
- Customer Relationship Management include licensing and implementation of a CRM system to be utilized by the Local Development and Marketing teams for program offerings



Overview of Expenses: General & Administrative

	FY 2019-20 BUDGET	FY 202 BUD	20-21 GET	Fiscal Yea Differ end
General & Administrative	2,290,000	2,146	6,000	(144,00
		2019-2020	2020-20	021
Small Equipment & Softw	vare	636,000	740,0	000
Dues & Memberships		556,000	400,0	000
Operational Expenses		491,000	496,0	000
Rent & Utilities		470,000	429,0	000
Conferences & Prof. Deve	elopment	137,000	81,0	000
Total		2,290,000	2,146,0	000

Examples of General & Administrative Expenditures

- Small Equipment & Software includes professional software across all functional areas as well as equipment replacement as needed
- Dues & Memberships includes \$375,000 for CalCCA and other essential cooperative organizations
- Operational Expenses include Energy Prepay fees, benefits-related administrative fees, office supplies, and miscellaneous expenses
- Rent & Utilities covers all planned costs for office space
- Conferences & Professional Development includes travel and lodging



Local Development Fund

In addition to the programs listed to the right, there is approximately an additional \$4.7M budgeted for locally generated wholesale energy projects, such as EBCE's Altamont wind project, which is expected to come online during the fiscal year

These figures do not reflect any local spending based on G&A expenses

LOCAL DEVELOPMENT FUND	FY 2019-20 BUDGET	FY 2020-21 BUDGET	Fiscal Year Difference
Program Funding			
Beginning Balance	-	1,398,000	1,398,000
Funding from General Fund Revenues	6,340,000	6,340,000	-
Grants/Credits	1,150,000	275,000	(875,000)
Mid-Year adjustment	500,000	-	(500,000)
Total Available Funding	7,990,000	8,013,000	23,000
Program Areas			
Demand Response	195,000	100,000	(95,000)
Energy Efficiency	100,000	740,000	640,000
Building Electrification	782,000	950,000	168,000
Vehicle Electrification	1,750,000	4,930,000	3,180,000
Collaboartive Procurement	810,000	958,000	148,000
Community Investment	250,000	-	(250,000)
Sponsorships/Events	65,000	-	(65,000)
Capital Set Aside	1,200,000	-	(1,200,000)
COVID-19 Relief Fund	1,440,000	335,000	(1,105,000)
Total Program Expenses	6,592,000	8,013,000	1,421,000
Ending Balance	1,398,000	-	



Proposed Local Development Budget

Program Areas	20	021 Budget (\$000)	% of Budget
Demand Response	\$	100	1.58%
Energy Efficiency	\$	340	5.36%
Building Electrification	\$	700	11.04%
Vehicle Electrification	\$	4,400	69.40%
Collaborative Procurement	\$	800	12.62%
Local Development Budget Sub-Total	\$	6,340	
Staffing*			
Staff Costs	\$	825	
Fellow	\$	25	
Staffing Sub-Total	\$	850	

*Staffing cost come from HR budget not Local Development



LDBP Program Budgeting Process

- Process
 - LDBP Document continues to be the blueprint for program planning and budgeting
 - Staff will continue to update the CAC and Board on an ongoing basis as programs are planned and implemented
- Budgets Overview
 - FY'21 Budget remains flat to FY'20
 - Programs are focused on job creation and local investment
 - Team has successfully pursued external funding from Federal and State sources (EPA funds and CEC) to leverage EBCE funding



Transportation Electrification (\$4,400k)

- Electric Vehicle Incentive Program (\$3,770k)
- Center for Sustainability Incentive Administration and Support (\$254k)
- EViP Site Development Outreach and Development (\$126)
- Municipal Fleet Electrification Technical Support (\$125k)
- Medium/Heavy Duty Fleet Electrification (\$125k)



Building Electrification and Demand Response (\$800k)

- Building Electrification (\$700k)
 - Heat Pump Hot Water heater incentives (\$250k)
 - Reach Code Implementation Support (\$50k)
 - Commercial Induction Grants (\$300k)
 - Electrification Consumer Awareness (\$100)
- Demand Response
 - Building Electrification Demand Response (\$100k)



EE and Collaborative Procurement (\$1,140k)

- Collaborative Procurement (\$800k)
 - Solar+Storage Resilience Program (\$400k)
 - Connected Communities, reducing customer disconnections (\$50k)
 - EBCE Asset development and ownership (\$200k)
 - Municipal Renewable Electrification (\$150k)
- Energy Efficiency (\$340k)
- –Continue data sharing to increase efficacy of EE program
- –Phase II Pay for Performance EE Procurement (\$240k)
- —Technical/legal consultant to pursue public Energy Efficiency funding (\$100k)



Overview of Expenses: Interest Due

	FY 2019-20	FY 2020-21	Fiscal Year
	BUDGET	BUDGET	Difference
Borrowing Interest	1,229,000	804,000	(425,000)

EBCE currently has an active credit facility with Barclays Bank with no outstanding debt

The Barclays Credit Facility is sized at \$80,000,000

- EBCE can draw up to \$60MM in cash and use up to \$35MM in letters of credit (LC's)
- EBCE pays 1% on the undrawn facility balance as a commitment fee, and 1-month LIBOR + 2.50% on any cash drawn facility balance up to \$25MM and LIBOR + 2.65% on any amount above \$25MM
- EBCE has outstanding LC's and pays 1.75% on this balance
- The interest on outstanding LC's and the commitment fee is the budgeted interest payments for FY 2020-2021—no cash borrowing is expected for this fiscal year



Review of Net Position and Reserve Policy

EBCE Estimated Net Position 2019-2020 \$ 63,690,000

Reserve Fund	2018-19	2019-20*	Total	2019-20 Target	Remaining
Operating/Credit	32,768,648	23,393,450	56,162,098	200,646,000	144,483,902
Rate Stabilization	1,936,260	2,339,345	4,275,605	7,018,035	2,742,430
Collateral	1,936,260	2,339,345	4,275,605	37,742,400	33,466,795
Local Development	3,872,519	4,678,690	8,551,209	46,786,900	38,235,691
Total	40,513,687	32,750,830	73,264,517	292,193,335	218,928,818

* Expected contributions set to minimum amount

EBCE anticipates funding reserves with the minimum allocation from 2019-2020 net revenues for the following reasons:

- 1. Narrow 2020-21 margins
- 2. Uncertainty with COVID load demand
- 3. Uncertainty with 2021 PCIA adjustments
- 4. Retain a minimum 1 month liquidity of operating expenses (~\$30MM) under discretionary cash

The allocation for fiscal year 2019-20 is pending completion of a formal audit and expected to occur in Nov/Dec 2020



Comparison of EBCE Budget to CCA Space



As percentage of expenses



Comparison of EBCE Budget to CCA Space



As percentage of expenses



Comparison of CCA Products & Value

	Base Product	% Renewable	% Carbon-Free	Discount to PG&E (denotes a premium)	Upgrade 2	% Renewable	Premium to PG&E
EBCE	Bright Choice	39.5%	10%-15%	1.0%/0.5% (proposed)	Renewable 100	100% RE	+\$0.01/kWh
SVCE	Green Start	100%		4.00%	Green Prime	100% RE	+\$0.008/kWh
MCE	Light Green	61%	N/A	0%-2%	Deep Green	100% RE	+11%
SCP	Clean Start	49%	42%	(2%-%4)	Evergreen	100% RE	+\$0.025/kWh
SJCE	Green Source	45%	41%	1.00%	Total Green	100% RE	+\$0.005 - \$0.01/kWh
МВСР	MB Choice	30%	N/A	2%-7%	MB Prime	100% RE	+\$0.01/kWh
PCE	Eco Plus	51%	35%	5.00%	Eco 100	100% RE	+\$0.01/kWh
CPSF	Green	48%	40%		SuperGreen	100% RE	

*Yellow cells denote discounts/premiums relative to last year; not yet confirmed if these CCAs have made adjustments to these values/



Review of EBCE Value Proposition and Staff Recommendation

- EBCE currently offers three products:
 - Bright Choice: 1.5% below PG&E rates
 - Brilliant 100: Parity with PG&E rates
 - Renewable 100: \$0.01/kWh more expensive than PG&E rates
 - For 2019/2020, EBCE expects to deliver over \$7M in bill savings to customers
- Staff is recommending an adjustment to the value proposition for 2020/2021
 - Bright Choice: either 1.0% or 0.5% below PG&E rates depending upon COVID scenario
 - Brilliant 100: No change until 2021, then 2.5% premium to PG&E rates to reflect the projected procurement costs or some other option to be determined in the fall of 2020
 - Renewable 100: unchanged at \$0.01/kWh more expensive than PG&E rates
 - For 2020/2021 EBCE expects to deliver approximately between \$2.4M and \$3.6M in bill savings to customers (scenario dependent)
- Every 0.5% discount for Bright Choice equals \$3.2M in incremental revenue (with 90% migration in 2021)



Risks and Mitigations

EBCE is exposed to a variety of risks in this coming fiscal year. The following discussion is a representative list of key risks, though it is not comprehensive in reflecting all risks.

- 1. COVID/Load Risk
- 2. Energy Market Risk
- 3. Uncollectables/Write-off Risk
- 4. Opt-out Risk
- 5. Regulatory Risk



Risks and Mitigants: COVID/Load

COVID impacts on load, collections, and energy market prices are being closely monitored and to date the Shelter in Place policy has been in effect for approximately three months. While this has provided meaningful data it is very difficult to extrapolate the insights to accurately forecast impacts on the next fiscal year.

- 1. Status quo: Quick reversion to a pre-COVID load. Electricity usage based on businesses partial reopening and moderately increased residential use that results in load that closely reflects EBCE load pre-COVID.
- 2. Moderate COVID load decrease: Volumetric load is down 5-6% in the near-term months with a peak reduction of 10-15% and there is a gradual reversion to about 1-3% volumetric load reduction and 4-6% peak reduction, which is returned to baseline in 2021. This is the basis for the current budget.
- 3. Moderate COVID load increase: With the partial reopening of small businesses and large commercial space, we could see commercial load revert close to normal levels and residential load remain high based on partial families working from home. This could result in a moderate increase in volumetric load of 2-4% and a peak load similar to normal levels
- 4. Significant and sustained macro recession where volumetric load continues to be depressed. Volumetric load is down 6-10% and peak load is down 10-15% for next 12-24 months



Risks and Mitigants: COVID/Load continued

Independent of which scenario EBCE experiences, there are some constant risk conditions to consider.

- Changes in volumetric load due to ongoing COVID conditions are expected to decrease costs proportional with the projected net decline in demand. Thus, any decline in revenue due to decline in demand should be reasonably offset with a decline in costs. EBCE is projecting a margin of 8.5% of revenues over electricity costs. This means for each dollar decrease in revenue from reduced demand, EBCE can expect an approximate \$0.915 reduction in costs. To date, EBCE has generally experienced a positive impact to load shape that has reduced the more expensive peak hours that help to mute this margin impact.
- Load changes can have material implications to EBCE as it relates to planning, procurement, and scheduling. Less certainty when it comes to load forecasting can increase risk. Generally speaking, an overall industry reduction in load directionally leads to softer energy prices. However, the uncertainty also leads to more significant day to day volatility, which can increase or decrease energy costs where EBCE is not hedged.



Risks and Mitigants: Energy Market Risk

EBCE currently has an estimated unhedged energy portfolio of between approximately \$68-\$73M (scenario dependent) which is exposed to the market. Short term energy prices can see volatility of 10-20% on a regular basis based on typical supply/demand dynamics that are heavily influenced by weather and electricity infrastructure.

The volatility may be exacerbated by COVID due to general uncertainty of load and the impacts from the macro oil industry on gas prices. We estimate that energy costs could increase reasonably as much as \$16-20M, though this could also result in a similar reduction in energy costs.

EBCE could consider hedging more load to reduce this risk. However in an environment of softer energy prices, there are follow-on repercussions of an increasing PCIA the following year.



Risks and Mitigants: Uncollectables

While EBCE has increased our expected exposure from 0.5% to 2.5% of uncollectable accounts, the current data is limited in measuring the extent of this impact. This forecast of 2.5% is at the higher end of the range compared to other CCAs based on our regional income demographics. In researching other load serving entity uncollectables through prior recessions, we have seen data ranging from non-material impacts up to 6% write-offs.

Based on the recessionary impacts of COVID and suspension of disconnects, we do anticipate an increase in uncollectables that would lead to write-offs. There is a potential for the uncollectables to exceed our current forecast based on a more severe and extended recession. To mitigate this risk, we have launched an active campaign to increase CARE enrollments as a way to lower customers bills and are continuing to create a plan to enroll customers onto longer term payment plans as necessary. While enrollment in payment plans should reduce write-offs, it would reduce near term cash inflows. EBCE has a strong cash and reserve position to support payments deferrals.



Risks and Mitigants: Opt-out Risk

Customer opt-outs are generally an ongoing risk to EBCE. However, we do not anticipate an increased risk of opt-outs for the fiscal year, depending on adjustments to the value propositions. There is currently no active discussion of DA expansion that would take effect in 2021 beyond what has already been in process.



Risks and Mitigants: Regulatory Risk

- The current budget assumes a deferral of approximately \$16M in costs, which is based on maintaining an undercollection (PUBA) of 7%. The most significant regulatory risk to revenues in the near term is the management of this PUBA cap. The CPUC could rule to fully collect on the undercollection, which would decrease fiscal year revenues or potentially elect for a larger undercollection, which could increase revenues.
- AB110 Power Content reporting is anticipated to be finalized in mid 2020. These regulations will include potential updates to the reporting treatment of PCC3 unbundled RECs and Asset Controlling Supply (ACS) procurement. Currently procurement of PCC3 RECs is limited and serves as a flexible tool to balance the portfolio. ACS is also currently procured largely as a source of clean large hydro energy, a cost-effective energy hedging tool, and a source of import RA from the pacific northwest. The new reporting regulation may require disclosure of all underlying energy as source specified, which could limit future procurements of this product depending on board direction.
- PCIA Working Group 3, which is focused on the disposition of PG&E's current long portfolio may benefit EBCE's energy procurement related costs. Though those benefits are not expected to be material for the fiscal year.



Risk and Mitigants: Mitigants

Considering the previous discussion, EBCE has the following options to further mitigate these economic risks.

- Additional changes to EBCE's value proposition:
 - Further reduction to Bright choice discount to 0% = \$3.2M (with 90% migration in 2021)
- EBCE has the opportunity to accept and resell PG&E's nuclear allocation. This could generate up to \$2 million based on market demand, which is uncertain. This could present some risk to EBCE's power content label, but we can likely structure an arrangement to largely mitigate this risk.



Alternative Budget Scenarios

- Alternative 1: Reduce Local Development budget by \$3.2M and keep Bright Choice discount at 1.5% in Non-COVID scenario or 1.0% in COVID scenario, resulting in the current projected margin in the presented budget
- Alternative 2: Reduce Local Development budget by \$2.5M and keep Brilliant 100 locked, but at parity, for the whole fiscal year, resulting in projected margin in either scenario





Staff Report Item 18

TO: East Bay Community Energy Board of Directors

FROM: Howard Chang, Chief Operating Officer

SUBJECT: Legal Services Contracts Approval (Action Item)

DATE: June 17, 2020

Recommendation

Adopt a Resolution authorizing the Chief Executive Officer to negotiate and execute contracts with select legal firms in accordance with the legal budget, totaling \$970,000 for fiscal year 2020-2021.

Background and Discussion

The EBCE budget presented to the Board tonight includes a total legal budget of \$970,000 across regulatory, policy, power procurement, general counsel services, and other areas. Staff is seeking approval to contract with the following law firms:

- Braun Blaising Smith Wynne Policy
- Davis Wright Tremaine Policy
- Hall Energy Law Power Procurement
- Keyes and Fox Policy/Compliance/Power Procurement
- Liebert Cassidy Whitmore Employment/HR
- Richards, Watson and Gershon General Counsel Services
- Wilson Sonsini Goodrich Rosati Power Procurement

All firms were utilized in fiscal year 2019-2020, and all except one were utilized in fiscal year 2018-2019. Staff is pleased with the prior services and is interested in maintaining strong working relationships as it seeks new agreements with these vendors for fiscal year 2020-2021.

The anticipated budget breakdown across these functions is:

- Outside General Counsel: \$300,000 \$350,000
- Regulatory & Policy \$200,000 \$250,000
- Power Procurement and Local Development: \$350,000 \$400,000
- Other \$30,000 \$60,000

Fiscal Impact

Staff anticipates contract amount with the selected law firms in varying amounts that may exceed \$100,000. Staff is seeking approval of a resolution that would approve the CEO to negotiate and execute contracts with the identified legal firms consistent with the total budget of \$970,000.

Attachments

A. Resolution Authorizing the Chief Executive Officer to Negotiate and Execute Legal Service Agreements with Selected Law Firms.

RESOLUTION NO.

A RESOLUTION OF THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY AUTHORIZING THE CHIEF EXECUTIVE OFFICER TO NEGOTIATE AND EXECUTE LEGAL SERVICE AGREEMENTS WITH SELECTED LAW FIRMS

WHEREAS The East Bay Community Energy Authority ("EBCE") was formed as a community choice aggregation agency ("CCA") on December 1, 2016, under the Joint Exercise of Power Act (California Government Code sections 6500 *et seq.*) among the County of Alameda, and the Cities of Albany, Berkeley, Dublin, Emeryville, Fremont, Hayward, Livermore, Piedmont, Oakland, San Leandro, and Union City to study, promote, develop, conduct, operate, and manage energy-related climate change programs in all of the member jurisdictions. The Cities of Newark and Pleasanton, located in Alameda County, along with the City of Tracy, located in San Joaquin County, were added as members of EBCE and parties to the JPA in March of 2020; and

WHEREAS in June 2018, EBCE issued a broad legal services RFP to select law firms to provide legal expertise across various staff functions, including legislative and regulatory policy, power procurement, general counsel services, finance, and other areas; and

WHEREAS EBCE has previously utilized the following firms to secure such legal expertise: Braun Blaising Smith Wynne, Davis Wright Tremaine, Hall Energy Law, Keyes and Fox, Liebert Cassidy Whitmore, Richards, Watson and Gershon, and Wilson Sonsini Goodrich Rosati; and

WHEREAS EBCE is pleased with the prior services provided by these law firms; and

WHEREAS EBCE intends to enter into Consulting Services Agreements with the aforementioned law firms and seeks to direct and authorize the CEO to negotiate and execute such agreements in varying amounts that may exceed \$100,000.

NOW, THEREFORE, THE BOARD OF DIRECTORS OF THE EAST BAY COMMUNITY ENERGY AUTHORITY DOES HEREBY RESOLVE AS FOLLOWS:

<u>Section 1.</u> The CEO is hereby authorized to negotiate and execute Consulting Services Agreements for one-year terms with the law firms of Braun Blaising Smith Wynne, Davis Wright Tremaine, Hall Energy Law, Keyes and Fox, Liebert Cassidy Whitmore, Richards, Watson and Gershon, and Wilson Sonsini Goodrich Rosati.

<u>Section 2.</u> The Consulting Services Agreements may be for varying amounts that exceed \$100,000 but the total compensation of all the Agreements shall be consistent with the Board approved legal budget of \$970,000 for fiscal year 2020-2021.

ADOPTED AND APPROVED this 17th day of June, 2020.

Dan Kalb, Chair

ATTEST:

Stephanie Cabrera, Clerk of the Board